

Skin carcinomas in organ-transplant recipients: from early oncogenic events to therapy

Graaf, Y.G.L. de

Citation

Graaf, Y. G. L. de. (2008, January 23). *Skin carcinomas in organ-transplant recipients: from early oncogenic events to therapy*. Department of Dermatology, Faculty of Medicine, Leiden University Medical Center (LUMC), Leiden University. Retrieved from https://hdl.handle.net/1887/12579

Version:	Corrected Publisher's Version
License:	Licence agreement concerning inclusion of doctoral thesis in the Institutional Repository of the University of Leiden
Downloaded from:	https://hdl.handle.net/1887/12579

Note: To cite this publication please use the final published version (if applicable).

Skin carcinomas in organ-transplant recipients:

from early oncogenic events to therapy

Leontien de Graaf

Skin carcinomas in organ-transplant recipients: from early oncogenic events to therapy

Proefschrift

ter verkrijging van de graad van Doctor aan de Universiteit Leiden, op gezag van Rector Magnificus prof.mr. P.F. van der Heijden, volgens besluit van het College voor Promoties te verdedigen op woensdag 23 januari 2008 klokke 16.15 uur

door

Ymke Grete Leontien de Graaf

geboren te 's Gravenhage in 1976

Promotiecommissie:

Promotor:	Prof. dr. R. Willemze
Co-promotores:	Dr. J.N. Bouwes Bavinck Dr. F.R. de Gruijl
Referent:	Prof. dr. J.W. de Fijter
Overige Leden:	Prof. dr. L. Mullenders Prof. dr. W. Spaan

Aan mijn ouders Aan Remko

The study described in Chapter 6 was financially supported by Zon MW. Publication of this thesis was financially supported by Astellas Pharma, Bauerfeind, Eucerin, Fagron B.V., Galderma, Glaxo Smith Kline, La Roche Posay, Leo Pharma, Louis Widmer, Meda Pharma B.V., Mölnlycke, Neutral huidverzorging, Novartis Pharma B.V., Roche, Schering-Plough, Vichy, and Wyeth Pharmaceuticals.

Skin carcinomas in organ-transplant recipients: from early oncogenic events to therapy

Thesis, Rijksuniversiteit Leiden, The Netherlands

Copyright © 2007, Leontien de Graaf, The Netherlands No part of this thesis may be reproduced, stored or transmitted without prior permission of the author.

CONTENTS

Chapter 1	General introduction	9
Chapter 2	More epidermal p53 patches adjacent to skin carcinomas in renal-transplant recipients than in immunocompetent individuals: the role of azathioprine	21
Chapter 3	p53-specific serum antibodies are not associated with a history of skin carcinoma in renal-transplant recipients and immunocompetent individuals	31
Chapter 4	UV-induced apoptosis is not diminished in the presence of beta-papillomaviruses in habitually unexposed skin, but does decrease with age	37
Chapter 5	Systemic and topical retinoids in the management of skin cancer in organ-transplant recipients	51
Chapter 6	Photodynamic therapy does not prevent cutaneous squamous cell carcinoma in organ-transplant recipients: results of a randomized-controlled trial	59
Chapter 7	The occurrence of residual or recurrent squamous-cell carci- nomas in organ-transplant recipients after curettage and electrodessication	67
Chapter 8	Summary and general discussion	75
Appendices	Nederlandse samenvatting	87
	Curriculum vitae	93
	Publicaties	97
	Nawoord	101

CHAPTER 1

General Introduction

Skin cancer in organ-transplant recipients

In 1954, the first renal transplantation was performed in Boston.¹ With the introduction of the immunosuppressant azathioprine in the 1960s renal transplantation became a good alternative to dialysis. In the early 1980s cyclosporine was introduced and since then a handful new immunosuppressant agents were marketed. With more effective immunosuppression, long-term survival after organ transplantation has increased substantially. As a result, the number of patients with long-term complications of transplantation is also increasing. Skin cancers are the most common post-transplantation malignancies and account for substantial morbidity and mortality.²⁻⁶ The largest group of organtransplant recipients is formed by renaltransplant recipients. The problem of skin cancer is not limited to renal-transplant recipients, but is also eminent in recipients of other organs. This introductory chapter highlights the problem of skin cancer in organtransplant recipients and will discuss a) risk factors and related mechanisms that are relevant to the development of skin cancer and b) the clinical management of organtransplant recipients with skin carcinomas and multiple precursor and associated skin lesions.

The incidence of skin cancer in organtransplant recipients increases with time after transplantation as well as with decreasing latitude (Figure 1).⁷ In countries with temperate climates, such as The Netherlands, 40% of organ-transplant recipients have skin cancer 20 years after transplantation³, compared to a percentage of 70% in subtropical countries like Australia.⁷

The most prevalent tumours in organtransplant recipients are squamous-cell carcinomas that are predominantly located on sun-exposed areas (Figure 2).^{3,8} Squamouscell carcinomas occur 65 to 250 times more frequently than in the general population.^{3,4,9} The incidence of squamous-cell carcinoma of the lip is also increased (15 to 20-fold).^{9,10} The incidence of basal-cell carcinomas is increased by a factor 10 in transplant recipients.³ This results in a reversed ratio of basal-cell to squamous-cell carcinomas in these patients compared with the general population.^{2,7,11} Moreover, squamous-cell carcinomas appear to be more aggressive in organ-transplant recipients than in immunocompetent individuals. This is manifest in increases in local recurrences, regional and distant metastases and mortality.^{12,13}



Figure 1. Cumulative incidence of skin cancer after transplantation in Queensland, Australia and Leiden, The Netherlands (modified from Bouwes Bavinck et al⁷).



Figure 2. Squamous cell carcinoma in an organ-transplant recipient.



Figure 3. Hyperkeratotic lesions in an organ-transplant recipient

The risk of metastasis from squamous-cell carcinoma in these immunocompromized patients is estimated to be approximately 7%.¹² The regional lymph node has been reported to be the primary site of metastasis of squamous-cell carcinomas.¹³ The presence of multiple skin cancers and localisation in the head and neck region is associated with an aggressive clinical course.^{13,14} Other types of skin cancer of which the incidences are increased in organ-transplant recipients are Kaposi's sarcoma (84 to 113-fold)^{6,9,15} melanoma (2 to 8-fold)^{5,9,16} and Merkel cell carcinoma.¹²

In addition to the increased incidence of skin cancer, these patients develop numerous viral warts and actinic keratoses.^{8,17} Compared to the general population, these lesions are more often resistant to therapy and frequently large areas are affected. In particular, the scalp and dorsal surfaces of the hands

and forearms can show multiple confluent, hyperkeratotic lesions (Figure 3).¹²

Risk factors for skin carcinogenesis in organ-transplant recipients

The pathogenesis of skin cancer is multifactorial, with extrinsic and intrinsic risk factors (Figure 4). Sun exposure and prolonged immunosuppressive therapy have been recognized as the most important risk factors for skin cancer in organ-transplant recipients. In addition, human papillomaviruses might play a role in skin carcinogenesis. These topics will be discussed in more detail below.

Other risk factors for the development of skin cancer in organ-transplant recipients are gender, age, smoking, time after transplantation and the duration of pre-transplantation dialysis.^{2,7,18} A fair complexion and an inability to tan are well-known genetic risk factors.^{19,20}



Figure 4. Hypothetical mechanisms of skin carcinogenesis in organ-transplant recipients

Ultraviolet radiation

The importance of exposure to sunlight as a risk factor is reflected by the fact that countries with high insolation have the highest incidence of skin cancer⁷ and the tumours predominantly develop in sun-exposed areas.⁸ It is assumed that the oncogenic properties of UV radiation are not only due to a direct mutagenic effect, but also to an immunosuppressive effect.

Mutagenic effect

Solar ultraviolet radiation at the Earth's surface, specifically the short wavelength UVB radiation, induces damage in DNA, mainly cyclobutane pyrimidine dimers and (6,4) photoproducts. If not adequately repaired, this damage gives rise to C \rightarrow T and CC \rightarrow TT transitions at dipyrimidine sites. These mutations are characteristic of UV radiation and are therefore called 'UV signature mutations'.²¹ In the general (immunocompetent) population these mutations are found in the p53 tumour suppressor gene in the majority of squamous-cell carcinomas²², and their precursor lesions, actinic keratoses.²³

Upon chronic UV irradiation, clusters of epidermal cells occur that are readily immunohistochemically detectable by overexpression of the p53 protein. It has been suggested that that these clones expand at the expense of neighbouring keratinocytes owing to differential apoptotic responses under UV exposure.²⁴ Previous studies using microdissection showed that 30-70% of the p53 patches in human skin contained p53 gene mutations, of which the majority has the typical UV signature.²⁵⁻²⁹

P53 patches are found long before the appearance of skin carcinomas in the hairless mouse model.^{30,31} As these p53 patches bear UV-specific mutations similar to those in the subsequent carcinomas, they appear to be early microscopic precursor lesions of the ultimate tumours.³² This hypothesis is also supported by studies in human skin. An earlier study showed a significant dose-response relation between UV radiation and frequency of p53 patches.²⁶ Another study reported

more p53 patches to be present adjacent to basal-cell carcinomas than adjacent to benign skin lesions²⁸, and again more adjacent to squamous-cell carcinomas than basal-cell carcinomas.³³

However, it is not clear whether p53 patches are more prevalent in immunocompromized patients. Therefore we studied whether the number of p53 patches in uninvolved skin adjacent to carcinomas was increased in organ-transplant recipients when compared to immunocompetent patients. This study is described in **Chapter 2**.

Immunosuppressive effect

From classic animal experiments it is known that UV-induced skin cancers are antigenic and subject to elimination by the immune system. Subcarcinogenic doses of UV radiation can suppress the rejection and even induce specific tolerance toward the tumour.^{34,35}

UV-induced immunosuppression is a highly complex process in which several different pathways are involved. UV radiation reduces the number and function of epidermal Langerhans cells, the major antigen presenting cells in the epidermis.³⁶ Next to DNA damage and oxidative damage³⁶, the formation of cis-urocanic acid by photoisomerisation of transurocanic acid can modify antigen presentation through ligation to serotonin receptors.³⁷ In addition, UV radiation stimulates keratinocytes, and subsequently leukocytes, to release immunosuppressive soluble mediators that affect antigen presentation, including interleukin 10, which enter the circulation and thereby also induce systemic immunosuppression.³⁸ Another important effect of UV radiation is the induction of regulatory T cells that appear to play a role in UV-induced tolerance.^{38,39}

Immunosuppressive treatment

The lifelong immunosuppressive therapy of organ-transplant recipients usually consists of prednisone in combination with immunosuppressants such as azathioprine, cyclosporine, mycophenolate mofetil, tacrolimus or more recently sirolimus (rapamycine). It is plausible that immunosuppressive drugs create a state in which immunosurveillance and eradication of malignant cells are impaired, facilitating carcinogenesis.¹² In addition, these immunosuppressive agents cause direct adverse effects on the skin cells (keratinocytes), that can increase the carcinoma risk. The classic immunosuppressives azathioprine (Imuran[®]) and cyclosporine (Neoral[®]) interfere with DNA repair.40,41 It has been shown that azathioprine also enhances DNA damage by photosensitization; it is incorporated into the DNA as a thio-guanine pseudobase, which can sensitize the DNA to solar UVA radiation-induced damage.42,43 In addition, it has been suggested that cyclosporine induces transforming growth factor β (TGF- β) production by tumour cells resulting in invasive growth.⁴⁴

An experimental study from the late 1980s in which mice were treated with different immunosuppressive agents and exposed to UV radiation to induce skin tumours showed that azathioprine had the greatest effect on skin cancer development. Azathioprine increased the number of tumours per mouse and decreased the time to the first tumour, while cyclosporine decreased only the time to tumour induction to a minor extent.⁴⁵

The clinical studies in which the role of the different treatments was studied concerning skin cancer risk are inconclusive. Some studies did not show a difference in skin cancer incidence between azathioprine and cyclosporine groups.^{2,7} Other studies reported that patients receiving cyclosporine, azathioprine and prednisone had an increased risk of squamous-cell carcinoma compared with patients taking only prednisone and azathioprine.^{9,46} Unfortunately, most of the clinical studies are retrospective and consist of large registry reports. Comparison of incidence rates is therefore difficult, because the patient populations (azathioprine vs. cyclosporine) are from different time periods. Moreover, the increased skin carcinoma risk in some of these studies may also be attributed to the immunosuppressive dosages,

i.e., level of immune suppression, in combination with the duration of the treatment. In a randomized prospective study in which lowdose cyclosporine was compared with standard-dose cyclosporine, the low-dose regimen resulted in a significantly lower incidence of skin cancer.⁴⁷ This was consistent with another prospective study that also found an association with the overall cumulative immunosuppressive dose.²⁰

More recent studies on newer drugs suggest that sirolimus, which has anti-tumour effects, confers a lower skin cancer risk compared with the classic immunosuppressive therapies.⁴⁸⁻⁵⁰ However, this needs to be confirmed in carefully designed prospective randomized clinical trials, since skin cancers take years after transplantation to develop.

Beta-papillomaviruses

It has been shown that the development of skin cancer in organ-transplant recipients is strongly associated with the number of keratotic skin lesions, mainly viral warts and actinic keratoses.^{8,17}

Numerous studies suggested that human papillomaviruses may be co-carcinogenic.⁵¹⁻ ⁵³ On the basis of their tropism human papillomaviruses may be classified as genital (mucosal) or cutaneous. Genital human papillomaviruses are subdivided into highand low-risk virus types according to their association with malignancies and their invitro cell-transforming capacity. The cutaneous human papillomaviruses can be subdivided into the classical types associated with warts, such as verrucae vulgares and verrucae plantares, and the epidermodysplasia verruciformis types. The latter have recently been renamed as beta-papillomaviruses (beta-PV).54

Role of beta-PV in skin carcinogenesis

Infection with beta-PV occurs frequently and may persist for many years.⁵⁵ A wide diversity of beta-PV-types can be detected in both pre-malignant skin lesions and skin carcinomas.^{17,56-59} Earlier studies provide indirect evidence that beta-PV may play a role in skin cancer development either directly or in combination with sun exposure. The hair follicle is a possible reservoir for the beta-PV types. It has been shown that the prevalence of beta-PV-DNA in plucked eyebrow hairs is higher in immunocompetent individuals with a history of squamous-cell carcinoma than in controls.⁵¹ Moreover, patients with a history of squamous-cell carcinoma are more likely than controls to have a sero-response against beta-PV.^{52,53} The early viral protein E6 of some beta-PV types may impair the process of DNA repair or prevent apoptosis after exposure to UV radiation.⁶⁰⁻⁶³ As a result, beta-PV infected, DNA-damaged cells may become genomically unstable, and survive. Such cells may ultimately give rise to actinic keratoses and squamous-cell carcinomas (Figure 5).⁶⁴ A recent study provided direct evidence for the carcinogenic potential of beta-PV by showing non-melanoma skin cancer development in HPV-8 transgenic mice without any treatment with physical or chemical carcinogens.65

It has been shown that beta-PV inhibit the apoptotic response to UV damage in-vitro. The aim of our study, described in **Chapter 4**, was to investigate whether apoptosis was decreased in the presence of beta-PV after an UVB challenge in human skin in-vivo.

Prevention and treatment options for skin cancer in organ-transplant recipients

The most important element of preventive management in organ-transplant recipients is patient education. All patients should receive information, before and after their transplantation, about the increased risk of skin cancer and the harmful effects of excessive sunlight exposure.⁶⁶ Furthermore, education on photoprotection, self-examination and the recognition of (pre)-malignant lesions is required. Monthly self-examination of skin as well as regular examination by physicians should be encouraged. Patients with pre-malignant skin lesions should be referred to a dermatologist in an early stage for intensive surveillance and active treatment of premalignant lesions and cancers.^{12,67}

Prevention of skin cancer

Available studies have suggested a beneficial effect of systemic retinoids in chemoprevention of transplant-related skin cancers. Retinoids are structural and functional analogues of vitamin A that display a wide range of biological activity. Possible mechanisms by which they prevent or reduce skin cancer development include induction of apoptosis, normal differentiation of keratinocytes, and immunomodulation.68,69 Organ-transplant recipients who may benefit from retinoid chemoprevention are those who are developing large numbers of skin cancers.^{70,71} Chapter 5 provides a review on the role of topical and systemic retinoids in the chemoprevention of skin cancer in organ-transplant recipients.

Another possible modality in the prevention of skin cancer is photodynamic therapy, which involves the use of a photosensitizing agent and a light system. Photodynamic therapy can be used to treat superficial skin carcinomas or precancerous lesions that are accessible to light.⁷² It has been shown that photodynamic therapy is a safe and effective treatment for actinic keratoses in organtransplant recipients.^{73,74} In addition, earlier experimental studies showed that photodynamic therapy can delay the development of UV-induced skin carcinomas.75,76 We studied this hypothesis in organ-transplant recipients in a randomized-controlled trial. This study is described in Chapter 6.

Obviously, aggressive treatment of premalignant lesions, such as actinic keratoses, is essential to minimize the progression to squamous-cell carcinoma. For this purpose, treatments such as cryotherapy, topical retinoids, 5-fluorouracil or the immune-response modifier imiquimod can be used. The same is true for actinic cheilitis because of the increased risk of high-risk squamous-cell carcinoma of the lip.¹²

Finally, reduction of immunosuppression is considered a reasonable adjuvant management strategy for organ-transplant recipients who develop numerous or life-threatening skin cancers.⁷⁷



Figure 5. A proposed scheme for the development of actinic keratoses and cutaneous squamous-cell carcinoma (adapted from Bouwes Bavinck et al^{64}).

Management of skin cancer

For squamous-cell carcinomas in organtransplant recipients the treatment of choice is surgical excision with histological examination.⁷⁸ Resurfacing the dorsum of the hand can be a useful in selected patients. With this surgical procedure the tumour(s) and the actinically damaged skin are resected.⁷⁹ On high-risk tumours, Mohs' micrographic surgery can be performed.⁸⁰

In selected tumours, curettage and electrodessication, a destructive modality, is an option, but there is not much evidence of its efficacy as a treatment of squamous-cell carcinomas in organ-transplant recipients in the literature. Only one case is described in which multiple squamous-cell carcinomas were successfully treated by curettage.⁸¹ Nevertheless, curettage and electrodessication appears to be widely used in organ-transplant recipients, usually for superficial or early skin cancers.⁸² Therefore, we evaluated the recurrence risk of squamouscell carcinomas after treatment with curettage and electrodessication in organtransplant recipients and compared the recurrence rates at different skin locations. This retrospective follow-up study is described in **Chapter 7**.

Aim and structure of this thesis

The aim of the studies presented in this thesis is broadly two fold:

i) identify early oncogenic events (such as high numbers of p53 patches and reduced apoptosis) that could explain the high risk of skin carcinoma in organ-transplant recipients, and

ii) contribute to improved prevention and therapy of skin carcinomas in organ-transplant recipients.

Clearly, advancements under point i) can contribute to prevention and improved clinical intervention, mentioned under point ii).

Chapters 2 through 4 are related to point i), whereas the Chapters 5 through 7 are related to point ii).

Chapter 2 investigates whether the enhanced risk of skin carcinomas in organtransplant recipients is reflected in increased p53 patches in their skin compared with immunocompetent patients. In addition, two possible mechanisms by which azathioprine might increase p53 patches were investigated: immunosuppression and impaired DNA repair.

Chapter 3 describes the prevalence of p53specific serum antibodies in both renaltransplant recipients and immunocompetent individuals with and without a history of squamous-cell carcinoma.

Chapter 4 investigates whether beta-PV affect UV-induced apoptosis in unexposed skin of organ-transplant recipients and immunocompetent individuals and studies the effect of UVB exposure on beta-PV presence.

Chapter 5 provides a review on the efficacy of topical and systemic retinoids in the prevention of skin cancer in organ-transplant recipients.

Chapter 6 describes a randomizedcontrolled trial with paired observations in 40 organ-transplant recipients in which the effect of photodynamic therapy on the occurrence of new squamous-cell carcinomas on sun-exposed skin was assessed.

In **Chapter 7** a series of squamous-cell carcinomas from organ-transplant recipients that were treated with curettage and coagulation was studied, in order to assess the recurrence rate after this treatment and to compare the recurrence rates at different skin locations.

Chapter 8 summarizes and discusses the findings described in the preceding chapters.

The results are compared with other, more recent studies. Furthermore, possibilities for future research are suggested.

References

- 1. Morris PJ. Transplantation--a medical miracle of the 20th century. N Engl J Med 2004;351:2678-2680.
- London NJ, Farmery SM, Will EJ, Davison AM, Lodge JP. Risk of neoplasia in renal transplant patients. Lancet 1995;346:403-406.
- Hartevelt MM, Bavinck JN, Kootte AM, Vermeer BJ, Vandenbroucke JP. Incidence of skin cancer after renal transplantation in The Netherlands. Transplantation 1990;49:506-509.
- Lindelof B, Sigurgeirsson B, Gabel H, Stern RS. Incidence of skin cancer in 5356 patients following organ transplantation. Br J Dermatol 2000;143:513-519.
- Adami J, Gabel H, Lindelof B, Ekstrom K, Rydh B, Glimelius B, Ekbom A, Adami HO, Granath F. Cancer risk following organ transplantation: a nationwide cohort study in Sweden. Br J Cancer 2003;89:1221-1227.
- Moloney FJ, Comber H, O'Lorcain P, O'Kelly P, Conlon PJ, Murphy GM. A population-based study of skin cancer incidence and prevalence in renal transplant recipients. Br J Dermatol 2006;154:498-504.
- Bouwes Bavinck JN, Hardie DR, Green A, Cutmore S, MacNaught A, O'Sullivan B, Siskind V, van der Woude FJ, Hardie IR. The risk of skin cancer in renal transplant recipients in Queensland, Australia. A follow-up study. Transplantation 1996;61:715-721.
- Bouwes Bavinck JN, De Boer A, Vermeer BJ, Hartevelt MM, van der Woude FJ, Claas FH, Wolterbeek R, Vandenbroucke JP. Sunlight, keratotic skin lesions and skin cancer in renal transplant recipients. Br J Dermatol 1993;129:242-249.
- Jensen P, Hansen S, Moller B, Leivestad T, Pfeffer P, Geiran O, Fauchald P, Simonsen S. Skin cancer in kidney and heart transplant recipients and different long-term immunosuppressive therapy regimens. J Am Acad Dermatol 1999;40:177-186.
- King GN, Healy CM, Glover MT, Kwan JT, Williams DM, Leigh IM, Worthington HV, Thornhill MH. Increased prevalence of dysplastic and malignant lip lesions in renaltransplant recipients. N Engl J Med 1995;332:1052-1057.
- Hardie IR, Strong RW, Hartley LC, Woodruff PW, Clunie GJ. Skin cancer in Caucasian renal allograft recipients living in a subtropical climate. Surgery 1980;87:177-183.
- Berg D, Otley CC. Skin cancer in organ transplant recipients: Epidemiology, pathogenesis, and management. J Am Acad Dermatol 2002;47:1-17.
- Martinez JC, Otley CC, Stasko T, Euvrard S, Brown C, Schanbacher CF, Weaver AL. Defining the clinical course of metastatic skin cancer in organ transplant recipients: a multicenter collaborative study. Arch Dermatol 2003;139:301-306.
- Adamson R, Obispo E, Dychter S, Dembitsky W, Moreno-Cabral R, Jaski B, Gordon J, Hoagland P, Moore K, King J, Andrews J, Rich M, Daily PO. High incidence and clinical course of aggressive skin cancer in heart transplant patients: a single-center study. Transplant Proc 1998;30:1124-1126.
- Serraino D, Piselli P, Angeletti C, Minetti E, Pozzetto A, Civati G, Bellelli S, Farchi F, Citterio F, Rezza G, Franceschi S, Busnach G. Risk of Kaposi's sarcoma and of other cancers in Italian renal transplant patients. Br J Cancer 2005;92:572-575.
- Le Mire L, Hollowood K, Gray D, Bordea C, Wojnarowska F. Melanomas in renal transplant recipients. Br J Dermatol 2006;154:472-477.
- 17. de Jong-Tieben LM, Berkhout RJ, ter Schegget J, Vermeer BJ, de Fijter JW, Bruijn JA, Westendorp RG, Bouwes Bavinck JN. The prevalence of human papillomavirus DNA in benign keratotic skin lesions of renal transplant recipients with and without a history of skin cancer is equally high: a clinical

study to assess risk factors for keratotic skin lesions and skin cancer. Transplantation 2000;69:44-49.

- Ramsay HM, Fryer AA, Reece S, Smith AG, Harden PN. Clinical risk factors associated with nonmelanoma skin cancer in renal transplant recipients. Am J Kidney Dis 2000;36:167-176.
- Lindelof B, Granath F, Dal H, Brandberg Y, Adami J, Ullen H. Sun habits in kidney transplant recipients with skin cancer: a case-control study of possible causative factors. Acta Derm Venereol 2003;83:189-193.
- Fortina AB, Piaserico S, Caforio AL, Abeni D, Alaibac M, Angelini A, Iliceto S, Peserico A. Immunosuppressive level and other risk factors for basal cell carcinoma and squamous cell carcinoma in heart transplant recipients. Arch Dermatol 2004;140:1079-1085.
- Wikonkal NM, Brash DE. Ultraviolet radiation induced signature mutations in photocarcinogenesis. J Investig Dermatol Symp Proc 1999;4:6-10.
- Brash DE, Rudolph JA, Simon JA, Lin A, McKenna GJ, Baden HP, Halperin AJ, Ponten J. A role for sunlight in skin cancer: UV-induced p53 mutations in squamous cell carcinoma. Proc Natl Acad Sci U S A 1991;88:10124-10128.
- Ziegler A, Jonason AS, Leffell DJ, Simon JA, Sharma HW, Kimmelman J, Remington L, Jacks T, Brash DE. Sunburn and p53 in the onset of skin cancer [see comments]. Nature 1994;372:773-776.
- 24. Zhang W, Remenyik E, Zelterman D, Brash DE, Wikonkal NM. Escaping the stem cell compartment: sustained UVB exposure allows p53-mutant keratinocytes to colonize adjacent epidermal proliferating units without incurring additional mutations. Proc Natl Acad Sci U S A 2001;98:13948-13953.
- Backvall H, Stromberg S, Gustafsson A, Asplund A, Sivertsson A, Lundeberg J, Ponten F. Mutation spectra of epidermal p53 clones adjacent to basal cell carcinoma and squamous cell carcinoma. Exp Dermatol 2004;13:643-650.
- Jonason AS, Kunala S, Price GJ, Restifo RJ, Spinelli HM, Persing JA, Leffell DJ, Tarone RE, Brash DE. Frequent clones of p53-mutated keratinocytes in normal human skin. Proc Natl Acad Sci U S A 1996;93:14025-14029.
- Ren ZP, Hedrum A, Ponten F, Nister M, Ahmadian A, Lundeberg J, Uhlen M, Ponten J. Human epidermal cancer and accompanying precursors have identical p53 mutations different from p53 mutations in adjacent areas of clonally expanded non-neoplastic keratinocytes. Oncogene 1996;12:765-773.
- Tabata H, Nagano T, Ray AJ, Flanagan N, Birch-MacHin MA, Rees JL. Low frequency of genetic change in p53 immunopositive clones in human epidermis. J Invest Dermatol 1999;113:972-976.
- Ponten F, Berg C, Ahmadian A, Ren ZP, Nister M, Lundeberg J, Uhlen M, Ponten J. Molecular pathology in basal cell cancer with p53 as a genetic marker. Oncogene 1997;15:1059-1067.
- 30. Berg RJ, van Kranen HJ, Rebel HG, de Vries A, van Vloten WA, van Kreijl CF, van der Leun JC, de Gruijl FR. Early p53 alterations in mouse skin carcinogenesis by UVB radiation: immunohistochemical detection of mutant p53 protein in clusters of preneoplastic epidermal cells. Proc Natl Acad Sci U S A 1996;93:274-278.
- 31. Rebel H, Mosnier LO, Berg RJ, Westerman-de Vries A, van Steeg H, van Kranen HJ, de Gruijl FR. Early p53-positive foci as indicators of tumor risk in ultraviolet-exposed hairless mice: kinetics of induction, effects of DNA repair deficiency, and p53 heterozygosity. Cancer Res 2001;61:977-983.
- 32. Rebel H, Kram N, Westerman A, Banus S, van Kranen HJ, de Gruijl FR. Relationship between UV-induced mutant p53 patches and skin tumours, analysed by mutation spectra and by induction kinetics in various DNA-repair-deficient mice. Carcinogenesis 2005;26:2123-2130.
- Backvall H, Wolf O, Hermelin H, Weitzberg E, Ponten F. The density of epidermal p53 clones is higher adjacent to squamous cell carcinoma in comparison with basal cell carcinoma. Br J Dermatol 2004;150:259-266.
- Fisher MS, Kripke ML. Systemic alteration induced in mice by ultraviolet light irradiation and its relationship to ultraviolet carcinogenesis. Proc Natl Acad Sci U S A 1977;74:1688-1692.

- Kripke ML. Antigenicity of murine skin tumors induced by ultraviolet light. J Natl Cancer Inst 1974;53:1333-1336.
- Ullrich SE. Mechanisms underlying UV-induced immune suppression. Mutat Res 2005;571:185-205.
- Walterscheid JP, Nghiem DX, Kazimi N, Nutt LK, McConkey DJ, Norval M, Ullrich SE. Cis-urocanic acid, a sunlightinduced immunosuppressive factor, activates immune suppression via the 5-HT2A receptor. Proc Natl Acad Sci U S A 2006;103:17420-17425.
- Schwarz T. Mechanisms of UV-induced immunosuppression. Keio J Med 2005;54:165-171.
- Norval M. The mechanisms and consequences of ultravioletinduced immunosuppression. Prog Biophys Mol Biol 2006;92:108-118.
- Kelly GE, Meikle W, Sheil AG. Scheduled and unscheduled DNA synthesis in epidermal cells of hairless mice treated with immunosuppressive drugs and UVB-UVA irradiation. Br J Dermatol 1987;117:429-440.
- Yarosh DB, Pena AV, Nay SL, Canning MT, Brown DA. Calcineurin inhibitors decrease DNA repair and apoptosis in human keratinocytes following ultraviolet B irradiation. J Invest Dermatol 2005;125:1020-1025.
- Kelly GE, Meikle WD, Moore DE. Enhancement of UVinduced skin carcinogenesis by azathioprine: role of photochemical sensitisation. Photochem Photobiol 1989;49:59-65.
- 43. O'Donovan P, Perrett CM, Zhang X, Montaner B, Xu YZ, Harwood CA, McGregor JM, Walker SL, Hanaoka F, Karran P. Azathioprine and UVA light generate mutagenic oxidative DNA damage. Science 2005;309:1871-1874.
- Hojo M, Morimoto T, Maluccio M, Asano T, Morimoto K, Lagman M, Shimbo T, Suthanthiran M. Cyclosporine induces cancer progression by a cell-autonomous mechanism. Nature 1999;397:530-534.
- 45. Kelly GE, Meikle W, Sheil AG. Effects of immunosuppressive therapy on the induction of skin tumors by ultraviolet irradiation in hairless mice. Transplantation 1987;44:429-434.
- Glover MT, Deeks JJ, Raftery MJ, Cunningham J, Leigh IM. Immunosuppression and risk of non-melanoma skin cancer in renal transplant recipients. Lancet 1997;349:398.
- Dantal J, Hourmant M, Cantarovich D, Giral M, Blancho G, Dreno B, Soulillou JP. Effect of long-term immunosuppression in kidney-graft recipients on cancer incidence: randomised comparison of two cyclosporin regimens. Lancet 1998;351:623-628.
- 48. Campistol JM, Eris J, Oberbauer R, Friend P, Hutchison B, Morales JM, Claesson K, Stallone G, Russ G, Rostaing L, Kreis H, Burke JT, Brault Y, Scarola JA, Neylan JF. Sirolimus therapy after early cyclosporine withdrawal reduces the risk for cancer in adult renal transplantation. J Am Soc Nephrol 2006;17:581-589.
- Mathew T, Kreis H, Friend P. Two-year incidence of malignancy in sirolimus-treated renal transplant recipients: results from five multicenter studies. Clin Transplant 2004;18:446-449.
- Guba M, Graeb C, Jauch KW, Geissler EK. Pro- and anticancer effects of immunosuppressive agents used in organ transplantation. Transplantation 2004;77:1777-1782.
- 51. Struijk L, Bouwes Bavinck JN, Wanningen P, van der ME, Westendorp RG, ter Schegget J, Feltkamp MC. Presence of human papillomavirus DNA in plucked eyebrow hairs is associated with a history of cutaneous squamous cell carcinoma. J Invest Dermatol 2003;121:1531-1535.
- 52. Feltkamp MC, Broer R, di Summa FM, Struijk L, van der ME, Verlaan BP, Westendorp RG, ter Schegget J, Spaan WJ, Bouwes Bavinck JN. Seroreactivity to epidermodysplasia verruciformis-related human papillomavirus types is associated with nonmelanoma skin cancer. Cancer Res 2003;63:2695-2700.
- 53. Karagas MR, Nelson HH, Sehr P, Waterboer T, Stukel TA, Andrew A, Green AC, Bavinck JN, Perry A, Spencer S, Rees JR, Mott LA, Pawlita M. Human papillomavirus infection and incidence of squamous cell and basal cell carcinomas of the skin. J Natl Cancer Inst 2006;98:389-395.
- de Villiers EM, Fauquet C, Broker TR, Bernard HU, zur HH. Classification of papillomaviruses. Virology 2004;324:17-27.

- Berkhout RJ, Bouwes Bavinck JN, ter Schegget J. Persistence of human papillomavirus DNA in benign and (pre)malignant skin lesions from renal transplant recipients. J Clin Microbiol 2000;38:2087-2096.
- 56. de Jong-Tieben LM, Berkhout RJ, Smits HL, Bouwes Bavinck JN, Vermeer BJ, van der Woude FJ, ter Schegget J. High frequency of detection of epidermodysplasia verruciformisassociated human papillomavirus DNA in biopsies from malignant and premalignant skin lesions from renal transplant recipients. J Invest Dermatol 1995;105:367-371.
- 57. Meyer T, Arndt R, Christophers E, Nindl I, Stockfleth E. Importance of human papillomaviruses for the development of skin cancer. Cancer Detect Prev 2001;25:533-547.
- Meyer T, Amdt R, Nindl I, Ulrich C, Christophers E, Stockfleth E. Association of human papillomavirus infections with cutaneous tumors in immunosuppressed patients. Transpl Int 2003;16:146-153.
- 59. Pfister H. Chapter 8: Human papillomavirus and skin cancer. J Natl Cancer Inst Monogr 2003;52-56.
- Jackson S, Harwood C, Thomas M, Banks L, Storey A. Role of Bak in UV-induced apoptosis in skin cancer and abrogation by HPV E6 proteins. Genes Dev 2000;14:3065-3073.
- Jackson S, Storey A. E6 proteins from diverse cutaneous HPV types inhibit apoptosis in response to UV damage. Oncogene 2000;19:592-598.
- Iftner T, Elbel M, Schopp B, Hiller T, Loizou JI, Caldecott KW, Stubenrauch F. Interference of papillomavirus E6 protein with single-strand break repair by interaction with XRCC1. EMBO J 2002;21:4741-4748.
- Giampieri S, Storey A. Repair of UV-induced thymine dimers is compromised in cells expressing the E6 protein from human papillomaviruses types 5 and 18. Br J Cancer 2004;90:2203-2209.
- Bouwes Bavinck JN, Feltkamp MC. Milk of human kindness?--HAMLET, human papillomavirus, and warts. N Engl J Med 2004;350:2639-2642.
- Schaper ID, Marcuzzi GP, Weissenborn SJ, Kasper HU, Dries V, Smyth N, Fuchs P, Pfister H. Development of skin tumors in mice transgenic for early genes of human papillomavirus type 8. Cancer Res 2005;65:1394-1400.
- DiGiovanna JJ. Posttransplantation skin cancer: scope of the problem, management, and role for systemic retinoid chemoprevention. Transplant Proc 1998;30:2771-2775.
- Stasko T, Brown MD, Carucci JA, Euvrard S, Johnson TM, Sengelmann RD, Stockfleth E, Tope WD. Guidelines for the management of squamous cell carcinoma in organ transplant recipients. Dermatol Surg 2004;30:642-650.
- Altucci L, Gronemeyer H. The promise of retinoids to fight against cancer. Nat Rev Cancer 2001;1:181-193.
- Orfanos CE, Zouboulis CC, Almond-Roesler B, Geilen CC. Current use and future potential role of retinoids in dermatology. Drugs 1997;53:358-388.
- DiGiovanna JJ. Retinoid chemoprevention in patients at high risk for skin cancer. Med Pediatr Oncol 2001;36:564-567.
- Euvrard S, Kanitakis J, Thivolet J, Claudy A. Retinoids for the management of dermatological complications of organ transplantation. Biodrugs. 8[3], 176-184. 1997. Adis international Limited.
- Morton CA, Brown SB, Collins S, Ibbotson S, Jenkinson H, Kurwa H, Langmack K, McKenna K, Moseley H, Pearse AD, Stringer M, Taylor DK, Wong G, Rhodes LE. Guidelines for topical photodynamic therapy: report of a workshop of the British Photodermatology Group. Br J Dermatol 2002:146:552-567.
- 73. Dragieva G, Prinz BM, Hafner J, Dummer R, Burg G, Binswanger U, Kempf W. A randomized controlled clinical trial of topical photodynamic therapy with methyl aminolaevulinate in the treatment of actinic keratoses in transplant recipients. Br J Dermatol 2004;151:196-200.
- 74. Dragieva G, Hafner J, Dummer R, Schmid-Grendelmeier P, Roos M, Prinz BM, Burg G, Binswanger U, Kempf W. Topical photodynamic therapy in the treatment of actinic keratoses and Bowen's disease in transplant recipients. Transplantation 2004;77:115-121.
- 75. Stender IM, Bech-Thomsen N, Poulsen T, Wulf HC. Photodynamic therapy with topical delta-aminolevulinic acid delays

UV photocarcinogenesis in hairless mice. Photochem Photobiol 1997;66:493-496.

- Sharfaei S, Juzenas P, Moan J, Bissonnette R. Weekly topical application of methyl aminolevulinate followed by light exposure delays the appearance of UV-induced skin tumours in mice. Arch Dermatol Res 2002;294:237-242.
- Otley CC, Berg D, Ulrich C, Stasko T, Murphy GM, Salasche SJ, Christenson LJ, Sengelmann R, Loss GE, Jr., Garces J. Reduction of immunosuppression for transplant-associated skin cancer: expert consensus survey. Br J Dermatol 2006;154:395-400.
- 78. Jemec GB, Holm EA. Nonmelanoma skin cancer in organ transplant patients. Transplantation 2003;75:253-257.
- van Zuuren EJ, Posma AN, Scholtens RE, Vermeer BJ, van der Woude FJ, Bouwes Bavinck JN. Resurfacing the back of the hand as treatment and prevention of multiple skin cancers in kidney transplant recipients. J Am Acad Dermatol 1994;31:760-764.
- Euvrard S, Kanitakis J, Claudy A. Skin cancers after organ transplantation. N Engl J Med 2003;348:1681-1691.
- Reymann F. Treatment of multiple squamous cell carcinomas of the skin in an immunosuppressed patient. Dermatologica 1981;162:304-306.
- Clayton AS, Stasko T. Treatment of nonmelanoma skin cancer in organ transplant recipients: review of responses to a survey. J Am Acad Dermatol 2003;49:413-416.

CHAPTER 2

More epidermal p53 patches adjacent to skin carcinomas in renal-transplant recipients than in immunocompetent patients: the role of azathioprine

Experimental Dermatology, in press

More epidermal p53 patches adjacent to skin carcinomas in renal transplant recipients than in immunocompetent patients: the role of azathioprine

Ymke G. L. de Graaf¹, Heggert Rebel¹, Abdoel Elghalbzouri¹, Patricia Cramers², Ruud G. L. Nellen¹*, Rein Willemze¹, Jan Nico Bouwes Bavinck¹ and Frank R. de Gruijl¹

Departments of ¹Dermatology and ²Toxicogenetics, Leiden University Medical Center, Leiden, The Netherlands *Correspondence:* Y. G. L. de Graaf, MD, Department of Dermatology, Leiden University Medical Center, PO Box 9600, 2300 RC Leiden, The Netherlands, Tel.: +31 71 5262638, Fax: +31 71 5248106, e-mail: y.g.l.de_graaf@lumc.nl **Current address: Department of Dermatology, Academic Hospital Maastricht, Maastricht, The Netherlands.*

Accepted for publication 18 September 2007

Abstract: Immunosuppressive medication in renal transplant recipients (RTR) strongly increases the risk of cancers on sunexposed skin. This increased risk was considered an inevitable collateral effect of immunosuppression, because UV-induced carcinomas in mice were found to be highly antigenic. Here, we posed the question whether immunosuppression also increases the frequency of p53-mutant foci ('p53 patches'), putative microscopic precursors of squamous cell carcinomas. As the majority of RTR was kept on azathioprine for most of the time, we investigated whether this drug could increase UV-induced p53 patches by immunosuppression. As azathioprine can impair UV-damaged DNA repair under certain conditions, we also investigated whether DNA repair was affected. Archive material of RTR and immunocompetent patients (ICP), as well as azathioprine-administered hairless mice were examined for p53 patches. DNA repair was investigated by ascertaining the effect of azathioprine on unscheduled DNA synthesis (UDS) in UVirradiated human keratinocytes. P53 patches were more prevalent in RTR than in ICP in normal skin adjacent to carcinomas (P = 0.02), in spite of a lower mean age in the RTR (52 vs 63 years, P = 0.001), but we found no increase in UV-induced p53 patches in mice that were immunosuppressed by azathioprine. We found a significant reduction in DNA repair activity in keratinocytes treated with azathioprine (P = 0.011). UV-induced UDS in humans is dominated by repair of cyclobutane pyrimidine dimers, and these DNA lesions can lead to 'UV-signature' mutations in the *P53* gene, giving rise to p53 patches.

Key words: azathioprine – DNA repair – p53 patches – renal transplant recipients

Please cite this paper as: More epidermal p53 patches adjacent to skin carcinomas in renal transplant recipients than in immunocompetent patients: the role of azathioprine. Experimental Dermatology 2007.

Introduction

Renal transplant recipients (RTR) are at an increased risk of developing skin cancer, of which squamous cell carcinomas (SCCs) are the most prevalent. These tumors develop primarily in areas exposed to the sun (1,2). The incidence of skin carcinomas in these patients increases with time after transplantation, reaching 40% in 20 years in the Netherlands (3) and in 10 years in Australia (4).

The pathogenesis of skin carcinoma is multifactorial. Solar ultraviolet radiation is recognized as a dominant etiological factor (5,6). Especially, the short wavelength (280–315 nm) UVB radiation induces DNA lesions, broad aspecific detection which by XL-PCR shows efficient repair at sub-lethal dosages in human keratinocytes, i.e. about 90% of the lesions are removed in 24 h (7). During chronic UV irradiation, clusters of epidermal cells develop that over-express the p53 protein in mutant conformation. These p53 foci or 'p53 patches' (p53-mutant clones) are detectable long before the appearance of skin carcinomas in the hairless mouse model (8), and are also found in human skin (9,10). P53 patches and SCCs show parallel UV dose-time dependencies in mice (11). As the p53 patches bear UV-specific mutations similar to those in the subsequent SCCs, they appear to be early microscopic precursor lesions of the ultimate tumors (12). Hence, the frequency of these patches can serve as a good marker of SCC risk (11,12).

Another important risk factor for skin cancer is immunosuppression. Classic animal experiments (13,14) have shown UV-induced skin tumors to be immunogenic, i.e. the tumors will be rejected upon transplantation into syngenic host, unless the host is immunosuppressed (UV radiation itself was found to induce a immunosuppressed and tumor-tolerant state). Suppression of tumor immunity facilitates UV-induced skin carcinogenesis (15). Thus, the increased risk of skin carcinomas in RTR would appear to be an inevitable consequence of the immunosuppressive medication.

Most of the early RTR started off on the immunosuppressant azathioprine (supplemented with prednisone), and cyclosporine made its entry later on as an alternative immunosuppressant. Experiments showed that these immunosuppressants accelerated UV carcinogenesis in the hairless mouse model (16). Besides causing immunosuppression, these drugs were reported to impair DNA repair in the hairless mice (17). The repair in the epidermis was measured by unscheduled DNA synthesis (UDS). This raises the question of whether RTR could suffer from a medicinally induced DNA repair syndrome, which would make it a far more common syndrome than any hereditary syndrome of DNA instability/mutation. This in turn would make the RTR an especially interesting group of patients for basic studies to further our understanding of (skin) cancerogenesis (18).

Here, we first of all posed the question whether the enhanced risk of SCCs in RTR is reflected in increases in p53 patches in their skin. To this end, we investigated in archive material whether p53 patches were more prevalent in normal skin adjacent to skin carcinomas excised from RTR when compared with immunocompetent patients (ICP). After finding confirmative data, we pursued to identify the underlying mechanism.

As the archive material stemmed from patients who were predominantly and for the longest period of time kept on azathioprine, we focused on this immunosuppressant. We investigated two potential mechanism by which azathioprine could cause an increase in p53 patches: (i) immunosuppression, (ii) impaired DNA repair. We resorted to the hairless mouse model to assess experimentally the effect of an azathioprine-immunosuppressive regimen on the UV induction of p53 patches. In supplementation of earlier experiments in mice, we investigated the impact of azathioprine on the repair of UV-induced DNA damage in human primary keratinocytes. The results suggest that the local adverse effects of azathioprine on DNA repair in human skin increase the induction of p53 patches, and may thus independently of immunosuppression - add to the risk of developing SCC in RTR.

Methods

Patients; selection of skin samples

Archived paraffin blocks from surgical excisions of skin carcinomas, of which the majority consisted of SCC, and the adjacent excision margins were obtained from both RTR and ICP. The adjacent skin margins had a minimal distance of 2 mm from the tumor mass. All skin samples were obtained from chronically sun-exposed sites (head, neck, dorsal surface of hands). Nineteen RTR and 13 ICP were randomly selected and included, matched for location of the tumor, and season of excision. Most skin samples were taken in autumn/winter.

Immunohistochemistry and scoring of p53 patches in human skin

P53 immune staining with DO-7 monoclonal antibody (M7001; Dakocytomation, Copenhagen, Denmark) was performed using standard procedures as described previously (10). Sections of skin tumors known to have strong p53 immunoreactivity with DO-7 were included as positive controls. Omission of the first antibody always yielded a negative result.

For description of the p53 immunoreactivity, criteria of Ren et al. were used (10). A p53 patch was defined as an uninterrupted cluster of at least 10 strongly and uniformly immunopositive nuclei in a sharply demarcated area of normal epidermis. Only these 'compact patterns' in the excision margins were scored, as this staining pattern was strongly associated with p53 mutations (19). The number of p53 patches per cm in the normal skin margins adjacent to carcinomas was determined in archive material from RTR and ICP. P53 patches were counted if there was no sign of connection to tumor in 10 successive sections. We also scored the size of the p53 patches in both groups.

Mice: UV irradiation and azathioprine treatment

Three groups of five hairless SKH-1 mice (Charles River, Maastricht, The Netherlands) entered the experiment at 9 weeks of age, under conditions as described earlier (11). Mice in the first group were both irradiated with UV and administered azathioprine, the second group was also irradiated, but received a placebo. Mice of the third group were not irradiated but did receive azathioprine. The procedure for UV irradiation and azathioprine administration was comparable to that described earlier (16) for the experiments on azathioprine-enhanced UV carcinogenesis. The mice were irradiated on working days: the first 2 weeks with 0.75 of the minimal erythemal dose (MED, 375 J/m^2 UV) per day, and in the third and fourth week with 1 MED (500 J/m² UV) per day from TL-12 lamps (Philips, Eindhoven, The Netherlands). Azathioprine (Pharmachemie, Haarlem, The Netherlands) was diluted in phosphatebuffered saline (PBS) at a concentration of 4 mg/ml. On Mondays, Wednesdays and Fridays, during the 4 weeks of irradiation, the mice were injected intra-peritoneally with an individual weight-corrected volume of the azathioprine solution resulting in 15 µg/g body weight. PBS injections served as placebo treatment. At 24 h after the final UV irradiation, all mice were killed by CO₂ asphyxiation. From

each mouse a defined rectangular dorsal part of the skin $(2.9 \times 1.9 \text{ cm})$ was dissected for preparation of epidermal sheets. Immunosuppression in the azathioprine-treated groups was confirmed by lymphocyte transformation tests on isolated splenocytes (P = 0.034 and 0.008 for the UV-exposed and unexposed groups, respectively, when compared with the control group that was not treated with azathioprine).

Immunohistochemistry and scoring of p53 patches in mouse skin

Preparation of epidermal sheets and immunostaining with the mutant-p53-specific PAb-240 antibody were described earlier (11). For scoring the p53 patches a grid, placed on top of each epidermal sheet preparation, was used to count p53 patches in 20 squares (total area 29.0×18.5 mm), using a light microscope equipped with a PI ×25/0.5 objective. A p53 patch was defined as a cluster of at least 10 Pab240-positive epidermal cells.

Unscheduled DNA synthesis in human keratinocytes

Primary cultures of normal human keratinocytes (PHKs) were established from skin derived from breast reduction according to earlier described procedures (20,21). PHKs were seeded in 10 cm diameter culture dishes at a density of 0.09×10^6 /cm². PHKs from two different donors were used for two independent UDS tests.

The UDS test was performed according to van Zeeland et al. (22). At the first and third day of culture, ³²P was added to the medium to label the PHK DNA overall. At the sixth day the medium was replaced with fresh medium supplemented with a series of azathioprine concentrations. Two independent experiments were performed, with azathioprine concentrations of 0, 5, 25 and 100 μ M in the first and 0, 10 and 50 μ M in the second experiment. Per concentration two or three dishes of keratinocytes were tested.

At the seventh day the PHK cells were rinsed with PBS and irradiated with 300 J/m² from TL-12 lamps. Subsequently the media with or without azathioprine were returned on the cells and cultured for 6 h with ³H-thymidine, after which the cells were harvested. ³H uptake and ³²P were measured by differentiated scintillation counting of alkaline gradient fractions as described (22), and used for ratio calculations of UDS and total DNA, respectively.

Parallel cultures of human keratinocytes on glass cover slips were used for assessment of the vitality of the cells that were subjected to 0, 5, 25 and 100 μ M azathioprine (two cover slips per concentration). After 7 days of culturing, slides were rinsed with PBS and stained with 0.15% trypan blue. Vital and non-vital cells were counted *in duplo* by light microscopy.

Statistical analyses

Because of high percentages of individuals without p53 patches and some RTR with exceptionally high numbers of p53 patches, the difference in the distributions of p53 patches among RTR and ICP (Fig. 2) was tested by chisquared statistics (as differences in Kaplan-Meier curves, calculated by Graphpath Prism 3.0 software). For further statistical analyses, we used SPSS version 12.0.1 for Windows. The Student's t-test was used to ascertain significances of the differences in age and in UDS measurements. A log t-test was used to test the difference in number of p53 patches in the mice. To calculate the difference in sizes of p53 patches between RTR and ICP, the Mann-Whitney U-test was performed. The density of p53 patches in humans was related to other factors such as age and the period of time after transplantation and tested on significance in linear regression analyses.

Results

The baseline characteristics of the RTR and ICP are listed in Tables 1 and 2. At the time of excision, the RTR were significantly younger than the ICP, with a mean age of 52 and 63 years old, respectively (P = 0.001); difference with [95% CI]: 11 [5–18].

P53 patches in human skin

Figure 1 shows an example of a p53 patch in a part of the epidermis. The number of p53 patches per cm epidermis adjacent to skin carcinomas was significantly higher in RTR; median of 1.4 vs 0.3 patches/cm in RTR and ICP, respectively (P = 0.02). Figure 2 shows a clear difference between the groups in the distributions of p53 patches, with 20% (n = 4) of the RTR and none of the ICP with more than 3 patches/cm. The sizes of the patches did not differ between the RTR and ICP: in both groups predominantly small patches (10–50 cells) were found (data not shown).

The number of patches was not associated with age, gender or season in either group or both groups combined. Additionally, no association was found between the number of p53 patches and the time since transplantation.

The majority of the RTR (16/19) used azathioprine, but exclusion of the three patients that used cyclosporine and/or mycophenolate mofetil did not alter the results.

UV-induced p53 patches in azathioprineimmunosuppressed mice

A slight erythema was found in some mice after increasing the daily UV dose after 2 weeks, but most of the mice showed no apparent sunburn skin reaction, both in the azathioprine-treated and non-treated groups. Table 1. Characteristics of renal transplant recipients ranked according to p53 patches/cm epidermis

Patient no.	Age (years)	Time after TX (years)	Sex	Type of medication	Type of skin cancer	Location of skin cancer	Season	P53 patches/cm epidermis
1	53	10	М	P+A	SCC	Neck	Spring	9.0
2	54	17	F	P+A	BCC	Dorsum of forearm	Autumn	6.7
3	60	21	Μ	P+A	SCC	Nose	Winter	4.6
4	55	11	F	P+C	SCC	Dorsum of hand	Autumn	4.2
5	48	19	F	P+A	SCC	Dorsum of finger	Autumn	2.5
6	54	18	F	P+A	SCC	Cheek	Winter	2.1
7	41	9	М	P+A	SCC	Shoulder	Autumn	2.0
8	43	16	F	P+A	SCC	Dorsum of forearm	Summer	1.9
9	46	24	F	P+A	SCC	Dorsum of forearm	Spring	1.7
10	51	22	F	P+A	SCC	Dorsum of hand	Winter	1.4
11	55	20	F	P+A	SCC	Dorsum of finger	Winter	0.9
12	49	7	М	P+C+M	SCC	Forehead	Autumn	0.8
13	55	23	Μ	P+A	SCC	Dorsum of hand	Summer	0.7
14	37	9	F	P+C	SCC	Dorsum of hand	Winter	0.1
15	69	25	М	P+A	SCC	Forehead	Summer	0.0
16	45	25	Μ	P+A	SCC	Dorsum of hand	Winter	0.0
17	64	17	М	P+A	SCC	Forehead	Winter	0.0
18	61	27	F	P+A	SCC	Cheek	Winter	0.0
19	49	18	F	P+A	SCC	Dorsum of forearm	Autumn	0.0

M, male; F, female; P, prednisone; A, azathioprine; C, cyclosporine; M, mycophenolate mofetil; BCC, basal cell carcinoma; SCC, squamous cell carcinoma; TX, transplantation.

Table 2. Characteristics of immunocompetent pa	atients ranked according to	p53 patches/	'cm epidermis
---	-----------------------------	--------------	---------------

Patient no.	Age (years)	AgeType ofLocation of skin(years)Sexskin cancercancer		Location of skin cancer	Season	P53 patches/cm epidermis	
1	77	F	SCC	Dorsum of hand	Autump	2.2	
2	56	M	SCC	Dorsum of forearm	Winter	1.6	
3	49	M	SCC	Scalp	Winter	1.1	
4	66	M	SCC	Forehead	Spring	0.9	
5	69	F	SCC	Dorsum of hand	Summer	0.8	
6	72	F	SCC	Nose	Autumn	0.7	
7	64	М	SCC	Cheek	Spring	0.3	
8	59	М	SCC	Forehead	Winter	0.0	
9	69	F	SCC	Upper arm	Spring	0.0	
10	69	М	SCC	Forehead	Winter	0.0	
11	72	М	SCC	Face	Summer	0.0	
12	34	М	SCC	Neck	Summer	0.0	
13	68	F	SCC	Cheek	Winter	0.0	

M, male; F, female; BCC, basal cell carcinoma; SCC, squamous cell carcinoma.

No p53 patches were detected in the epidermal sheets from the azathioprine-treated control mice that were not UV irradiated. Clusters of epidermal cells with p53-positive nuclei (example in Fig. 3) were found in the two groups that were UV exposed for 4 weeks. The median numbers of the p53 patches per epidermal sheet were not significantly different between the two groups of UVB-irradiated mice. The median numbers of p53 patches [95% CI] in the azathioprine-treated mice and non-treated mice were 11 [7–17] and 24 [13–46], respectively (P = 0.11).

UDS in human keratinocytes

To investigate the effect of azathioprine on primary human keratinocytes (PHKs), we assessed the DNA repair by UDS assay, 6 h after UV irradiation. At the day of radiation the cultures were 90-100% confluent. Overall, UDS was



Figure 1. p53 patch in 5 μ m section of human epidermis from a renal transplant recipient.



Figure 2. Plot of the percentages of patients (on ordinate) with densities of p53 patches greater than a density X (on abscissa); a significant difference (P = 0.02, chi-squared test) between the distributions from a renal transplant recipient (RTR; dashed line) and a immunocompetent patient (ICP; solid line).



Figure 3. p53 patch in murine epidermal sheet; bottom view (circular structures are hair follicles), insert with detail of compact cluster of cells with p53-positive nuclei.

significantly inhibited when azathioprine (5–100 μ M) was present in the medium in comparison with controls (0 μ M, P = 0.011), with an apparent maximum inhibition around



Figure 4. Effect of azathioprine on unscheduled DNA synthesis (UDS) in human keratinocytes (whiskers represent SE).

10 μ M (Fig. 4). The trypan blue test revealed >95% vitality at several tested azathioprine concentrations (5, 25 and 100 μ M).

Discussion

In the present study we found significantly more p53 patches in uninvolved skin neighboring carcinomas of RTR than of ICP. In contrast to this observation in humans, we found no increase in p53 patches in chronically UV-exposed mice that were treated with azathioprine. As we confirmed the immunosuppression of azathioprine in these mice, we conclude that p53 clones do not appear to be immunoreactive. This finding is in line with the results of Remenvik et al. (23) who found no differences in the induction and regression of p53 clones between immunosuppressed RAG-1 knockout and wild-type mice. Although azathioprine did not increase the number of p53 patches in our mice, it did increase the number of UV-induced skin tumors in mice in earlier experiments (16). Hence, azathioprine-induced immunosuppression did appear to affect the ultimate development of SCC. Thus the p53 patch increase in RTR and not in mice on azathioprine is not likely to be related to immunosuppression, but by another action of azathioprine.

Impairment of DNA repair is another effect of azathioprine and we found a diminished repair of UV-induced DNA damage in PHKs that were treated with azathioprine. This is in line with earlier published results of UV-induced DNA repair inhibition by azathioprine in peripheral blood mononuclear cells (24). We did not find a simple monotonous increase in inhibition with increasing azathioprine concentration, but a maximum inhibition around 10 μ M. The inhibition is most likely specific inhibition, as no toxicity was measured up to 100 μ M. Based on dosages per kg body weight and by the metabolites (6-mercaptopurin and 6-thiouric acid) in circulation, we estimated about 7.0 μ M azathioprine present in a patient (25), i.e. within the range that was tested in our experiments.

Unscheduled DNA synthesis is impaired by azathioprine in both mice and men (17,24); thus one would expect an increase in p53 mutations and patch formation as observed in men. However, this increase is not clear in the well-controlled experiment in mice. A plausible explanation is the difference in DNA repair between mouse and man: the cyclobutane pyrimidine dimer is the dominant carcinogenic DNA damage (26), which can cause the typical 'UV-signature' mutations in the P53 gene (27), and this damage is poorly repaired in mice and very well repaired in men (28-30). Consequently, DNA repair impairment is likely to have more of an impact in men than in mice. The presence of p53 patches in human skin may, therefore, be attributable to a local effect on DNA repair in human keratinocytes rather than to a systemic impairment of immune surveillance and elimination.

Next to an impairment of DNA repair, azathioprine can introduce a UVA phototoxicity from thio-guanines incorporated in DNA (31,32). However, the mutation spectrum of *P53* in carcinomas from early cohorts of RTR showed no apparent deviation from the expected UVB-related point mutations normally found in ICP (33). This result provides evidence for an enhanced UVB-related mutation rate from lowered DNA repair in RTR rather than for additional mutations from UVA sensitization. The number of mutations (12) may however been too small to pick up an added effect from UVA sensitization.

In general, p53 patches can serve as a marker of skin carcinoma risk in humans. Next to the evidence from animal experiments, a significant dose–response relationship between UV exposure and frequency of p53 clones in human skin was also reported (9). Additionally, Backvall et al. found significantly more p53 clones adjacent to SCC than to basal cell carcinomas and melanocytic nevi (34). Another study reported significantly more p53 patches adjacent to basal cell carcinomas compared with benign skin lesions (35).

Earlier experimental and some clinical studies have shown that the frequency of p53 patches increases with age (11,36,37). In contrast, Jonason et al. did not find such an association (9). In the present study we did not find an association between age and number of patches either. This may, however, be due to the relatively small number of patients that was included.

Female patients appeared somewhat overrepresented among the RTR when compared with the ICP, but overall, the number of patches showed no dependence on gender. On close scrutiny, there appears to be a difference between both groups regarding location of the tumors. In the RTR the dorsum of the fingers and hands, or forearms occurs more frequent than in the ICP. Although there probably is no difference in the level of sun exposure, it cannot be entirely excluded that this site difference introduced a bias in p53 patches; a larger study with substratification to tumor site would be required.

In conclusion, in our archive material we found a higher density of p53 patches in RTR than in ICP. P53 patches in mice were not subject to immune detection and elimination (23). However, we found that azathioprine lowered DNA repair in human keratinocytes, and may thus increase mutagenesis in the *P53* gene, and consequently the development of p53 patches in human. Whether, aside from the systemic immunosuppression, this impact of azathioprine on the keratinocytes in the skin can ultimately increase the risk of skin carcinomas in RTR requires further (*in vivo*) experiments.

Because p53 patches represent an early step in skin carcinogenesis, the impact of novel immunosuppressive agents, such as sirolimus, on DNA repair and the formation of p53 patches in human should also be studied to ascertain whether these novel agents lack this additional risk.

Acknowledgements

We would like to thank Marjolein Lauwen for assistance with lymphocyte transformation test and Ronald Filon for assistance with the UDS test. We also thank the animal caretakers for assistance and care for the hairless mice.

References

- Euvrard S, Kanitakis J, Claudy A. Skin cancers after organ transplantation. N Engl J Med 2003: 348: 1681–1691.
- 2 London N J, Farmery S M, Will E J, Davison A M, Lodge J P. Risk of neoplasia in renal transplant patients. Lancet 1995: 346: 403–406.
- 3 Hartevelt M M, Bavinck J N, Kootte A M, Vermeer B J, Vandenbroucke J P. Incidence of skin cancer after renal transplantation in The Netherlands. Transplantation 1990: 49: 506–509.
- 4 Bouwes Bavinck J N, Hardie D R, Green A et al. The risk of skin cancer in renal transplant recipients in Queensland, Australia. A follow-up study. Transplantation 1996: 61: 715–721.
- 5 Brash D E, Rudolph J A, Simon J A et al. A role for sunlight in skin cancer: UV-induced p53 mutations in squamous cell carcinoma. Proc Natl Acad Sci U S A 1991: 88: 10124–10128.
- 6 Ziegler A, Jonason A S, Leffell D J *et al.* Sunburn and p53 in the onset of skin cancer [see comments]. Nature 1994: **372**: 773–776.
- 7 Lisby S, Gniadecki R, Wulf H C. UV-induced DNA damage in human keratinocytes: quantitation and correlation with long-term survival. Exp Dermatol 2005: 14: 349–355.
- 8 Berg R J, van Kranen H J, Rebel H G et al. Early p53 alterations in mouse skin carcinogenesis by UVB radiation: immunohistochemical detection of mutant p53 protein in clusters of preneoplastic epidermal cells. Proc Natl Acad Sci U S A 1996: 93: 274–278.
- 9 Jonason A S, Kunala S, Price G J et al. Frequent clones of p53mutated keratinocytes in normal human skin. Proc Natl Acad Sci U S A 1996: 93: 14025–14029.
- 10 Ren Z P, Ponten F, Nister M, Ponten J. Two distinct p53 immunohistochemical patterns in human squamous-cell skin cancer, precursors and normal epidermis. Int J Cancer 1996: 69: 174–179.

- 11 Rebel H, Mosnier L O, Berg R J et al. Early p53-positive foci as indicators of tumor risk in ultraviolet-exposed hairless mice: kinetics of induction, effects of DNA repair deficiency, and p53 heterozygosity. Cancer Res 2001: 61: 977–983.
- Rebel H, Kram N, Westerman A, Banus S, van Kranen H J, de Gruijl F R. Relationship between UV-induced mutant p53 patches and skin tumours, analysed by mutation spectra and by induction kinetics in various DNA-repair-deficient mice. Carcinogenesis 2005: 26: 2123–2130.
- **13** Fisher M S, Kripke M L. Systemic alteration induced in mice by ultraviolet light irradiation and its relationship to ultraviolet carcinogenesis. Proc Natl Acad Sci U S A 1977: **74**: 1688–1692.
- 14 Kripke M L. Antigenicity of murine skin tumors induced by ultraviolet light. J Natl Cancer Inst 1974: 53: 1333–1336.
- 15 Fisher M S, Kripke M L. Suppressor T lymphocytes control the development of primary skin cancers in ultraviolet-irradiated mice. Science 1982: 216: 1133–1134.
- **16** Kelly G E, Meikle W, Sheil A G. Effects of immunosuppressive therapy on the induction of skin tumors by ultraviolet irradiation in hairless mice. Transplantation 1987: **44**: 429–434.
- 17 Kelly G E, Meikle W, Sheil A G. Scheduled and unscheduled DNA synthesis in epidermal cells of hairless mice treated with immuno-suppressive drugs and UVB-UVA irradiation. Br J Dermatol 1987: 117: 429–440.
- 18 Thoms K M, Kuschal C, Emmert S. Lessons learned from DNA repair defective syndromes. Exp Dermatol 2007: 16: 532–544.
- 19 Ren Z P, Hedrum A, Ponten F *et al.* Human epidermal cancer and accompanying precursors have identical p53 mutations different from p53 mutations in adjacent areas of clonally expanded non-neoplastic keratinocytes. Oncogene 1996: 12: 765–773.
- 20 El Ghalbzouri A, Hensbergen P, Gibbs S, Kempenaar J, van der S R, Ponec M. Fibroblasts facilitate re-epithelialization in wounded human skin equivalents. Lab Invest 2004: 84: 102–112.
- 21 Ponec M, Kempenaar J A, De Kloet E R. Corticoids and cultured human epidermal keratinocytes: specific intracellular binding and clinical efficacy. J Invest Dermatol 1981: 76: 211–214.
- 22 van Zeeland A A, Smith C A, Hanawalt P C. Sensitive determination of pyrimidine dimers in DNA of UV-irradiated mammalian cells. Introduction of T4 endonuclease V into frozen and thawed cells. Mutat Res 1981: 82: 173–189.
- 23 Remenyik E, Wikonkal N M, Zhang W, Paliwal V, Brash D E. Antigen-specific immunity does not mediate acute regression of UVB-induced p53-mutant clones. Oncogene 2003: 22: 6369–6376.

- 24 Herman M, Weinstein T, Korzets A *et al.* Effect of cyclosporin A on DNA repair and cancer incidence in kidney transplant recipients. J Lab Clin Med 2001: **137**: 14–20.
- **25** Chan G L, Erdmann G R, Gruber S A, Matas A J, Canafax D M. Azathioprine metabolism: pharmacokinetics of 6-mercaptopurine, 6-thiouric acid and 6-thioguanine nucleotides in renal transplant patients. J Clin Pharmacol 1990: **30**: 358–363.
- **26** Jans J, Schul W, Sert Y G *et al.* Powerful skin cancer protection by a CPD-photolyase transgene. Curr Biol 2005: **15**: 105–115.
- **27** Benjamin C L, Ananthaswamy H N. p53 and the pathogenesis of skin cancer. Toxicol Appl Pharmacol 2006: in press.
- 28 Ruven H J, Berg R J, Seelen C M et al. Ultraviolet-induced cyclobutane pyrimidine dimers are selectively removed from transcriptionally active genes in the epidermis of the hairless mouse. Cancer Res 1993: 53: 1642–1645.
- 29 Tang J Y, Hwang B J, Ford J M, Hanawalt P C, Chu G. Xeroderma pigmentosum p48 gene enhances global genomic repair and suppresses UV-induced mutagenesis. Mol Cell 2000: 5: 737–744.
- **30** Alekseev S, Kool H, Rebel H *et al.* Enhanced DDB2 expression protects mice from carcinogenic effects of chronic UV-B irradiation. Cancer Res 2005: **65**: 10298–10306.
- 31 Kelly G E, Meikle W D, Moore D E. Enhancement of UV-induced skin carcinogenesis by azathioprine: role of photochemical sensitisation. Photochem Photobiol 1989: 49: 59–65.
- 32 O'Donovan P, Perrett C M, Zhang X et al. Azathioprine and UVA light generate mutagenic oxidative DNA damage. Science 2005: 309: 1871–1874.
- 33 McGregor JM, Berkhout RJ, Rozycka M et al. p53 mutations implicate sunlight in post-transplant skin cancer irrespective of human papillomavirus status. Oncogene 1997: 15: 1737–1740.
- 34 Backvall H, Wolf O, Hermelin H, Weitzberg E, Ponten F. The density of epidermal p53 clones is higher adjacent to squamous cell carcinoma in comparison with basal cell carcinoma. Br J Dermatol 2004: 150: 259–266.
- 35 Tabata H, Nagano T, Ray A J, Flanagan N, Birch-MacHin M A, Rees J L. Low frequency of genetic change in p53 immunopositive clones in human epidermis. J Invest Dermatol 1999: 113: 972–976.
- 36 Zhang W, Remenyik E, Zelterman D, Brash D E, Wikonkal N M. Escaping the stem cell compartment: sustained UVB exposure allows p53-mutant keratinocytes to colonize adjacent epidermal proliferating units without incurring additional mutations. Proc Natl Acad Sci U S A 2001: 98: 13948–13953.
- **37** Backvall H, Stromberg S, Gustafsson A *et al.* Mutation spectra of epidermal p53 clones adjacent to basal cell carcinoma and squamous cell carcinoma. Exp Dermatol 2004: **13**: 643–650.

CHAPTER 3

P53-specific serum antibodies are not associated with a history of skin carcinoma in renal-transplant recipients and immunocompetent individuals

Journal of Dermatological Science 2005; 38: 228-230





www.intl.elsevierhealth.com/journals/jods

LETTER TO THE EDITOR

p53-Specific serum antibodies are not associated with a history of skin carcinoma in renal transplant recipients and immunocompetent individuals

KEYWORDS

p53-Specific antibodies; Tumor marker; Skin cancer

Alteration of the p53 tumor suppressor gene is the most frequent genetic event found in human cancers [1]. A recent review reported a correlation between the presence of a p53-specific antibody response and p53 mutations in the tumors [1]. Titers of p53-specific antibodies have shown to be prognostic markers of lowered survival in patients with breast and colorectal cancers and may serve to detect lung cancer in early stages [1].

A high rate of p53 mutations is observed in skin carcinomas: in more than 90% of the squamous cell carcinomas (SCC) and in 75–80% of the precursor lesions, solar keratoses (SK) [2]. Also, in renal transplant recipients a large portion (48%) of the skin carcinomas contains p53 mutations [3].

The relevance of p53-specific antibodies to skin cancer is unclear [4]. To our knowledge, only one study reported on p53-specific antibodies in patients with skin carcinomas. Moch et al. [5] found a low prevalence of p53-specific serum antibodies in 105 immunocompetent patients with nonmelanoma skin cancer. Eight per cent (2/25) of the SCC patients and 1.5% (1/68) of the basal cell carcinoma (BCC) patients showed a p53-specific antibody response [5].

Renal transplant recipients run an increased risk of developing cutaneous SCCs compared to immunocompetent patients [6]. In the general population the ratio BCC over SCC is approximately 5:1, this ratio is reversed in renal transplant recipients [6]. Furthermore, SCCs in renal transplant recipients tend to be more aggressive. Therefore, it is important to detect skin cancer in an early stage in this specific population [6].

To determine whether p53-specific serum antibodies could be a relevant marker for the development of skin carcinomas in renal transplant recipients, we resorted to assess p53-specific antibodies in available archived sera (stored at -80 °C) from a well-documented bank collected from renal transplant recipients and immunocompetent subjects with and without a history of skin carcinoma. Data about age, sex and tumors after transplantation were collected. All renal transplant recipients were treated with prednisone and azathioprine [7]. The medical ethical committee of the LUMC approved the study.

In total, sera from 157 patients of the Leiden University Medical Center (LUMC) were studied: thirty-four renal transplant recipients with a history of one or more skin carcinomas, and 43 without skin carcinoma (collected in 1988) [7]. Among the patients who had a carcinoma, 17 individuals had a history of SCC, 7 of BCC, and 10 patients had a history of both tumors. As controls, sera of 80 immunocompetent individuals were randomly selected from the Leiden Skin Cancer Study [8] (collected in 1998). Among them, 39 individuals had a history of SCC and 41 had no skin cancer. None of the individuals displayed lymph node metastases or were known to have a history of other types of cancer (Table 1).

p53-Specific serum antibodies were detected using a quantitated enzyme-linked immunoabsorbent assay kit (anti-p53 ELISA, PharmaCell, France). The threshold value for presence of antibodies was set at 1.15 U/ml following the instructions of the manufacturer.

The mean age of the immunocompetent individuals was 61 years, while the mean age of renal transplant recipients was 47 years (p < 0.01; two-tailed Student's *t*-test). The mean period after transplantation was 13 years in the individuals with a history of skin carcinoma and 12 years in the skin cancer free group (p = 0.32).

Fig. 1 depicts levels of p53-specific antibodies in the renal transplant recipients and immunocompe-

^{0923-1811/\$30.00 © 2005} Japanese Society for Investigative Dermatology. Published by Elsevier Ireland Ltd. All rights reserved. doi:10.1016/j.jdermsci.2005.03.008

	SCC only	BCC only	Both SCC and BCC	No skin cancer
No. of renal transplant recipients	17	7	10	43
p53 pos (no.)	1	2	0	2
p53 neg (no.)	16	5	10	41
No. of immunocompetent patients	28	0	11	41
p53 pos (no.)	1	_	1	2
p53 neg (no.)	27	-	10	39

SCC = squamous cell carcinoma; BCC = basal cell carcinoma.

tent patients. The distributions in these two groups were not discernibly different for SCC only. BCC only, or the combination of SCC and BCC (Table 1). The skin cancer groups were, therefore, combined in the statistical analyses.

In 6.5% (5/77) of the renal transplant recipients and 5.0% (4/80) of the immunocompetent patients p53-specific antibodies were present (p = 0.69). Altogether, 8.8% (3/34) of the renal transplant recipients with a history of skin carcinoma and 5.1% (2/ 39) of the immunocompetent patients with a history of skin carcinoma showed p53-specific antibodies (p = 0.53). One renal transplant recipient and one immunocompetent individual without skin cancer showed highly elevated levels of p53-specific antibodies: 16.4 and 18.7 U/ml, respectively. No obvious clinical reasons were found for these elevated serum levels. In the analyses the outliers were



Fig. 1 Presence of p53-specific antibodies in the different patient groups. ^{*}RTR = renal transplant recipients; ICP = immunocompetent patients; the dashed line represents the cut-off value for the presence of antibodies (=1.15 U/ml); the small horizontal lines represent the mean value after exclusion of the two outliers.

excluded, but inclusion of these outliers did not alter the outcome. Statistical analyses of the p53specific serum antibody levels revealed that there were no significant differences in the mean p53specific antibody levels between the groups. There were no significant age or sex differences between patients with and without p53-specific antibodies either.

Our results show that renal transplant recipients with a history of skin carcinoma and commonly carrying SK rarely display circulating p53-specific IgG antibodies. Despite the limitations of our study on archival sera sampled at various times after removal of skin carcinomas, we infer that there is no indication that circulating p53 antibodies are a useful prognostic marker of skin carcinoma risk in renal transplant recipients (nor in immunocompetent individuals).

Strikingly, 15% (10/66) of a cohort of Japanese renal transplant recipients not bearing any skin carcinomas were reported sero-positive for p53 [9]. Although we used sera of 157 patients, we did not find any indication of elevated titers of p53 antibodies [9]. Because skin carcinomas are extremely rare in Japanese renal transplant recipients [6], these patients cannot simply be compared with their Caucasian counterparts. Interestingly, the sero-positivity in this group of Japanese patients was only found among those who were treated with cyclosporine (n = 56), not in those taking only prednisone (n = 10) [9]. Cyclosporine treatment can result in p53 accumulation [10], which may well explain the Japanese results. None of the transplant recipients tested in our study was treated with cyclosporine, because they were transplanted before the cyclosporine era.

Likely explanations of the low humoral immunization against p53 in our study could be the low skin tumor mass and lack of necrosis and inflammation, in contrast to what is usually found in colon cancers.

In sum, we conclude that there is no indication of an elevated humoral response against p53 in people treated for skin carcinoma, irrespective of whether these people were on immunosuppressive medication or not.

References

- Soussi T. p53 Antibodies in the sera of patients with various types of cancer: a review. Cancer Res 2000;60:1777–88.
- [2] Ziegler A, Jonason AS, Leffell DJ, Simon JA, Sharma HW, Kimmelman J, et al. Sunburn and p53 in the onset of skin cancer [see comments]. Nature 1994;372:773-6.
- [3] McGregor JM, Berkhout RJ, Rozycka M, Ter Schegget J, Bouwes Bavinck JN, Brooks L, et al. p53 Mutations implicate sunlight in post-transplant skin cancer irrespective of human papillomavirus status. Oncogene 1997;15:1737–40.
- [4] Black AP, Ogg GS. The role of p53 in the immunobiology of cutaneous squamous cell carcinoma. Clin Exp Immunol 2003;132:379-84.
- [5] Moch C, Moysan A, Lubin R, De La Salmoniere P, Soufir N, Galisson F, et al. Divergence between the high rate of p53 mutations in skin carcinomas and the low prevalence of antip53 antibodies. Br J Cancer 2001;85:1883–6.
- [6] Euvrard S, Kanitakis J, Claudy A. Skin cancers after organ transplantation. N Engl J Med 2003;348:1681–91.
- [7] Bavinck JN, Gissmann L, Claas FH, Van der Woude FJ, Persijn GG, Ter Schegget J, et al. Relation between skin cancer, humoral responses to human papillomaviruses, and HLA class II molecules in renal transplant recipients. J Immunol 1993;151:1579–86.
- [8] Feltkamp MC, Broer R, Di Summa FM, Struijk L, Van der Meijden E, Verlaan BP, et al. Seroreactivity to epidermodysplasia verruciformis-related human papillomavirus types is associated with nonmelanoma skin cancer. Cancer Res 2003;63:2695–700.
- [9] Shimada H, Nakajima K, Sakamoto K, Takeda A, Hori S, Hayashi H, et al. Existence of serum p53 antibodies in cyclosporine A-treated transplant patients: possible detec-

tion of p53 protein over-expression. Transplant Proc 2000;32:1779.

[10] Pyrzynska B, Serrano M, Martinez A, Kaminska B. Tumor suppressor p53 mediates apoptotic cell death triggered by cyclosporin A. J Biol Chem 2002;277:14102–8.

Ymke G.L. de Graaf* Bert Jan Vermeer Jan Nico Bouwes Bavinck Rein Willemze Frank R. de Gruijl Department of Dermatology, Leiden University Medical Center, Albinusdreef 2, 2300 RC, Leiden The Netherlands

Daniel Schiefer Anke Redeker Sjoerd H. Van der Burg Department of Immunohematology and Blood Transfusion, Leiden University Medical Center Albinusdreef 2, 2300 RC, Leiden, The Netherlands

> *Corresponding author. Tel.: +31 71 5262638 fax: +31 71 5248106 *E-mail address*: y.g.l.de_graaf@lumc.nl

> > 13 January 2005


CHAPTER 4

UV-induced apoptosis is not diminished in the presence of beta-papillomaviruses in habitually unexposed skin, but does decrease with age

Submitted

UV-induced apoptosis is not diminished in the presence of beta-papillomaviruses in habitually unexposed skin, but does decrease with age.

Y.G.L. de Graaf, M.N.C. de Koning,*† W.H. Zoutman, S.J. Weissenborn,‡ H. Pfister,‡ L. Struijk,* M.C.W. Feltkamp,* J. ter Schegget,*† R. Willemze, J. N. Bouwes Bavinck and F.R. de Gruijl.

Departments of Dermatology and *Medical Microbiology, Leiden University Medical Center, Leiden, and †DDL Diagnostic Laboratory, Voorburg, The Netherlands; ‡Institute of Virology, University of Cologne, Cologne, Germany.

Summary

Background Beta-papillomaviruses (beta-PV) may play a role in early skin cancer development either alone or in combination with UV exposure. Earlier in-vitro studies showed that beta-PVs decreased UV-induced apoptosis.

Objectives To investigate whether beta-PVs decrease UV-induced apoptosis in-vivo and whether UV exposure affects the presence and quantity of beta-PVs.

Methods Thirty organ-transplant recipients (OTR) and 30 immunocompetent individuals (ICI) were exposed to 3 minimal erythema doses (MED) UVB irradiation on unexposed buttock skin. Biopsies were taken from both irradiated and non-irradiated skin. Beta-PV types were detected and identified by PCR and reverse hybridization and quantified by q-PCR. Apoptotic cells were stained with anti-active caspase 3.

Results Beta-PV DNA was detected in 47% of non-irradiated and 53% of UVB-irradiated skin samples from OTR, and in 27% of both non-irradiated and UVB-irradiated skin samples from ICI. The quantity of beta-PV was generally below 1 copy per 1000 cells. There was no effect of a single UV exposure on the detected presence of beta-PV. We did not detect an association between beta-PV presence and UV-induced apoptosis in both groups studied. We did, however, find in both OTR and ICI a decrease in UV-induced apoptosis with age, which was not linked to any significant increase in beta-PV with age.

Conclusions In line with low copy numbers of beta-PV, we did not detect any overall decrease in UV-induced apoptosis in beta-

PV positive skin samples, nor any immediate effect of UV irradiation on the beta-PV types detected. In both OTR and ICI, the number of UV-induced apoptotic cells decreased with age.

Introduction

The majority of organ-transplant recipients (OTR) develop multiple skin cancers, especially squamous-cell carcinomas, on sun-exposed skin.¹ In transplant recipients the development of skin cancer is strongly associated with the number of keratotic skin lesions, consisting of viral warts and actinic keratoses.^{2;3}

Recently, epidermodysplasia verruci-formis (EV)-related human papillomaviruses were re-classified into beta-papillomaviruses (beta-PV). Infection with beta-PV occurs frequently and may persist for many years.⁴ A wide diversity of beta-PV-types can be detected in both pre-malignant skin lesions and skin carcinomas.^{2,5-7} Beta-PV-DNA has been detected in up to 90% of squamous-cell carcinomas of OTR.⁸ In immunocompetent individuals (ICI) detection rate of beta-PV-DNA in skin carcinomas is lower, namely between 30 and 50%.89 The presence of beta-PV-DNA in plucked eyebrow hairs of ICI is associated with a history of squamouscarcinoma.¹⁰ cell Sero-epidemiological studies also showed an association between sero-response against beta-PV and skin cancer risk.^{11;12} The presence of antibodies against HPV8 was particularly associated with the development of squamous-cell carcinomas^{11;13-15} and actinic keratoses.¹³ All these findings together provide indirect evidence that beta-PV may play a role in skin cancer development in organ-transplant recipients and immunocompetent individuals either directly or in combination with ultraviolet exposure.^{16;17}

Ultraviolet B (UVB) exposure has been recognized as the most important etiological factor in the development of skin cancer. UVB irradiation causes the induction of somatic mutations through the formation of pyrimidine dimers and 6-4 photoproducts. The failure to repair this DNA damage or to remove severely damaged cells by apoptosis may ultimately lead to skin cancer.¹⁸⁻²⁰

In-vitro, beta-PV have been reported to hamper DNA repair.^{21;22} Recently, it also has been shown that beta-PV E6 proteins inhibit apoptosis in-vitro in response to UV damage.^{23;24} The impact on apoptosis has not been studied in-vivo. The aim of the present study was to investigate whether apoptosis was decreased in the presence of beta-PV after an UVB challenge in human skin invivo. Furthermore, we investigated whether the presence and diversity of beta-PV in previously non-sunexposed skin were modified by a single UVB exposure. For this purpose we used a novel, highly sensitive beta-PV detection technique.²⁵

Material and methods

Subjects OTR who were regularly seen at the Department of Dermatology from the Leiden University Medical Center (LUMC) in the period between January 2003 and January 2005 were eligible for the study. To study the impact of beta-PV separately in immunocompetent individuals, volunteers were recruited from the LUMC and the University of Leiden through advertisements. Consequentially, the groups were not matched, but studied independently for the effects of beta-PV. Inclusion criteria among OTR were a functioning graft of 5 years or longer; for both patients and controls inclusion criteria were an age of 18 years or older; skin type I through III and sun-unexposed buttock skin, without preexistent solar damage. ICI with systemic diseases were excluded. The medical ethical committee of the LUMC approved the study and all participants provided written informed consent. In total, 30 organtransplant recipients and 30 immunocompetent individuals were included in the study.

MED testing and UVB irradiation For the determination of the minimal erythema dose (MED), which is defined as the smallest dose of UV radiation to result in just detectable erythema, a phototesting device designed by Diffey *et al.* was used.^{26;27} The device is made of two flexible metal foils, containing 10 ports with mesh attenuators giving 10 irradiance levels with an 8 : 1 ratio from the highest to the lowest exposure. The device was slightly modified to maintain a constant distance of 3 mm to the skin.

At the day-1 visit, the MED test was performed on the previously non-sunexposed buttock skin of each individual. The phototesting device was placed on the skin and irradiated with broadband UVB TL-12 tubes (Philips, Eindhoven, The Netherlands) for 4 minutes, giving an irradiance ranging from 30 to 240 mJ cm⁻² UV. The MED was assessed visually 24 hours later. No difference between the MED values of the OTR and the ICI was observed. The average MED values [95% CI] were 125 [117-133] and 135 [120-150] mJ cm⁻², respectively. Subsequently, an area of 1.5 cm^2 of the opposite unexposed buttock skin was exposed to 3 MED UVB irradiation. On the third day, 24 hours after irradiation, one 4mm punch biopsy was taken from the 3-MED area, together with a control biopsy from adjacent (a maximum distance of 1 cm) non-irradiated buttock skin of OTR and ICI. All biopsies were embedded in Tissue-Tek (Sakura, Zoeterwoude, The Netherlands), snap frozen in liquid nitrogen and stored at -80 °C.

To investigate chronically sunexposed skin, we also took paired biopsies of UVirradiated and non-irradiated dorsal forearm skin. MED determination was, however,

very difficult in this heavily sun-damaged skin with high numbers of solar keratoses and other hyperkeratotic skin lesions. The MED readings were considered verv unreliable. Therefore, these forearm specimens were not suitable for comparative measurements of apoptosis. After testing 9 OTR on the forearm, this part of the experiment was discontinued, and only the presence of beta-PV types was ascertained in the acquired forearm skin biopsies.

Processing of the samples For all samples, first 5 µm cryosections for immunohistochemistry were cut. Six serial cryosections were used for DNA isolation. DNA was extracted for both the PM-PCR RHA method (see below) and the quantitative PCR using the Genomic Tip kit (Qiagen, Hilden, Germany).

To prevent contamination, after each sample strict cleaning of the blade was performed. Furthermore, a control (blank) sample, consisting solely of Tissue-Tec, was cut after each pair of biopsies of one individual. All control samples were processed identically to the skin samples. During analysis seven out of 60 blank control samples were beta-PV-positive, each for a different type of beta-PV. Since these types showed no correspondence with the ones found in samples cut directly preceding or following the control sample, they were considered false positives from minor random contamination. An airborn infection cannot be excluded and appears to be the most likely explanation. This false positive rate constituted a low background noise, which was substantially and significantly lower than the positive rates in the skin samples of both OTR and ICI (p<0.001).

The PM-PCR RHA method and quantitative PCR Beta-papillomavirus detection and genotyping was carried out with the PM PCR Reverse Hybridization Assay (PM-PCR RHA) method (Skin (beta) HPV prototype research assay; Diassay BV, The Netherlands). We have applied strict guidelines adapted from Kwok et al.²⁸ to

prevent contamination. The method was designed for the identification of 25 established beta-PV types, namely beta-PV genotypes 5, 8, 9, 12, 14, 15, 17, 19, 20, 21, 22, 23, 24, 25, 36, 37, 38, 47, 49, 75, 76, 80, cand92, 93 and cand96. This method was described in detail earlier.²⁵ In short, a broadspectrum consensus PCR primer set was used for amplification of a 117 bp fragment from the beta-PV E1 gene. Generated amplimers were subsequently analysed in the RHA on nitrocellulose membrane strips containing oligo nucleotide probes for 25 beta-PV types. The RHA includes several hybridisation and (stringent) washing steps followed by an enzymatic colouring reaction that visualises the presence of biotinylated amplimer bound to the probes. The analytical sensitivity of the PM-PCR RHA method was reported to be 10 100 viral genomes. Intraand to interlaboratory variability experiments showed that the reproducibility of the assay was very high. Furthermore, no aspecific results were reported as assessed by theoretical alignments, challenging the genus specificity of the PCR and analysis of high concentrations of amplimers derived from the 25 beta-PV types.²⁵

For a subset of positive buttock skin samples a quantitative PCR for beta-PV types 5, 8, 15, 20, 23, 24, 36, 38 was performed, as described previously.²⁹ The quantity of beta-PV was expressed as 1 beta-PV copy per n cells.

Activated caspase 3 immunohistochemistry and scoring apoptosis Frozen skin sections $(5 \mu m)$ on 3-amino-propyltriaethoxysilanecoated glass slides were fixed for 10 min. in acetone at room temperature. Before and between incubation steps, the slides were washed three times with phosphate-buffered saline (PBS) containing 0.05% Tween-20 for 5 min. After blocking with 2% normal human serum diluted in PBS /1% BSA for 20 min, the slides were incubated overnight with the primary antibody; polyclonal rabbit anti-active caspase 3 (Cat. No. 557035, BD Pharmingen) diluted 1:30 in PBS/1% BSA. After washing, skin sections were incubated with the second antibody for 60 min; biotinylated goat anti-rabbit (IgG) (1:200; Vector. Burlingame. CA. USA). Subsequently, the slides were incubated with a horseradish peroxidase (HRP)-labelled avidin-biotin complex (DAKO, diluted 1:100 in PBS/1% BSA) for 60 min. Antibody binding was visualized by incubating the sections in 0.1 M acetate buffer (pH 5) containing 20 mg of 3-amino-ethyl-carbazole (Sigma, St Louis, MO, USA) and 100 µl of 30% hydrogen peroxide per 100 ml during 7 min. The skin sections were counterstained with haematoxylin and rinsed with running tap water for 10 min. All slides were mounted with Kaiser's glycerin and kept at room temperature. Anti-active caspase 3 positive cells in the epidermis were quantified per high-power field (original magnification x 400). The number of positive cells in all fields (mean 17; range 4-36) was assessed. Subsequently, the average number of positive epidermal cells per highpower field was calculated.

Statistical analyses The sample size of this study was calculated according to the criterion that a difference of 2 apoptotic cells/high-power field between beta-PVpositive and -negative skin still should be detectable. Beta-PVs have been detected more frequently in normal skin samples of organ-transplant recipients (17-87%) immunocompetent positivity) than in individuals (16-35% positivity).^{4;6;30;31} It was expected that in 80% of the OTR and 30% of immunocompetent individuals beta-PV would be present in healthy skin. With a power of 80% and a significance level (alpha) of 0.05, we calculated that 30 OTR would be sufficient to distinguish significantly between beta-PV-positive and negative skin in level of apoptosis. We also included 30 ICI and studied both groups separately.

For the statistical analyses, we used SPSS version 12.0.1 for Windows. To compare the mean levels of apoptosis between beta-PV-infected and non-infected skin in and between OTR and ICI the Student's T-Test was used. The association between age and apoptosis was tested by linear regression analysis. All analyses were performed for the OTR and ICI, separately.

Results

Baseline characteristics

The baseline characteristics of the organtransplant recipients and immunocompetent individuals are listed in Table 1.

Beta-PV DNA was frequently detected in low quantities and not affected by UV exposure

Tables 2 and 3 show the results of beta-PV DNA typing in skin biopsies of 30 OTR and 30 ICI, respectively. In the OTR beta-PV DNA was detected in 14 (47%) nonirradiated and 16 (53%) irradiated buttock skin biopsies (Table 2). In the 30 ICI beta-PV DNA was detected in 8 (27%) of both non-irradiated and irradiated buttock skin biopsies (Table 3). In the forearm skin of 9 OTR the beta-PV positive percentages were 89% and 63%, respectively (Table 2). In both groups, the most prevalent beta-PV types in buttock skin were beta-PV 8, 23, and 38. In OTR vs. ICI beta-PV 8 was present in 12% (7/60) and 7% (4/60) of the samples respectively, beta-PV 23 was found in 17% (10/60) and 7% (4/60), and beta-PV 38 in 15% (9/60) and 3% (2/60) of the samples, respectively (Tables 2 and 3).

Age was not associated with the presence of beta-PV (Figure 1). The mean age (SD) of the OTR with 1 or more beta-PV types in any biopsy was 55 (11) compared with an age of 54 (14) years in OTR without beta-PV. In ICI with and without beta-PV present the mean age was 33 (12) and 32 (12)respectively. Time after transplantation did not influence beta-PV positivity either. The beta-PV positive patients were tested at a mean of 19 years after transplantation, while the beta-PV negative patients were 16 years post transplantation. Also, gender was not significantly associated with presence of beta-PV DNA in the OTR and ICI. There was a large inter- and intraindividual variation in the number and types of beta-PV

Table 1. Characteristics

	Organ-transplant recipients	Immunocompetent individuals
No. of individuals	30	30
Sex		
Male	15	15
Female	15	15
Age (years)		
Mean \pm SD	55 ± 12	32 ± 12
Range	32 – 77	19 – 57
Time after transplantation (years)		
Mean \pm SD	18 ± 9	
Range	5 - 36	
History of skin cancer		
No. of patients with skin cancer (%)	19 (63%)	0 (0%)
Mean no. of skin cancers \pm SD	6 ± 9	
Range	1-38	
Medication type (No. of patients; %)		
P + A	14 (47%)	
P + C	8 (27%)	
P + C + A	2 (7%)	
P + M	2 (7%)	
P + C + M	2 (7%)	
P	1 (3%)	
Unknown (clinical trial)	1 (3%)	

M=mycophenolate mofetil

 Table 2. Beta-PV types in buttocks and forearms and apoptosis in buttock skin of organ-transplant

 recipients, presented with increasing levels of apoptosis.

No.	Gender	Butt	<u>ocks</u>	Fore	earms	Apoptosis
	and age	Control	Biopsies	Control biopsies	Biopsies irradiated	after UV in
	(yrs)	biopsies (not	irradiated	(not-irradiated)	skin	buttock skin
		irradiated)	skin			(no. of pos.
						cells/ field)
1	M 56	-	-	76	-	0.1
2	F 53	19	23,37	-	-	0.4
3	M 53	20,23	8,17,38	22,23,80,92	22,23,80,92,93	0.6
4	M 65	22,24,76	24,36	N.d.	N.d.	0.9
5	F 52	-	5,23,80	17,24	-	1.0
6	M 44	-	-	20,37	93	1.2
7	F 42	20,23	9	8,12,20,23,75	23,24	1.4
8	F 69	8,25	25,38	17,23,25,37,38,96	5,15,17,23,36,38	1.4
9	M 57	-	-	N.d.	N.d.	1.6
10	F 45	96	-	N.d.	N.d.	1.6
11	F 50	-	38,76	12,22,24,80	12,22,37	1.6
12	F 63	8,36,38,92	8,36	N.d.	N.d.	2.0
13	M 77	-	-	N.d.	N.d.	2.9
14	M 74	8,23,38	-	N.d.	N.d.	2.9
15	F 57	-	15,92	N.d.	N.d.	3.0
16	M 70	-	-	N.d.	N.d.	3.2
17	M 61	-	-	N.d.	N.d.	3.4
18	F 65	75	-	N.d.	N.d.	3.5
19	F 64	38	38	N.d.	N.d.	3.6
20	M 72	19,23,93	23,37,80,93	N.d.	N.d.	4.5
21	F 36	17	-	N.d.	N.d.	4.9
22	F 60	-	22	38	-	6.0
23	M 39	-	8,23	N.d.	N.d.	6.7
24	M 68	-	-	N.d.	N.d.	7.0
25	M 44	-	-	N.d.	N.d.	7.3
26	M 44	22,38,92	23	N.d.	N.d.	7.4
27	M 32	-	-	N.d.	N.d.	8.0
28	F 42	-	38	N.d.	N.d.	12.4
29	F 57	8,23,80,92,93	24,92	N.d.	N.d.	12.9
30	F 40	-		N.d.	N.d.	24.5

Abbreviations: N.d.=not done; M=male; F=female; - =no beta-PV present

No.	Gender		Butto	cks
	and age	Control bionsies (not	Bionsies irradiated	Apoptosis after UV (no. of pos. calls/
	(vrs)	Control biopsies (not	Biopsies maulateu	Apoptosis after 0 v (no. of pos. cens/
	(915)	irradiated)	skin	field)
1	M 56	-	-	1.0
2	F 57	-	-	2.4
3	F 30	15	-	2.6
4	M 24	-	-	3.3
5	M 57	-	8,38	3.7
6	M 25	-	-	3.7
7	M 42	-	-	3.8
8	F 51	-	24	4.3
9	F 36	-	23	4.3
10	M 32	-	-	5.5
11	F 26	-	-	6.3
12	M 46	-	-	6.4
13	M 21	-	38	6.5
14	F 21	-	-	7.9
15	M 40	8	-	8.6
16	F 39	-	-	9.3
17	F 25	80	15	9.9
18	M 44	-	-	10.1
19	F 20	-	-	10.2
20	F 25	9	-	10.6
21	F 23	-	-	10.7
22	M 27	-	25	11.6
23	M 23	8	-	12.0
24	M 19	-	-	12.0
25	M 24	-	-	12.2
26	M 23	14	-	13.9
27	F 20	-	-	15.7
28	F 32	23	23	16.9
29	F 49	8	-	20.0
30	F 22	-	5,22,23	37.0
Abbrev	viations: M=	male; F=female; - = no b	eta-PV present	

Table 3. Beta-PV-types and apoptosis in buttock skin of immunocompetent individuals, presented with increasing levels of apoptosis.

detected in both OTR and ICI (Tables 2 and 3). In the OTR, the number of types detected in irradiated skin varied from 1 to 4 (mean 1.9), and in the non-irradiated skin from 1 to 5 (mean 2.3). In the ICI the number of types in irradiated skin varied from 1 to 3 (mean 1.4), and in the non-irradiated biopsies all beta-PV positive samples showed only 1 beta-PV type. UV-irradiation did not affect the presence or number of beta-PV in buttock skin nor in forearm skin (Tables 2

and 3).

The quantity of beta-PV was determined in 26 beta-PV positive samples, of which only two samples yielded a positive signal in the quantitative PCR (for beta-PV 23 and 38). One copy of beta-PV DNA was found in less than around 1000 cells in both samples. In the remaining 24 samples beta-PV-DNA loads were apparently below the detection limit (< 10 copies of beta-PV per reaction) of the quantitative PCR (data not shown).²⁹



Figure 1. Beta-PV positivity (%) with 95% confidence intervals in different age groups based on median age in organ-transplant recipients and immunocompetent individuals. The findings were based on the cumulative test results obtained from two buttock skin biopsies per individual

Apoptosis was not reduced in beta-PV-infected skin but decreased with age

As expected no apoptotic cells were detected in non-UVB irradiated skin. In UVBexposed buttock skin of organ-transplant recipients the number of apoptotic cells/high-power field varied from 0.1 to 24.5 (median 3.1). In the immunocompetent individuals the number of apoptotic cells varied from 1.0 to 37.0 (median 9.0) (Figure 2) (Tables 2 and 3).

Interestingly, in the UVB-irradiated biopsies of both the OTR and ICI, the number of apoptotic cells decreased with age (p=0.05 and 0.06 respectively) (Figure 3). Gender was not associated with apoptosis.

In both groups there was no difference in apoptosis between beta-PV positive and negative buttock skin (Figure 4). Also, when examining the effect of the most prevalent beta-PV-types, beta-PV 8, 23, 38,

separately, no effect on apoptosis could be detected. Additionally, there was no difference in apoptosis levels between organtransplant recipients with and without a history of skin cancer. The type of medication (cyclosporine or azathioprine) was not associated with the level of apoptosis either (data not shown). Adjusting our analyses for age and sex of the OTR and ICI did not change these findings.



Figure 2. Anti-active caspase 3 positive cells in the epidermis (original magnification x 400).



Figure 3. Relation between mean number of anti-active caspase 3 cells per high-power field and age in organ-transplant recipients and immunocompetent individuals. \bullet : beta-PV positive, \blacktriangle : beta-PV negative



Figure 4. Mean number of anti-active caspase 3 cells per high-power field with 95% confidence intervals in beta-PV-positive and -negative skin samples in organ-transplant recipients and immunocompetent individuals

Discussion

In our study, we could not confirm in-vivo the hypothesis that beta-PVs may decrease UV-induced apoptosis. Based on our study, however, we cannot immediately conclude, that beta-PVs do not decrease UV-induced apoptosis. The low quantity of beta-PV DNA in the skin biopsies, with generally less than one copy of beta-PV per 1000 cells, would explain that we did not find any appreciable decrease in the number of apoptotic cells in beta-PV positive biopsies. In-situ hybridization beta-PV with types of individual apoptotic and non-apoptotic cells is a more precise method to study the effect

of beta-PV infection on UV-induced apoptosis in individual cells. Unfortunately, this technique is not available yet and would be not conclusive in our biopsies because of the low viral DNA loads.

It cannot be entirely excluded that part of the beta-PV DNA is present on the surface of the skin. But this DNA probably originated from local, viral replication as can be inferred from the recent findings that beta-PV DNA is persistently present in plucked hairs³² and, importantly, also in swaps of skin surface.³³ These data strongly suggest that the presence of this viral DNA is due to persistent beta-PV infections and not to contamination.

It is likely that not all beta-PV types exert similar effects on UV-induced apoptosis,

which may be an alternative explanation of non-detectable effect of the beta-PV infection on UV-induced apoptosis. To study the effect of all 25 known individual beta-PV types, separately, much larger series of skin biopsies would be needed. To discriminate better between the effects of the different beta-PV genotypes and to overcome the problem of low viral loads, the impact on UV-induced apoptosis could be studied in cultures of beta-PV transfected keratinocytes.

The dynamics of beta-PV infection in reaction to UV-irradiation is also not known yet. We observed no appreciable effect, but a single UV exposure may not be sufficient to measure an effect of beta-PV infection on UV-induced apoptosis in-vivo. Multiple exposures may lead to a more pronounced effect. We did find a tendency of a higher prevalence of beta-PV in the forearms than in the buttocks of OTR, which supports the notion that beta-PV is more prevalent in chronically sun-exposed, lesional skin. A recent study by Harwood et al. did not show such a difference between chronically sunexposed and unexposed skin.³¹ An earlier study by de Jong-Tieben et al. showed a higher prevalence of beta-PV-DNA in benign skin lesions from chronically sunexposed sites compared with unexposed skin.² This latter result is in line with our supposition that chronic sun-exposure may induce beta-PV replication. It is known that at least some beta-PVs have UV responsive elements 34;35

In our samples numerous beta-PV genotypes were detected and the mean number of beta-PV types in buttock skin was clearly higher in skin from OTR than from

ICI. However, the OTR and ICI were not age and sex-matched, and therefore not directly comparable. But in the overlapping age range from around 30 to 57 years the beta-

PV positive rate was higher in OTR than ICI (75 vs. 50 %, Figure 1), although not significantly.

Beta-PV effects were meant to be assessed in each group separately. The viral loads were indeed found to be low, and most frequently below the detection limits of the quantitative PCR, but in the similar range as reported before.²⁹ Corresponding to a sparseness of the viruses in the biopsies, there was a large inter- and intraindividual variation in the number and types of beta-PV.

Interestingly, our study showed that the level of UV-induced apoptosis decreased with increasing age in both OTR and ICI without a link to any appreciable age dependence of beta-PV or of MED values (the overall difference in apoptosis between OTR and ICI is fully attributable to age differences).

A relation between decreased apoptosis and skin cancer development has been established in animal experiments.³⁶ A study in which UV-induced apoptosis was measured in peripheral blood lymphocytes, showed a significant decrease in apoptosis with increasing age in melanoma patients.³⁷

In conclusion, our data show that normal unexposed skin frequently contained viral DNA of various beta-PV types, but in very low quantities. We observed a decreased level of apoptosis with increasing age, not linked to any increase in beta-PV. However, we did not find a relation between beta-PV presence and overall UV-induced apoptosis. To ascertain the effect of beta-PV infection on UV-induced apoptosis in-vivo studies need to focus specifically on beta-PVcarrying cells.

Acknowledgments

We thank Patrick Wanningen, Aat Mulder, Coby Out, Remco Dijkman and Heggert Rebel for their technical assistance and Ron Wolterbeek for helpful discussion and power calculation.

References

- 1. Euvrard S, Kanitakis J, Claudy A. Skin cancers after organ transplantation. *N.Engl.J.Med.* 2003; **348**: 1681-91.
- de Jong-Tieben LM, Berkhout RJ, ter Schegget J *et al*. The prevalence of human papillomavirus DNA in benign keratotic skin lesions of renal transplant recipients with and without a history of skin cancer is equally high: a clinical study to assess risk factors for keratotic skin lesions and skin cancer. *Transplantation* 2000; 69: 44-9.
- 3. Bouwes Bavinck JN, De Boer A, Vermeer BJ *et al.* Sunlight, keratotic skin lesions and skin cancer in renal transplant recipients. *Br.J.Dermatol.* 1993; **129**: 242-9.
- Berkhout RJ, Bouwes Bavinck JN, ter Schegget J. Persistence of human papillomavirus DNA in benign and (pre)malignant skin lesions from renal transplant recipients. *J.Clin.Microbiol.* 2000; 38: 2087-96.
- Meyer T, Arndt R, Christophers E et al. Importance of human papillomaviruses for the development of skin cancer. Cancer Detect. Prev. 2001; 25: 533-47.
- Meyer T, Arndt R, Nindl I *et al.* Association of human papillomavirus infections with cutaneous tumors in immunosuppressed patients. *Transpl.Int.* 2003; 16: 146-53.
- de Jong-Tieben LM, Berkhout RJ, Smits HL et al. High frequency of detection of epidermodysplasia verruciformisassociated human papillomavirus DNA in biopsies from malignant and premalignant skin lesions from renal transplant recipients. J. Invest Dermatol. 1995; 105: 367-71.
- Pfister H. Chapter 8: Human papillomavirus and skin cancer. J.Natl.Cancer Inst.Monogr 2003; 52-6.
- Harwood CA, Proby CM. Human papillomaviruses and nonmelanoma skin cancer. *Curr. Opin. Infect. Dis.* 2002; 15: 101-14.
- Struijk L, Bouwes Bavinck JN, Wanningen P et al. Presence of human papillomavirus DNA in plucked eyebrow hairs is associated with a history of cutaneous squamous cell carcinoma. J.Invest Dermatol. 2003; 121: 1531-5.
- Feltkamp MC, Broer R, di Summa FM *et al.* Seroreactivity to epidermodysplasia verruciformis-related human papillomavirus types is associated with nonmelanoma skin cancer. *Cancer Res.* 2003; 63: 2695-700.
- Karagas MR, Nelson HH, Sehr P et al. Human papillomavirus infection and incidence of squamous cell and basal cell carcinomas of the skin. J.Natl. Cancer Inst. 2006; 98: 389-95.
- Bouwes Bavinck JN, Stark S, Petridis AK *et al.* The presence of antibodies against virus-like particles of epidermodysplasia verruciformis-associated humanpapillomavirus type 8 in patients with actinic keratoses. *Br.J.Dermatol.* 2000; 142: 1039.
- Masini C, Fuchs PG, Gabrielli F *et al.* Evidence for the association of human papillomavirus infection and cutaneous squamous cell carcinoma in immunocompetent individuals. *Arch.Dermatol.* 2003; **139**: 890-4.
- Struijk L, Hall L, van der ME *et al.* Markers of cutaneous human papillomavirus infection in individuals with tumor-free skin, actinic keratoses, and squamous cell carcinoma. *Cancer Epidemiol.Biomarkers Prev.* 2006; 15: 529-35.
- Bouwes Bavinck JN, Feltkamp MC. Milk of human kindness?--HAMLET, human papillomavirus, and warts. *N.Engl.J.Med.* 2004; 350: 2639-42.
- 17. Hall L, Struijk L, Neale RE *et al.* Re: Human papillomavirus infection and incidence of squamous cell and basal cell carcinomas of the skin. *J.Natl.Cancer Inst.* 2006; **98**: 1425-6.

- Brash DE, Rudolph JA, Simon JA et al. A role for sunlight in skin cancer: UV-induced p53 mutations in squamous cell carcinoma. Proc.Natl.Acad.Sci.U.S.A 1991; 88: 10124-8.
- Ziegler A, Leffell DJ, Kunala S *et al.* Mutation hotspots due to sunlight in the p53 gene of nonmelanoma skin cancers. *Proc.Natl.Acad.Sci.U.S.A* 1993; **90**: 4216-20.
- Ziegler A, Jonason AS, Leffell DJ *et al*. Sunburn and p53 in the onset of skin cancer [see comments]. *Nature* 1994; **372**: 773-6.
- Giampieri S, Storey A. Repair of UV-induced thymine dimers is compromised in cells expressing the E6 protein from human papillomaviruses types 5 and 18. *Br.J.Cancer* 2004; 90: 2203-9.
- Iftner T, Elbel M, Schopp B *et al.* Interference of papillomavirus E6 protein with single-strand break repair by interaction with XRCC1. *EMBO J.* 2002; 21: 4741-8.
- Jackson S, Harwood C, Thomas M *et al*. Role of Bak in UVinduced apoptosis in skin cancer and abrogation by HPV E6 proteins. *Genes Dev.* 2000; 14: 3065-73.
- Jackson S, Storey A. E6 proteins from diverse cutaneous HPV types inhibit apoptosis in response to UV damage. *Oncogene* 2000; 19: 592-8.
- de Koning M, Quint W, Struijk L *et al.* Evaluation of a Novel Highly Sensitive, Broad-Spectrum PCR-Reverse Hybridization Assay for Detection and Identification of Beta-Papillomavirus DNA. *J.Clin.Microbiol.* 2006; 44: 1792-800.
- Diffey BL, De Berker DA, Saunders PJ *et al.* A device for phototesting patients before PUVA therapy. *Br.J.Dermatol.* 1993; **129**: 700-3.
- Gordon PM, Saunders PJ, Diffey BL *et al*. Phototesting prior to narrowband (TL-01) ultraviolet B phototherapy. *Br.J.Dermatol*. 1998; **139**: 811-4.
- Kwok S, Higuchi R. Avoiding false positives with PCR. *Nature* 1989; 339: 237-8.
- Weissenborn SJ, Nindl I, Purdie K et al. Human papillomavirus-DNA loads in actinic keratoses exceed those in non-melanoma skin cancers. J. Invest Dermatol. 2005; 125: 937.
- Astori G, Lavergne D, Benton C *et al.* Human papillomaviruses are commonly found in normal skin of immunocompetent hosts. *J.Invest Dermatol.* 1998; **110**: 752-5.
- Harwood CA, Surentheran T, Sasieni P *et al*. Increased risk of skin cancer associated with the presence of epidermodysplasia verruciformis human papillomavirus types in normal skin. *Br.J.Dermatol.* 2004; **150**: 949-57.
- de Koning MN, Struijk L, Bavinck JN et al. Betapapillomaviruses frequently persist in the skin of healthy individuals. J. Gen. Virol. 2007; 88: 1489-95.
- Hazard K, Karlsson A, Andersson K et al. Cutaneous human papillomaviruses persist on healthy skin. J.Invest Dermatol. 2007; 127: 116-9.
- 34. Akgul B, Lemme W, Garcia-Escudero R *et al.* UV-B irradiation stimulates the promoter activity of the high-risk, cutaneous human papillomavirus 5 and 8 in primary keratinocytes. *Arch. Virol.* 2005; **150**: 145-51.
- Purdie KJ, Pennington J, Proby CM *et al.* The promoter of a novel human papillomavirus (HPV77) associated with skin cancer displays UV responsiveness, which is mediated through a consensus p53 binding sequence. *EMBO J.* 1999; 18: 5359-69.
- Ziegler A, Jonason AS, Leffell DJ et al. Sunburn and p53 in the onset of skin cancer. *Nature* 1994; 372: 773-6.
- Pedeux R, Sales F, Pourchet J *et al*. Ultraviolet B sensitivity of peripheral lymphocytes as an independent risk factor for cutaneous melanoma. *Eur.J. Cancer* 2006; 42: 212-5.

CHAPTER 5

Systemic and topical retinoids in the management of skin cancer in organ-transplant recipients

Dermatologic Surgery 2004; 30: 656-661

Systemic and Topical Retinoids in the Management of Skin Cancer in Organ Transplant Recipients

Y. G. L. DE GRAAF, MD,* S. EUVRARD, MD,[†] AND J. N. BOUWES BAVINCK, MD, PHD*

*Department of Dermatology, Leiden University Medical Center, Leiden, The Netherlands, and [†]Department of Dermatology, Hôpital Edouard Herriot, Lyon, France

BACKGROUND. Nonmelanoma skin cancers are the most frequent malignancies in organ transplant patients. Patients who develop multiple new skin cancers may benefit from retinoid chemoprevention.

OBJECTIVE. The objective of this study was to advise on the use of retinoids in organ transplant recipients.

METHODS. A summary was performed of the existing literature regarding experience with retinoid chemoprevention for skin cancer in organ transplant patients.

RESULTS. Systemic retinoids, specifically acitretin, are effective in inhibiting tumor development in organ transplant patients. This effect is only present during therapy, however. Topical retinoids have some effect in the treatment of actinic keratoses.

CONCLUSIONS AND RECOMMENDATIONS. Systemic retinoids can be used for chemoprevention of skin cancer. For a good result, long-term treatment with acitretin is necessary. Side effects, however, limit the use of retinoid chemoprevention. It is advised that treatment be started at a low dose, and patients should be monitored regularly for triglyceride and cholesterol levels and transaminases.

Y. G. L. DE GRAAF, MD, S. EUVRARD, MD, AND J. N. BOUWES BAVINCK, MD, PHD HAVE INDICATED NO SIGNIFICANT INTEREST WITH COMMERCIAL SUPPORTERS.

Skin Cancer in Organ Transplant Recipients

NONMELANOMA SKIN cancers, especially squamous cell carcinomas, are the most frequent malignancies in organ transplant recipients.^{1,2} The frequency of skin cancers increases progressively with time after transplantation. In temperate climates such as in the Netherlands, 40% of the renal transplant recipients will develop skin cancer 20 years after transplantation.³ In sunny climates like Australia these percentages are even higher.⁴

Furthermore, in this particular population squamous cell carcinomas tend to be multiple and may have a more aggressive course with frequent recurrences compared to the immunocompetent host. Often there is an association with warts, premalignant keratoses, Bowen's disease, and keratoacanthomas.²

Prevention and Management of Skin Cancer

In organ transplant recipients, intensive surveillance and treatment of precancers and cancers in an early stage is necessary.⁵ This consists mainly of multiple sessions of cryotherapy, curettage and coagulation, and repeated surgical excisions of tumors. Resurfacing the dorsum of the hand may also be useful in selected patients.⁶

Organ transplant recipients who may benefit from retinoid chemoprevention are those who are actively developing large numbers of skin cancers.^{7,8} In this overview we will summarize the most important literature regarding the experience with systemic and topical retinoids in the chemoprevention of skin cancer in organ transplant recipients.

Mechanism of Action

Retinoids are structural and functional analogs of vitamin A that display a wide range of biologic activity.⁹ Results of basic investigations suggest that retinoids act chemopreventively by inducing growth arrest or apoptosis of tumor cells, by modulating the immune response or differentiation of keratinocytes, or a combination of these events.^{10–12}

Intracellularly, retinoids interact with cytosolic proteins and two families of nuclear receptors, the retinoic acid receptors and retinoid X receptors. As a result, specific genes are expressed.¹² It has also been shown that retinoic acid is more effective in inhibiting the growth of human papillomavirus (HPV)-infected human keratinocytes than that of normal

Address correspondence and reprint requests to: J. N. Bouwes Bavinck, Leiden University Medical Center, Department of Dermatology, Albinusdreef 2, PO Box 9600, 2300 RC Leiden, the Netherlands, or e-mail: J.N.Bouwes_Bavinck@lumc.nl.

keratinocytes, probably by suppressing the early transcription of HPV genes.^{13,14}

Rook et al.¹⁵ reported a marked increase of Langerhans cells within the epidermis in three patients after treatment with etretinate and topically applied tretinoin. The increase of Langerhans cells was correlated with duration of therapy. This may indicate a possible immunologic mechanism. In addition, a (statistically not significant) trend was reported for an increase in Langerhans cell density in squamous cell carcinomas during etretinate therapy.¹⁶

Warts in renal transplant recipients showed pronounced K13 expression in contrast to warts in the immunocompetent host. This may reflect an important molecular event inherent in the malignant degeneration of warts in renal transplant recipients. Retinoid treatment significantly correlated with a specific pattern of K13 expression in skin lesions of renal transplant recipients. It has been suggested that by keeping keratinocytes in this esophageal-type differentiation, retinoids might act chemopreventively.¹⁷ Six months of treatment with topical tretinoin resulted clinically in an increased skin thickness. Histologic and ultrastructural examination of retinoic acid treated "normal skin" of renal transplant recipients revealed epidermal and dermal changes evoking increased cellular metabolism in the treated sites.¹⁸

Use of Retinoids in Nonimmunosuppressed Individuals

Systemic Retinoids

Systemic retinoids can be administered to treat precancers and cancers, but more often are used as chemopreventive agents. Treatment of actinic keratoses and Bowen's disease in immunocompetent patients was effective with the use of various retinoids.¹² Treatment of basal cell carcinomas and squamous cell carcinomas, however, showed no clear beneficial effects.¹²

In most trials, retinoids have been administered to prevent skin cancer. These trials were performed in immunocompetent individuals and in patients with xeroderma pigmentosum.¹⁹

Xeroderma pigmentosum is a rare recessive disease of defective DNA repair, in which patients have a 1000-fold increased risk of developing skin cancers.²⁰ In a 3-year controlled prospective study of oral isotretinoin at a dose of 2 mg/kg/day in five xeroderma pigmentosum patients, a reduction of 63% of tumors (p = 0.02) was reported in the treatment period compared to the pretreatment interval.²⁰ With these high doses, however, severe adverse effects were reported. As a result two of the seven patients were unable to complete the protocol.²⁰

In a double-blind, randomized controlled trial of retinol versus placebo in 2297 immunocompetent individuals with actinic keratoses, daily retinol was associated with a statistically significant reduction in the risk of developing squamous cell carcinomas.²¹ In contrast, low doses of isotretinoin were not effective in reducing basal cell carcinomas.²²

The adverse effect profile of synthetic retinoids resembles the symptoms associated with hypervitaminosis A and may hamper the use of these agents.¹² The most frequent systemic side effect is an increase in plasma triglycerides. Increases of total and LDL cholesterol levels are also frequently observed.²³ Other adverse effects are abnormalities of liver function tests, arthralgias, mucocutaneous xerosis, and alopecia.¹² In addition, treatment at high doses or for a long term has been associated with various skeletal abnormalities such as calcifications of tendons and ligaments around joints, hyperostosis of the spine, and osteoporosis.^{7,12} Retinoids are highly teratogenic if given orally during embryogenesis.¹² Isotretinoin's link to psychiatric disorders such as depression is still controversial.²⁴

Topical Retinoids

The advantage of topical retinoids is the avoidance of systemic toxicity. Good results were obtained in the treatment and chemoprevention of actinic keratoses in immunocompetent patients using topical retinoids, in particular isotretinoin and tretinoin.¹² Topical retinoids have not been systematically tested in the treatment or prevention of nonmelanoma skin cancer in immunocompetent individuals.¹²

Retinoid Chemoprevention in Organ Transplant Recipients

Systemic Retinoids

The tables provide a summary of the clinical studies with retinoids in organ transplant recipients that are discussed in this overview.

Initially, studies were performed with etretinate. Later etretinate was replaced by its active metabolite, acitretin.

The four studies summarized with etretinate were all without an explicit control group and showed a reduction of new skin cancers, compared to the pretreatment period (Table 1). The well-known side effects of retinoids were observed but were well manageable (Table 1). Specifically no deterioration of renal function was noted.^{15,25–27}

Study	Technical Details of Study*	Treatment Result	Side Effects
Shuttleworth et al. ²⁵	NCG, <i>n</i> = 6, 1 mg/kg/day, 6 months	5 patients, no new skin cancers; 1 patient, 2 new SCCs after 6 months.	Cheilitis, increased triglycerides ($n = 3$), increased cholesterol ($n = 2$); no deterioration in renal function.
Kelly et al. ²⁶	NCG, <i>n</i> = 4, 50 mg/day, 8–13 months	Considerable reduction new SCCs.	Mild mucocutaneous side effects, thrombocytopenia ($n = 1$); renal function unchanged.
Rook et al. ¹⁵	NCG, $n = 11$, 10 mg/day in combination with topically applied 0.025% tretinoin ($n = 7$) or tretinoin alone ($n = 4$), 9 months	3 of 4 patients, no new SCCs after 9 months of etretinate + topical tretinoin (2 of 3 patients, no new SCCs after 9 months of topical tretinoin alone); 4 patients, nonevaluable.	Mild mucocutaneous side effects.
Gibson et al. ²⁷	NCG, <i>n</i> = 11, 0.3 mg/kg/day, 17 months	10 patients, significant reduction of new skin cancer after 3 + 6 months compared to pretreatment period; 1 patient, nonevaluable.	Mild mucocutanous side effects, increased triglycerides ($n = 3$), increased cholesterol ($n = 1$).

 Table 1. Clinical Studies With Etretinate in Organ Transplant Recipients

*Respectively, the type of trial (NCG, study without any explicit control group), the study size, the dose of etretinate, and the duration of the study are depicted. Abbreviation: SCC, squamous cell carcinoma.

Table 2.	Clinical Studies	With Acitretin	in Organ	Transplant	Recipients
----------	-------------------------	----------------	----------	------------	------------

Study	Technical Details of Study*	Treatment Result	Side Effects
Vandeghinste et al. ³⁹	CR, <i>n</i> = 1, 0.5 mg/kg/day, 15 months	No new dysplastic skin lesions observed.	Mild mucocutaneous side effects.
Bouwes Bavinck et al. ²⁸	RCT (compared with placebo), n = 38, 30 mg/day, 6 months	Statistically significantly fewer patients with new skin cancers compared to placebo reduction of number of actinic keratoses.	Side effects: mild mucocutaneous, hair loss, increased cholesterol + triglycerides ($n = 3$); renal function unchanged.
Yuan et al. ³⁰	NCG, <i>n</i> = 15, 10–50 mg/day, <6 to >12 months	Variable effects on skin cancer.	Mild mucocutaneous side effects.
McKenna et al. ³¹	NCG, <i>n</i> = 16, 0.3 kg/mg/day, 5 years	Statistically significant reduction of new skin cancers after 4 years compared to pretreatment period.	Mild mucocutanous side effects, increased triglycerides ($n = 1$), increased cholesterol ($n = 1$).
McNamara et al. ³²	NCG, <i>n</i> = 5, 10–25 mg/day, 10–24 months	3 patients, significant decrease in new tumors compared to pretreatment period; 2 patients, moderate decrease in new tumors.	Mild mucocutaneous side effects.
George et al. ³³	RCT (compared to no therapy), n = 23, 25 mg/day, 2 year	Number of SCCs significantly lower on acitretin compared to drug-free period.	Cheilitis, headache, increase of musculoskeletal symptoms, gastritis, increased triglycerides.
de Sévaux et al. ²⁹	RCT (0.4 mg/kg/day compared to 0.2 mg/kg/day), <i>n</i> = 26, 0.4 or 0.2 mg/kg/day, 1 year	Decrease of actinic keratoses by 50% in both groups; no effect on development of skin cancers in both groups compared to pretreatment period.	Mild mucocutaneous side effects, mild hair loss.

*Respectively, the type of trial (CR, case report of single cases; RCT, randomized controlled trial; NCG, study without any explicit control group), the study size, the dose of acitretin, and the duration of the study are depicted.

Abbreviation: SCC, squamous cell carcinoma.

With acitretin, one placebo-controlled trial and two other randomized controlled trials were performed (Table 2). In addition, three studies without an explicit control group and one case report were published (Table 2). The randomized, double-blind, placebocontrolled trial showed a significant effect on the prevention of skin cancer at a acitretin dose of 30 mg/ day. During a 6-month period, 2 of 19 patients (11%) in the acitretin group reported a total of two new squamous cell carcinomas, compared with 9 of 19

patients (47%) in the placebo group who developed a total of 18 new carcinomas (p = 0.01).²⁸ Furthermore, the relative decrease of keratotic skin lesions was 13% in the acitretin group, compared to a relative increase of 28% in the placebo group (p < 0.01). After the end of treatment, however, skin cancers and keratotic skin lesions relapsed.²⁸ The other studies generally confirmed the favorable effect of acitretin on the reduction of development of new skin cancers and actinic keratoses.²⁹⁻³³ In the Australian open randomized crossover trial, 25 mg acitretin/day was prescribed to 14 patients, and 9 patients received no retinoid treatment. They crossed over after 1 year. A significantly lower number of squamous cell carcinomas was shown in patients while on acitretin compared to the drug-free period (p = 0.002).³³ A recent randomized study with low-dose acitretin was performed by de Sévaux et al.²⁹ in 26 renal transplant patients of whom 13 were treated with 0.4 mg/kg/day acitretrin and 13 with 0.2 mg/kg/day for 1 year. A decrease in actinic keratoses of 50% was reported in both groups. Nevertheless, no effect on the development of nonmelanoma skin cancer was reported in the two lowdose groups of patients when compared to the pretreatment period.29

As far as we know, there are a few reports about the use of isotretinoin in transplant patients with skin cancer.^{8,34} In one study with one patient a 50% reduction of new lesions compared to the pretreatment period was observed.³⁴

Side effects of retinoids in organ transplant recipients are identical to those reported in the immunocompetent host. In organ transplant recipients there is a theoretical concern of immunostimulation caused by the systemic retinoids, which theoretically could lead to rejection of the transplanted organ.⁸ Osteoporosis is also a particular problem for the posttransplant recipient, who is already at risk for osteoporosis from glucocorticoid exposure.⁵ Furthermore, elevation of serum lipids is more common in organ transplant recipients also without systemic retinoids. This problem is usually manageable by the use of antihyperlipemics.³⁵

Topical Retinoids

Two studies with topically applied tretinoin and one with topically applied adapalene are summarized in Table 3. A statistically significant reduction of actinic keratoses was observed in two studies.^{36,37} Side effects were generally mild.^{36–38} No significant effect was shown in the study of Smit et al.,³⁸ possibly because of the smaller sample size, the lower concentration of tretinoin, the shorter study duration, the complexity of the study, the focus on immunohistochemical and histologic parameters, or a combination of these factors. As far as we know, no studies with topically applied retinoids were performed to study the effect on the development of new skin cancers.

Personal Experience With Acitretin in Organ Transplant Recipients

Acitretin in a dose between 10 and 30 mg per day has, in our experience, a clear effect on the number and thickness of the hyperkeratotic skin lesions and the number of new skin cancers. Most of the approximately 75 patients we have treated in Leiden (The Netherlands) and Lyon (France) together could tolerate this regimen for a 4- to 6-month period. Approximately one-third of these patients were able to continue this regimen for at least 5 years without significant side effects. About half of the patients, however, have preferred to discontinue this treatment after 4 to 6 months because of dry skin, dry lips, significant hair loss, pruritus, or arthralgias. With

Study	Technical Details of Study*	Treatment Result	Side Effects
Euvrard et al. ³⁶	RCT (vs. placebo), $n = 22$, 0.05% tretinoin once daily, 3 months	Statistically significant reduction of keratotic lesions after 3 months compared to placebo (45% vs. 23%).	Local tolerance generally good.
Euvrard et al. ³⁷	RCT (vs. placebo), $n = 40$, 0.3% vs. 0.1% adapalene, 6 months	Significant decrease in actinic keratoses in 0.3% group compared to placebo (32% vs. 21%).	Skin irritation ($n = 1$).
Smit et al. ³⁸	CT (four treatment modalities), $n = 13$, 0.02% tretinoin vs. calcipotriol vs. both vs. emollient twice daily, 6 weeks	Treatment of actinic keratoses, alone or in combination with calcipotriol, showed no effect on clinical, immunohistochemical, and histologic measures.	Pruritus in all test areas (number unknown).

Table 3. Clinical Studies With Topical Retinoids in Organ Transplant Recipients

*Respectively, the type of trial (RCT, randomized controlled trial; CT, controlled trial), the study size, the dose of the topical retinoids, and the duration of the study are depicted.

regard to objective adverse effects, we have observed liver toxicity in two kidney transplant recipients necessitating treatment discontinuation, although we have not observed significant renal dysfunction or allograft rejection. Elevated serum triglycerides and cholesterol levels have been easily manageable. We have utilized intermittent therapy in some patients with alternating 1 to 3 months, with and without acitretin. This regimen seems to be effective, but randomized-controlled trials should be performed to prove the efficacy of this treatment regimen.

Recommendations and Conclusion

Systemic retinoids, specifically etretinate and acitretin, have shown to be effective in inhibiting the development of new dysplastic skin lesions and skin cancers in organ transplant recipients. Because etretinate is not marketed anymore, acitretin is the medication most often utilized and the recommendations in this overview therefore only pertain to acitretin. Experience with isotretinoin in organ transplant patients is limited. Nevertheless, it is probably the most widely studied retinoid in the chemoprevention of skin cancers in nonimmunosuppressed patients. The results in transplant patients are expected to be similar as compared to nonimmunosuppressed patients. We have no data to compare the antitumoral effect of either drug in transplant or in nontransplant patients. Isotretinoin could be useful in female patients because of the shorter time required to avoid pregnancy (1 month after the discontinuation of the treatment compared to acitretin (2 years)).

Systemic retinoids are variably tolerated by organ transplant patients. Because the benefit of acitretin is present only during treatment, long-term treatment usually is necessary. Long-term use of systemic retinoids, however, may be limited by the occurrence of side effects in individual patients. Specifically, subjective side effects, such as dry skin, dry lips, and hair loss, may be reason to discontinue therapy with acitretin, despite treatment with topical moisturizers.

Although no patients were reported in whom retinoid therapy affected the transplanted organ, the possible immune-stimulating effect of retinoids causes some theoretical concern for rejection of the transplanted organ. Use of acitretin therefore should in our opinion be limited to patients with dozens of hyperkeratotic skin lesions and/or at least one squamous cell carcinoma in the medical history, but other levels of involvement of the skin have also been suggested.

Because the minimal effective dose varies greatly between patients, it may be best to start at a low dose (between 10 and 30 mg/day), which can be increased as tolerated if insufficient benefit is observed.⁵ In our experience a dose of 30 mg/day acitretin or less was sufficiently effective. When treatment is limited by subjective side effects, intermittent acitretin treatment could be considered with periods on acitretin therapy intermingled with treatment-free periods. In addition to the intermittent acitretin regimen, lowering the dose of acitretin is an alternative method of decreasing the side effects.

Patients should be monitored for the usual measures when using oral retinoids. Triglyceride levels, cholesterol levels, and transaminases should be tested before the start of the treatment and at regular intervals during treatment, initially monthly and later at 3month intervals. In renal transplant recipients, renal function should be monitored. Radiologic examination of the spine and long bones could be performed in patients with long-term use of retinoids but, in our opinion, is not useful in organ transplant recipients, because osseous side effects of treatment with corticosteroids can be expected.

Although topical retinoids seem effective in the treatment of actinic keratoses and warts in organ transplant patients, the effect is clearly less than with treatment with systemic retinoids. Topical retinoids may be warranted as first-line therapy in high-risk patients once actinic keratoses begin to develop, but before a critical threshold necessitating oral retinoids is reached.

References

- 1. London NJ, Farmery SM, Will EJ, Davison AM, Lodge JP. Risk of neoplasia in renal transplant patients. Lancet 1995;346:403-6.
- 2. Euvrard S, Kanitakis J, Claudy A. Skin cancers after organ transplantation. N Engl J Med 2003;348:1681–91.
- 3. Hartevelt MM, Bouwes Bavinck JN, Kootte AM, Vermeer BJ, Vandenbroucke JP. Incidence of skin cancer after renal transplantation in The Netherlands. Transplantation 1990;49:506–9.
- 4. Bouwes Bavinck JN, Hardie DR, Green A, et al. The risk of skin cancer in renal transplant recipients in Queensland, Australia: a follow-up study. Transplantation 1996;61:715–21.
- DiGiovanna JJ. Posttransplantation skin cancer: scope of the problem, management, and role for systemic retinoid chemoprevention. Transplant Proc 1998;30:2771–5.
- van Zuuren EJ, Posma AN, Scholtens RE, et al. Resurfacing the back of the hand as treatment and prevention of multiple skin cancers in kidney transplant recipients. J Am Acad Dermatol 1994; 31(5 Pt 1):760–4.
- 7. DiGiovanna JJ. Retinoid chemoprevention in patients at high risk for skin cancer. Med Pediatr Oncol 2001;36:564–7.
- 8. Euvrard S, Kanitakis J, Thivolet J, Claudy A. Retinoids for the management of dermatological complications of organ transplantation. Biodrugs 1997;8:176–84.
- 9. DiGiovanna JJ. Retinoids for the future: oncology. J Am Acad Dermatol 1992;27(6 Pt 2):S34–7.
- 10. Niles RM. Recent advances in the use of vitamin A (retinoids) in the prevention and treatment of cancer. Nutrition 2000;16:1084–9.
- 11. Altucci L, Gronemeyer H. The promise of retinoids to fight against cancer. Nat Rev Cancer 2001;1:181–93.

- Orfanos CE, Zouboulis CC, Almond-Roesler B, Geilen CC. Current use and future potential role of retinoids in dermatology. Drugs 1997;53:358–88.
- Bartsch D, Boye B, Baust C, zur HH, Schwarz E. Retinoic acidmediated repression of human papillomavirus 18 transcription and different ligand regulation of the retinoic acid receptor beta gene in non-tumorigenic and tumorigenic HeLa hybrid cells. EMBO J 1992;11:2283–91.
- Khan MA, Jenkins GR, Tolleson WH, Creek KE, Pirisi L. Retinoic acid inhibition of human papillomavirus type 16-mediated transformation of human keratinocytes. Cancer Res 1993;53:905–9.
- Rook AH, Jaworsky C, Nguyen T, et al. Beneficial effect of lowdose systemic retinoid in combination with topical tretinoin for the treatment and prophylaxis of premalignant and malignant skin lesions in renal transplant recipients. Transplantation 1995;59: 714–9.
- 16. Gibson GE, O'Grady A, Kay EW, Leader M, Murphy GM. Langerhans cells in benign, premalignant and malignant skin lesions of renal transplant recipients and the effect of retinoid therapy. J Eur Acad Dermatol Venereol 1998;10:130–6.
- 17. Blokx WA, Smit JV, de Jong EM, et al. Retinoids strongly and selectively correlate with keratin 13 and not keratin 19 expression in cutaneous warts of renal transplant recipients. Arch Dermatol 2002;138:61–5.
- De Lacharriere O, Escoffier C, Gracia AM, et al. Reversal effects of topical retinoic acid on the skin of kidney transplant recipients under systemic corticotherapy. J Invest Dermatol 1990;95:516–22.
- 19. Sankaranarayanan R, Mathew B. Retinoids as cancer-preventive agents. IARC Sci Publ 1996;139:47–59.
- Kraemer KH, DiGiovanna JJ, Moshell AN, Tarone RE, Peck GL. Prevention of skin cancer in xeroderma pigmentosum with the use of oral isotretinoin. N Engl J Med 1988;318:1633–7.
- Moon TE, Levine N, Cartmel B, Bangert JL. Retinoids in prevention of skin cancer. Cancer Lett 1997;114:203–5.
- 22. Tangrea JA, Edwards BK, Taylor PR, et al. Long-term therapy with low-dose isotretinoin for prevention of basal cell carcinoma: a multicenter clinical trial. Isotretinoin-Basal Cell Carcinoma Study Group. J Natl Cancer Inst 1992;84(5):328–32.
- 23. Staels B. Regulation of lipid and lipoprotein metabolism by retinoids. J Am Acad Dermatol 2001;45:158–67.
- Wysowski DK, Pitts M, Beitz J. An analysis of reports of depression and suicide in patients treated with isotretinoin. J Am Acad Dermatol 2001;45:515–9.
- Shuttleworth D, Marks R, Griffin PJ, Salaman JR. Treatment of cutaneous neoplasia with etretinate in renal transplant recipients. Q J Med 1988;68:717–25.

- 26. Kelly JW, Sabto J, Gurr FW, Bruce F. Retinoids to prevent skin cancer in organ transplant recipients. Lancet 1991;338:1407.
- Gibson GE, O'Grady A, Kay EW, Murphy GM. Low-dose retinoid therapy for chemoprophylaxis of skin cancer in renal transplant recipients. J Eur Acad Dermatol Venereol 1998;10:42–7.
- Bouwes Bavinck JN, Tieben LM, Van der Woude FJ, et al. Prevention of skin cancer and reduction of keratotic skin lesions during acitretin therapy in renal transplant recipients: a doubleblind, placebo-controlled study. J Clin Oncol 1995;13:1933–8.
- 29. de Sevaux RG, Smit JV, de Jong EM, van de Kerkhof PC, Hoitsma AJ. Acitretin treatment of premalignant and malignant skin disorders in renal transplant recipients: clinical effects of a randomized trial comparing two doses of acitretin. J Am Acad Dermatol 2003;49:407–12.
- Yuan ZF, Davis A, Macdonald K, Bailey RR. Use of acitretin for the skin complications in renal transplant recipients. N Z Med J 1995; 108:255–6.
- McKenna DB, Murphy GM. Skin cancer chemoprophylaxis in renal transplant recipients: 5 years of experience using low-dose acitretin. Br J Dermatol 1999;140:656–60.
- McNamara IR, Muir J, Galbraith AJ. Acitretin for prophylaxis of cutaneous malignancies after cardiac transplantation. J Heart Lung Transplant 2002;21:1201–5.
- George R, Weightman W, Russ GR, Bannister KM, Mathew TH. Acitretin for chemoprevention of non-melanoma skin cancers in renal transplant recipients. Australas J Dermatol 2002;43: 269–73.
- Bellman BA, Eaglstein WH, Miller J. Low dose isotretinoin in the prophylaxis of skin cancer in renal transplant patients. Transplantation 1996;61:173.
- Andany MA, Kasiske BL. Dyslipidemia and its management after renal transplantation. J Nephrol 2001;14(Suppl 4):S81–8.
- Euvrard S, Verschoore M, Touraine JL, et al. Topical retinoids for warts and keratoses in transplant recipients. Lancet 1992;340: 48–9.
- Euvrard S, Kanitakis J, Claudy A. Topical retinoids for the management of dysplastic epithelial lesions. In: Skin Diseases after Organ Transplantation. Montrouge: John Libbey Eurotext; 1998. p. 175–82.
- 38. Smit JV, Cox S, Blokx WA, et al. Actinic keratoses in renal transplant recipients do not improve with calcipotriol cream and all-trans retinoic acid cream as monotherapies or in combination during a 6-week treatment period. Br J Dermatol 2002;147:816–8.
- Vandeghinste N, De Bersaques J, Geerts ML, Kint A. Acitretin as cancer chemoprophylaxis in a renal transplant recipient. Dermatology 1992;185:307–8.

CHAPTER 6

Photodynamic therapy does not prevent cutaneous Squamous-cell carcinoma in organ-transplant recipients: results of a randomized-controlled trial

Journal of Investigative Dermatology 2006; 126: 569-574

Photodynamic Therapy does not Prevent Cutaneous Squamous-Cell Carcinoma in Organ-Transplant Recipients: Results of a Randomized-Controlled Trial

Ymke G.L. de Graaf¹, Cornelis Kennedy¹, Ron Wolterbeek², Annemie F.S. Collen¹, Rein Willemze¹ and Jan N. Bouwes Bavinck¹

A randomized-controlled trial with paired observations was performed with 40 organ-transplant recipients to assess the preventive effect of photodynamic therapy (PDT) on the development of new squamous-cell carcinomas and to evaluate the effect of PDT on the number of keratotic skin lesions. The treatment area consisted of a randomly assigned forearm and the corresponding hand, whereas the other forearm and hand served as the control area. After the initial visit, follow-up visits were scheduled at 3-monthly intervals during 2 years. No statistically significant difference was found in the occurrence of new squamous-cell carcinomas between the treated and untreated arms: after 2 years of follow-up, we observed 15 squamous-cell carcinomas in nine out of 40 PDT-treated arms and 10 squamous-cell carcinomas in nine out of 40 control arms. The number of keratotic skin lesions increased in both arms, but was less pronounced in the PDT-treated arm. After 1 year of follow-up, a trend in favor of the PDT-treated arm was observed, but statistical significance was not reached. Nearly 80% of the patients reported mild to severe adverse effects consisting of pain and a burning sensation, immediately after the treatment. No long-term adverse events were noted. In conclusion, PDT does not appear to prevent the occurrence of new squamous-cell carcinomas in organ-transplant recipients, but to some degree, reduces the increase of keratotic skin lesions.

Journal of Investigative Dermatology (2006) 126, 569-574. doi:10.1038/sj.jid.5700098; published online 5 January 2006

INTRODUCTION

Skin carcinomas, especially squamous-cell carcinomas, that develop on sun-exposed skin, are a serious hazard to organtransplant recipients (London *et al.*, 1995; Euvrard *et al.*, 2003). The frequency of skin carcinomas increases progressively with time after transplantation. In the Netherlands, the risk to develop squamous-cell carcinoma increases up to 40%, 20 years after transplantation (Hartevelt *et al.*, 1990).

The majority of organ-transplant recipients develop multiple squamous-cell carcinomas, which appear to be more aggressive in these patients in comparison with immuno-competent individuals (Euvrard *et al.*, 2003).

An association exists between squamous-cell carcinoma and multiple keratotic skin lesions, and most lesions are localized on sun-exposed skin such as the forearms and dorsum of the hands (Bouwes Bavinck *et al.*, 1993; Euvrard *et al.*, 2003).

Abbreviations: ALA, aminolevulinic acid; PDT, photodynamic therapy Received 20 June 2005; revised 17 August 2005; accepted 10 October 2005; published online 5 January 2006 Currently available therapies for skin carcinomas such as excision and for solar keratoses such as topical application of liquid nitrogen are less satisfactory for the treatment of large affected areas, which is often the case in organ-transplant recipients.

Prevention of squamous-cell carcinomas and reduction of keratotic skin lesions with topical photodynamic therapy (PDT) would, therefore, substantially improve the quality of life of organ-transplant recipients.

PDT is a relatively safe procedure where a photosensitizer is applied to the affected area and subsequently irradiated with a light system. This treatment can be used to treat superficial skin carcinomas or precancerous lesions that are accessible to light (Morton *et al.*, 2002). PDT involves the activation of intracellular photosensitizers by visible light in order to generate cytotoxic singlet oxygen and other free radicals, which selectively destroy rapidly proliferating cells (Hopper, 2000; Ormrod and Jarvis, 2000). As photosensitizers, aminolevulinic acid (ALA) or methylaminolevulinic acid are used (Morton *et al.*, 2002).

PDT with topical ALA is a safe and effective treatment for solar keratoses (Ormrod and Jarvis, 2000; Brown *et al.*, 2004). The efficacy has been demonstrated for non-hyperkeratotic solar keratoses on the face or scalp of immunocompetent individuals. Response rates were comparable with cryotherapy and 5-fluorouracil, but a better cosmetic result was

¹Department of Dermatology, Leiden University Medical Center, Leiden, The Netherlands and ²Department of Biostatistics, Leiden University Medical Center, Leiden, The Netherlands

Correspondence: Dr Jan N. Bouwes Bavinck, Department of Dermatology, Leiden University Medical Center, PO Box 9600, 2300 RC Leiden, The Netherlands. E-mail: J.N.Bouwes_Bavinck@lumc.nl

reached (Morton *et al.*, 2002). Furthermore, Bowen's disease and superficial basal-cell carcinomas can also be effectively treated with PDT (Varma *et al.*, 2001; Kormeili *et al.*, 2004). PDT has also been shown to be effective in treating recalcitrant viral warts (Stender *et al.*, 2000; Ibbotson, 2002).

Only two studies concerning the efficacy of PDT in immunosuppressed patients are available. Although lower cure rates are reported in organ-transplant recipients compared to the immunocompetent controls (Dragieva *et al.*, 2004a), both studies report a good efficacy of PDT for solar keratoses in organ-transplant recipients (Dragieva *et al.*, 2004a, b). In addition, experimental studies in hairless mice have shown that PDT can delay the development of UVinduced skin carcinomas (Stender *et al.*, 1997; Sharfaei *et al.*, 2002).

The primary objective of this trial was to assess the occurrence of new squamous-cell carcinomas in organtransplant recipients in the treated arm compared with the control arm following PDT. A second objective was to evaluate the difference in (re)occurrence of keratotic skin lesions in the arm treated with PDT compared to the control arm. Finally, we studied the difference in efficacy between a regimen with one PDT treatment or two PDT treatments with a 6-month period in between the treatments.

RESULTS

A total of 45 patients gave informed consent. Forty of them were randomized two-fold: 21 to receive PDT to the left forearm and hand, 19 to receive PDT to the right forearm and hand and 23 patients were randomized to one PDT treatment (at T0) and 17 to two PDT treatments (at T0 and T6). The other five patients had been included previously in the non-randomized pilot part of the study and they received only one PDT treatment (at T0).

The characteristics of the patients are listed in Table 1. During the 2-year follow-up period, seven patients were lost to follow-up. Five patients died from another cause then skin cancer after 3 months (two patients), and after 9, 12, and 21 months, respectively. Two patients were lost to follow-up after 18 and 24 months, respectively, without any apparent reason. The characteristics of the treated and control arm at the start of the trial are depicted in Table 2.

During the 2-year follow-up period, a total of 25 squamous-cell carcinomas and no other types of skin cancer were detected on the forearms and hands of the 40 patients in the randomized part of the trial. Figure 1b shows the distribution of the clinical outcome among these patients. In six patients one squamous-cell carcinoma developed in the control arm and none in the PDT-treated arm, and in one patient two squamous-cell carcinomas developed in the control arm and none in the PDT-treated arm. By contrast, in six patients one squamous-cell carcinoma developed in the PDT-treated arm and none in the control arm, in one patient three squamous-cell carcinomas developed in the PDTtreated arm and none in the control arm, and in one patient five squamous-cell carcinomas developed in the PDT-treated arm and one in the control arm. In one patient one squamous-cell carcinoma developed in the PDT-treated

	Randomized trial	Pilot trial
No. of patients	40	5
Sex		
Male	21	2
Female	19	3
Age (years)		
Mean \pm SD	55.0 ± 8.8	58.0 ± 6.9
Range	39–71	50–68
Immunosuppressive treatment		
Prednisone and azathioprine	37	5
Prednisone and cyclosporine	1	0
Prednisone, azathioprine, and cyclosporine	2	0
Time after transplantation (years)		
Mean \pm SD	22.0 ± 6.3	17.9 ± 3.3
Range	7–34	12–21
History of SCC anywhere on the body		
No. of patients with SCC (%)	31 (78)	4 (80)
Mean no. of SCC \pm SD	8.2 ± 9.9	6.8 ± 6.5
Range	0–42	0–17

Table 1. Characteristics of the patients

SCC: squamous-cell carcinoma; SD: standard deviation.

Table 2. Characteristics of the PDT-treated and the control arm and hand at the start of the trial

	PDT-treated		
	arm	Control arm	
Randomized part of the trial			
No. of arms	40	40	
Side of the arm			
Left	19	21	
Right	21	19	
History of SCC per arm			
No. of arms with SCC (%)	18 (45)	19 (48)	
Mean no. of SCC \pm SD	1.0 ± 1.3	1.1 ± 1.7	
Range	0-4	0–8	
No. of keratotic lesions per arm			
Mean \pm SD	31 ± 22	27 ± 19	
Range	9–96	6–103	
Pilot part of the trial			
No. of arms	5	5	
Side of the arm			
Left	2	3	
Right	3	2	
History of SCC per arm			
No. of arms with SCC (%)	2 (40)	2 (40)	
Mean no. of SCC \pm SD	0.6 ± 0.9	0.4 ± 0.5	
Range	0–2	0–1	

SCC: squamous-cell carcinoma; SD: standard deviation.



Figure 1. Expected and observed number of patients after 2 years of followup. (a) Expected number of patients with 95% confidence intervals based on the assumption that after a 2-year follow-up period, 5% of the PDT-treated arms and 25% of the control arms would develop one or more squamous-cell carcinomas. The numbers are provided for a total of 40 patients (80 arms). (b) Observed number of patients with squamous-cell carcinoma with 95% confidence intervals after a 2-year follow-up period, showing that the control arm was better than the PDT-treated arm in eight (20%) of the 40 patients and the PDT-treated arm was better in seven (17.5%) of the 40 patients.

arm and one in the control arm and in 24 patients no squamous-cell carcinomas developed in any arm. Altogether, 15 squamous-cell carcinomas developed in nine out of 40 PDT-treated arms and 10 squamous-cell carcinomas in nine out of 40 control arms.

In the pilot study, four squamous-cell carcinomas developed in two out of five patients: one in the PDT-treated arm and three in the control arm. One patient developed one squamous-cell carcinoma in the control arm and none in the PDT-treated arm and one patient developed two squamouscell carcinomas in the control arm and one in the PDTtreated arm.

The occurrence of new squamous-cell carcinomas was not statistically significantly different between the PDT-treated arm and the control arm within the 2-year follow-up period neither in the randomized part of the study (P=0.80) nor in the pilot study (P=0.16) and also not when the outcomes of both studies were combined (P=0.81). One or two PDT treatments did not influence the outcome of squamous-cell carcinoma significantly (data not shown).

Figure 2 depicts the distribution and mean number of keratotic skin lesions in the control arms (panel a) and in the PDT-treated arms (panels b–d). Panel b shows the distribution and mean number of keratotic skin lesions of all PDT-treated arms, irrespective of whether they were treated one time or two times, panel c shows the arms that received one PDT treatment only, and panel d shows the arms that received two PDT treatments only.

Unexpectedly, at the start of the study, we found a relevant difference between the number of keratotic skin lesions in the arms randomized to be treated with PDT and the control arm with a mean of 4.5 more keratotic skin lesions in the arms that were randomly selected to be treated with PDT (Figure 2e). For this reason, all time points were compared with baseline (T0) as depicted in Figure 2f.

Planned pairwise comparison on the basis of a linear mixed model approach focused on the comparison between time points T3, T6, T9, and T12 versus T0 resulted in a statistically significant difference at time points T9 and T12 (Figure 2f). The overall fixed effect of time, however, was not statistically significant (P=0.06).

Two PDT treatments appeared to reduce the increase of keratotic skin lesions at time points T9 and T12 slightly more than one PDT treatment (Figure 2c and d), but using a summary measure approach based on Wilcoxon's rank-sum test on regression slopes from simple linear regression of individual patients, the difference in the median values of the regression slopes was not statistically significant (P=0.39).

In total, 79% of patients reported adverse effects during PDT treatment or in the first week after treatment. Most prevalent were a burning sensation (38%) and pain (31%). A minority of patients reported itch (9%) and blisters (2%). No long-term adverse effects were reported. There were no withdrawals of the patients because of adverse effects.

DISCUSSION

PDT, using topical δ -ALA and violet light (400–450 nm) applied to the forearm and corresponding dorsum of the hand, did not significantly prevent the development of new squamous-cell carcinomas in organ-transplant recipients within a 2-year follow-up period. The PDT procedure, however, to some extent, diminished the increase of keratotic skin lesions in the PDT-treated arm and hand compared to the nontreated control area, but at the expense of pain and irritation during and shortly after the procedure.

Generally, the forearms and dorsum of the hands of organtransplant recipients have undergone a field change, meaning that the skin in this area shows histological atypia (Berg and Otley, 2002). This may be clinically visible by the presence of numerous premalignant actinic keratoses and other keratotic skin lesions and results in an increased risk to develop squamous-cell carcinomas in this area of the skin. Hypothetically, treating the whole area with PDT and thus selectively destroying the premalignant cells should result in a reduced risk of squamous-cell carcinoma. Unfortunately, such an effect was not observed in this randomizedcontrolled trial.

A possible explanation that we did not observe a preventive effect of the PDT procedure on the development of new squamous-cell carcinomas may be that the follow-up period of 2 years was too short. However, after 2 years of follow-up, nine out of 40 patients (22.5%) developed new squamous-cell carcinomas in the control arm, which is close to the approximately 25% of patients who were expected to develop new squamous-cell carcinomas in the control arm during the 2-year follow-up period. In addition, 23 out of 40



Figure 2. Distribution and mean number of keratotic skin lesions at the start of the trial and at 3-monthly intervals (a) in the control arm, (b) in the PDT-treated arm, irrespective of one or two treatments, (c) with one treatment only, and (d) with two treatments only. (e) Uncorrected difference between the number of keratotic skin lesions in the PDT-treated arm and the control arm and (f) difference corrected for the difference at the start of the trial are shown. The bars indicate 95% confidence intervals.

patients (57.5%) had developed a total of 84 squamous-cell carcinomas on the rest of the body during the 2-year follow-up period (data not shown), indicating that a 2-year follow-up period should be sufficient to harbor enough squamous-cell carcinoma events in the PDT-treated arm and the control arm. If the PDT procedure had been effective, a significantly lower occurrence of squamous-cell carcinoma would have been observed in the PDT-treated arm, which, obviously, was not the case.

Another possible explanation that the PDT procedure was not effective may be the selection of the patients. We selected patients who needed intervention the most, that is, patients who often had a history of one or more squamous-cell carcinomas and at least 10 hyperkeratotic skin lesions present on the forearms and dorsum of the hands. This selection is illustrated by the fact that the time period since transplantation ranged between 7 and 34 years. We cannot exclude that PDT intervention at an earlier time period after transplantation, at an earlier stage of skin cancer development, may be more effective.

Still another possible explanation that the PDT procedure was not successful in our hands may be the character of the light source and the photosensitizer that we used. Red light (570–750 nm) is often used as the light source for PDT because of the greater depth of penetration. We used violet light (400–450 nm) because violet light has a maximum overlap with the excitation spectrum of protoporphyrin IX, and, therefore less light energy is required for the same effect (Dijkstra *et al.*, 2001). As photosensitizer we used δ -ALA. Methylaminolevulinic acid, the methyl ester of ALA, has the advantage of penetrating deeper into the skin, and possibly could result in a better therapeutic result.

Additionally, we did not pretreat the keratotic skin lesions with curettage, because this was not feasible considering the large amount of keratotic skin lesions in our patients. As a result, thick keratotic skin lesions were possibly not optimally treated with PDT, because of insufficient penetration of the photosensitizer in these lesions. Inferior response to PDT of thick keratotic skin lesions on the hands compared to the thinner lesions on the face has been demonstrated in immunocompetent patients (Morton *et al.*, 2002), and in immunocompromised patients it is probably even more difficult to treat these keratotic skin lesions with PDT (Dragieva *et al.*, 2004a).

Still another explanation that the PDT procedure may not have been successful may be that the favorable effect of PDT was counterbalanced by a harmful effect of the violet light source we used, but data to substantiate this suggestion are not available.

Whereas the occurrence of new squamous-cell carcinomas is a solid end point of this study, the number of keratotic skin lesions is a much weaker end point. Counting keratotic skin lesions is difficult to standardize; observers tend to improve their counting scales during time, and intra-observer and inter-observer variation may be significant. The advantage of paired analyses is that intra-observer variation is minimized and variation during time is reduced, because inaccurate counting should equally affect both the treated and control arms in all patients. Despite the methodological limitations, a trend to reduction of the increase of keratotic skin lesions in the PDT-treated arm and hand compared to the nontreated control area was observed. With a more accurate measurement of keratotic skin lesions, this effect is likely to be more discernible and would possibly reach statistical significance.

In summary, PDT using topical δ -ALA with a violet light source (400-450 nm) does not appear to prevent the development of new squamous-cell carcinomas in organtransplant recipients. PDT has also no significant effect on the reduction of keratotic skin lesions although a trend in favor of the PDT-treated arm was observed. A possible positive effect of PDT, however, goes at the expense of significant short-term side effects, and, therefore, PDT performed with topical δ -ALA and violet light does not appear to be a promising preventive therapy in organtransplant recipients.

MATERIALS AND METHODS Eligible patients

Organ-transplant recipients who were regularly evaluated at the Department of Dermatology and/or Nephrology from the Leiden University Medical Center in the period between August 2000 and February 2003 were eligible to participate in the trial.

Inclusion criteria were a history of skin carcinoma and/or at least 10 keratotic skin lesions present on both forearms and hands at the day of inclusion; an age of 18 years or older; and a functioning graft of 5 years or longer.

Keratotic skin lesions consisted of solar keratoses, flat warts, seborheic warts, and common warts. Because it is difficult to discriminate these different keratotic skin lesions on clinical and histological grounds, they were considered together. Patients who presented with a skin carcinoma at the start of the trial were only included after excision of the skin cancer, which needed to be fully healed before topical application with the photosensitizer. Patients who were using acitretin and women with childbearing potential were excluded from the study.

The medical ethical committee of the Leiden University Medical Center approved the study, and all participants provided written informed consent. The study adheres to the Declaration of Helsinki Principles and was approved by the local medical ethics commission.

Study design

A randomized-controlled trial with a self-controlled design, consisting of a right/left comparison was performed. One forearm and the corresponding hand were randomly allocated to the PDT procedure, and the other forearm and hand served as the control area and remained untreated. In addition, patients were randomly allocated to one or two PDT procedures. The first group received only one treatment at the start of the trial (T0), whereas the second group received a second PDT procedure 6 months later (T6).

The clinical pharmacy from our hospital performed the randomization procedure. A computer program automatically generated a randomization list with a study number and the patients were randomized accordingly. Owing to the nature of the treatment, blinding of the patients was not possible. The physician, however, was blinded for the treatment arm at the follow-up visits and the patients were requested not to inform the physician which arm had been treated. Follow-up visits were scheduled every 3 months during 2 years. At the start of the trial and at each follow-up visit, the skin of both forearms and hands was checked for the presence of squamouscell carcinomas and any other possible type of skin cancer. All newly suspected lesions were evaluated histologically. Only histologically confirmed squamous-cell carcinomas were included in the study. A follow-up time of 24 months was completed to evaluate the occurrence of new squamous-cell carcinomas. The numbers of keratotic skin lesions on both forearms and corresponding hands were counted during a follow-up time of 12 months. To minimize inter-observer variation, the same physician evaluated each patient at all visits, with few exceptions.

The randomized-controlled trial had been preceded by an open non-randomized-pilot phase with five different organ-transplant recipients to optimize the PDT procedure itself. The same inclusion and exclusion criteria had been applied to the pilot study as for the later randomized-controlled trial. In the pilot study, only new skin cancers were evaluated during follow-up.

PDT treatment

The PDT treatment was adapted from a protocol used by our colleagues in Utrecht, The Netherlands (Dijkstra *et al.*, 2001). The clinical pharmacy from our hospital freshly produced the ALA formulation for each patient visit. Patients received topical cream containing 200 mg δ -ALA HCl per 1 g of Lanette cream base on the randomly allocated forearm and hand. After application of the cream, the forearm and hand were covered with a Tegaderm dressing, which was applied for a duration of 4 hours. The dressing was removed shortly before the irradiation procedure. The light source we used produced a wavelength band of 400–450 nm (Philips HPM-10, 400 W) (Dijkstra *et al.*, 2001). The duration of the irradiation procedure was 17 minutes, resulting in a total light dose of 5.5–6 J/cm². The patients were instructed to cover the treated arm

and hand after the PDT procedure for the rest of the day to avoid extra irradiation.

We decided not to pretreat the keratotic skin lesions with curettage, because this was not feasible considering the large amount of keratotic skin lesions in our patients. Other "field" treatments, such as 5-fluorouracil or imiquimod cream, were not allowed during the 2-year post-treatment period.

Statistical analysis

The sample size was calculated based on the findings of a previous double-blind placebo-controlled study assessing the efficacy of acitretin in the prevention of new squamous-cell carcinomas (Bouwes Bavinck *et al.*, 1995). We had observed that in the placebo group within a 6-month period, nine out of 19 patients had developed altogether 15 new squamous-cell carcinomas anywhere on the body.

Based on these findings, it was expected that, within a 2-year follow-up period, at least 50% of the organ-transplant recipients would develop one or more squamous-cell carcinomas on both forearms or hands (25% per arm and hand). In order to have an effective preventive treatment, we stipulated that only 5% of the treated arms were allowed to develop new squamous-cell carcinomas compared to 25% in the control arm during the same 2-year follow-up period. With a power of 90% and a significance level of 0.05, we calculated that 45 patients would be sufficient to distinguish significantly between a skin cancer occurrence (at the 2-year follow-up) of 5% in the treated arm and 25% in the untreated arm (Figure 1a).

For the statistical analyses, we used SPSS version 12.0.1 for Windows. The difference regarding the number of new squamouscell carcinomas between the PDT-treated and the control arm was calculated with a Wilcoxon's signed rank test.

For the effect of the PDT treatment on the number of keratotic skin lesions, we used a linear mixed model approach with fixed time effects for repeated measurements based on the differences in the number of keratotic skin lesions between the PDT-treated arm and hand and the control arm and hand at different time points. We used a summary measure approach based on Wilcoxon's rank-sum test on regression slopes from simple linear regression of individual patients to assess whether there was a difference in the time course of change of number of lesions for arms and hands receiving one or two PDT treatments after 9 (T9) and 12 months (T12) of follow-up.

CONFLICT OF INTEREST

The authors state no conflict of interest.

ACKNOWLEDGMENTS

We thank Huib van Weelden for providing the light source, Joost Govaert and Nelleke van der Zwan for assistance in monitoring the patients, Jo Hermans for assisting in the study design, Judith Wessels for randomization and providing the ALA cream, and Hans de Fijter for referring some of the patients. This research was financially supported by a grant from ZON MW (The Netherlands Organization for Health Research and Development: 98-1-552, http://www.zonmw.nl). No support from the Industry was obtained.

REFERENCES

- Bouwes Bavinck JN, De Boer A, Vermeer BJ, Hartevelt MM, van der Woude FJ, Claas FH *et al.* (1993) Sunlight, keratotic skin lesions and skin cancer in renal transplant recipients. *Br J Dermatol* 129:242–9
- Bouwes Bavinck JN, Tieben LM, van der Woude FJ, Tegzess AM, Hermans J, ter Schegget J et al. (1995) Prevention of skin cancer and reduction of keratotic skin lesions during acitretin therapy in renal transplant recipients: a double-blind, placebo-controlled study. J Clin Oncol 13:1933–8
- Berg D, Otley CC (2002) Skin cancer in organ transplant recipients: epidemiology, pathogenesis, and management. J Am Acad Dermatol 47:1–17
- Brown SB, Brown EA, Walker I (2004) The present and future role of photodynamic therapy in cancer treatment. *Lancet Oncol* 5:497–508
- Dijkstra AT, Majoie IM, van Dongen JW, van Weelden H, van Vloten WA (2001) Photodynamic therapy with violet light and topical 6-aminolaevulinic acid in the treatment of actinic keratosis, Bowen's disease and basal cell carcinoma. *J Eur Acad Dermatol Venereol* 15:550-4
- Dragieva G, Hafner J, Dummer R, Schmid-Grendelmeier P, Roos M, Prinz BM et al. (2004a) Topical photodynamic therapy in the treatment of actinic keratoses and Bowen's disease in transplant recipients. *Transplantation* 77:115–21
- Dragieva G, Prinz BM, Hafner J, Dummer R, Burg G, Binswanger U *et al.* (2004b) A randomized controlled clinical trial of topical photodynamic therapy with methyl aminolaevulinate in the treatment of actinic keratoses in transplant recipients. *Br J Dermatol* 151:196–200
- Euvrard S, Kanitakis J, Claudy A (2003) Skin cancers after organ transplantation. N Engl J Med 348:1681–91
- Hartevelt MM, Bouwes Bavinck JN, Kootte AM, Vermeer BJ, Vandenbroucke JP (1990) Incidence of skin cancer after renal transplantation in The Netherlands. *Transplantation* 49:506–9
- Hopper C (2000) Photodynamic therapy: a clinical reality in the treatment of cancer. *Lancet Oncol* 1:212–9
- Ibbotson SH (2002) Topical 5-aminolaevulinic acid photodynamic therapy for the treatment of skin conditions other than non-melanoma skin cancer. *Br J Dermatol* 146:178–88
- Kormeili T, Yamauchi PS, Lowe NJ (2004) Topical photodynamic therapy in clinical dermatology. *Br J Dermatol* 150:1061–9
- London NJ, Farmery SM, Will EJ, Davison AM, Lodge JP (1995) Risk of neoplasia in renal transplant patients. *Lancet* 346:403–6
- Morton CA, Brown SB, Collins S, Ibbotson S, Jenkinson H, Kurwa H *et al.* (2002) Guidelines for topical photodynamic therapy: report of a workshop of the British Photodermatology Group. *Br J Dermatol* 146:552-67
- Ormrod D, Jarvis B (2000) Topical aminolevulinic acid HCl photodynamic therapy. Am J Clin Dermatol 1:133-9
- Sharfaei S, Juzenas P, Moan J, Bissonnette R (2002) Weekly topical application of methyl aminolevulinate followed by light exposure delays the appearance of UV-induced skin tumours in mice. Arch Dermatol Res 294:237–42
- Stender IM, Bech-Thomsen N, Poulsen T, Wulf HC (1997) Photodynamic therapy with topical delta-aminolevulinic acid delays UV photocarcinogenesis in hairless mice. *Photochem Photobiol* 66:493–6
- Stender IM, Na R, Fogh H, Gluud C, Wulf HC (2000) Photodynamic therapy with 5-aminolaevulinic acid or placebo for recalcitrant foot and hand warts: randomised double-blind trial. *Lancet* 355:963–6
- Varma S, Wilson H, Kurwa HA, Gambles B, Charman C, Pearse AD *et al.* (2001) Bowen's disease, solar keratoses and superficial basal cell carcinomas treated by photodynamic therapy using a large-field incoherent light source. *Br J Dermatol* 144:567–74

CHAPTER 7

The occurrence of residual or recurrent squamous-cell carcinomas in organ-transplant recipients after curettage and electrodessication

British Journal of Dermatology 2006; 154: 493-497

The occurrence of residual or recurrent squamous cell carcinomas in organ transplant recipients after curettage and electrodesiccation

Y.G.L. de Graaf, V.R. Basdew, N. van der Zwan-Kralt, R. Willemze and J.N. Bouwes Bavinck

Department of Dermatology, Leiden University Medical Centre, Albinusdreef 2, PO Box 9600, 2300 RC Leiden, the Netherlands

Summary

Correspondence

Y.G.L. de Graaf. E-mail: y.g.l.de_graaf@lumc.nl

Accepted for publication

10 August 2005

Key words

curettage and electrodesiccation, organ transplant recipients, recurrence rates, squamous cell carcinoma

Conflicts of interest

None declared.

Background Organ transplant recipients frequently develop multiple squamous cell carcinomas (SCCs). Surgical excision and Mohs micrographic surgery are frequently used treatments for these carcinomas; however, curettage and electrodes-iccation are a useful alternative in these patients.

Objectives To evaluate the efficacy of curettage and electrodesiccation for the treatment of appropriately selected low-risk SCCs in organ transplant recipients at different sites.

Methods Between April 1989 and December 2004, 211 SCCs in 48 organ transplant recipients were treated by curettage and electrodesiccation. Only histologically confirmed SCCs were considered in this study. The charts of these patients were retrospectively reviewed and checked for the rate of residual or recurrent SCCs. The occurrence of residual or recurrent SCCs at different locations after treatment of SCCs with curettage and electrodesiccation was estimated with Kaplan–Meier survival analysis.

Results The mean follow-up time after curettage and electrodesiccation of the individual SCCs was 50 months (median 41; range 3–186). In total, 13 residual or recurrent SCCs were observed in 10 patients. The overall rate of residual or recurrent SCCs was 6%, with 7% for SCCs on the dorsum of the hands or fingers, 11% for SCCs on the head and neck, 0% for the forearms, and 5% for the remaining nonsun-exposed areas (shoulder, legs). No major clinical or cosmetic adverse events were registered after treatment.

Conclusions In organ transplant recipients with many SCCs curettage and electrodesiccation can be a safe therapy for appropriately selected low-risk SCCs, with an acceptable cure rate.

Organ transplant recipients are at an increased risk of developing nonmelanoma skin cancer, of which cutaneous squamous cell carcinomas (SCCs) are the most prevalent tumours.^{1,2} In these immunocompromised patients SCCs appear to be more aggressive than SCCs in immunocompetent patients, and multiple tumours frequently develop in short periods of time.³

SCCs in organ transplant recipients are often treated by surgical excision with histological examination.⁴ Moh's micrographic surgery can be performed in high-risk tumours.¹ Recently, two international groups of mainly dermatologists published guidelines about treatment and prevention of SCC in the transplant population and recommended that these lesions should be managed by destructive or excisional modalities.^{5,6}

Curettage and electrodesiccation is a treatment option in selected tumours, for example nonulcerated SCCs with well-

defined margins, smaller than 2 cm and localized on low-risk locations such as the trunk and extremities.⁶ Successful outcome is associated with the physician's experience.^{6,7}

Large case series in immunocompetent patients are available that report on the efficacy of curettage and electrodesiccation for the treatment of skin cancer.⁸ However, there are few data published on the specific outcome after curettage and electrodesiccation of different clinical types of SCCs and larger tumours.⁹ A study in which 981 SCCs were treated with curettage and electrodesiccation reported recurrence rates of $1\cdot3-3\cdot7\%$.¹⁰ Three larger series of 947, 894 and 104 cases, respectively, of both SCCs and basal cell carcinomas that were treated with curettage and electrodesiccation, reported excellent 5-year cure rates ranging from 96% to 100%.^{11–13}

As far as we know there are no studies examining curettage and electrodesiccation as a treatment of SCCs in organ transplant recipients. In the literature, there is only one case of multiple SCCs in an organ transplant recipient that were successfully treated by curettage.¹⁴ Although not substantiated by the literature, curettage and electrodesiccation have been widely used in organ transplant recipients, usually for superficial or early skin cancers.¹⁵ The purpose of this retrospective follow-up study was to evaluate the cure and recurrence rate of SCCs after treatment with curettage and electrodesiccation in organ transplant recipients and to compare the cure and recurrence rates at different skin locations.

Patients and methods

Since 1966, roughly 2000 patients received a kidney or kidney–pancreas transplant at the Leiden University Medical Centre (LUMC), Leiden, the Netherlands. Approximately 200 organ transplant recipients with skin problems were regularly seen at the Department of Dermatology at the LUMC. Liver and heart transplant recipients were also seen occasionally; these had received their organs at other centres. Initially, all SCCs were treated by surgical excision. In April 1989 we started to treat some SCCs with curettage and electrodesiccation. Gradually this treatment became a more common scenario in our clinic in appropriately selected low-risk SCCs.

Low-risk SCCs were selected based on clinical grounds only: we used curettage and electrodesiccation in tumours which were less than 2 cm in size, which had developed within less than 3 months, and which did not clinically appear to infiltrate into the deeper tissues. After local anaesthesia most of the tumour mass was removed with a scalpel and this material was always sent for histological examination. The remaining tumour mass was removed with a curette and the bottom and the margins of the tumour were subjected to electrodesiccation. The procedure of curettage and coagulation was repeated several times.

Only SCCs confirmed by histological examination and treated with curettage and electrodesiccation between April 1989 and December 2004 were included in the study. All patients in the study were routinely seen in the outpatient dermatology clinic of the LUMC at 3-monthly intervals or more frequently when indicated. Their medical records were reviewed. Data were collected on localization of the tumour; possible residual or recurrent SCC, and time period to this occurrence; length of follow-up; and complications of treatment.

Using curettage and electrodesiccation, the difference between noncured or residual tumours and *de novo* or recurrent tumours cannot be made clearly based on clinical or histological grounds. An SCC was considered not to be cured or to recur if a histologically confirmed SCC occurred at the same location as the primary tumour during the follow-up period. Most of the time the locations of new SCCs were clearly different compared with the initial SCC. If there was any doubt about the location, the new SCC was considered to be a residual or recurrent SCC. Sun-exposed skin was defined as skin on the dorsum of the hands and fingers, forearms, and the head and neck region; nonsun-exposed skin was defined as all remaining locations.

Data were analysed using SPSS version 12.0 for Windows (SPSS, Chicago, IL, U.S.A). The rate of residual or recurrent SCCs after treatment of SCC by curettage and electrodesiccation was calculated with a Kaplan–Meier survival analysis. The date of the curettage and electrodesiccation was used as the opening date for this calculation. As the closing dates we used the date of the histological diagnosis of the residual or recurrent SCC, the patient's death, or the last visit of the patient in our outpatient clinic. Survival functions for the different locations of the SCCs were compared using the log rank test.

Results

Altogether, 211 SCCs occurring in 48 organ transplant recipients were treated with curettage and electrodesiccation in our Medical Centre between 1989 and the end of 2004. All the lesions with a few exceptions were treated by one experienced dermatologist (J.N.B.B.) or under the direct supervision of this dermatologist.

The main characteristics of the patients with and without residual or recurrent SCCs are depicted in Table 1. The 48 patients were followed for a mean \pm SD period of 73 \pm 48 months (median 70, range 3–186) as the first SCC was treated with curettage and electrodesiccation.

The mean follow-up period of the individual 211 SCCs was 50 months (median 41, range 3–186). Residual or recurrent SCCs were observed in 10 of 48 patients. Patients with residual or recurrent SCCs tended to be younger, were more often female and had more SCCs in their medical history (Table 1). These differences did not reach statistical significance with the exception of the total number of SCCs treated with curettage and electrodesiccation, which was much higher in the patients with recurrent SCCs (Table 1).

Most of the SCCs treated with curettage and electrodesiccation were located on sun-exposed skin (n = 129, 61%), and predominantly on the dorsum of the hands and fingers (n = 81, 38%). The tumour characteristics for the different skin localizations are displayed in Table 2.

Residual or recurrent SCCs were clinically and histologically documented in 13 (6%) of the 211 treated SCCs occurring in 10 of the 48 patients (Table 3). The mean \pm SD time to recurrence was 10 \pm 10 weeks (median 8, range 2–40). All residual or recurrent skin cancers were treated by surgical excision and we did not observe any additional recurrences after this excision during the follow-up period of between 8 and 128 months (Table 3). Four patients died from non-SCC-related causes during the follow-up period. One of these patients had a residual or recurrent SCC on the left temple, 11 years before his death.

The rate of residual or recurrent SCC after treatment with curettage and electrodesiccation of the SCCs for the different locations is shown in Figure 1. The differences were not statistically significant (P = 0.44). The majority of the residual or recurrent SCCs, 11 of 13, developed within 12 weeks of

Table 1 Characteristics of the patients

	Without residual or recurrent SCCs	With residual or recurrent SCCs
Total number of patients	38	10
Sex: M/F	22/16	3/7
Type of transplantation		
Kidney	33	9
Kidney-pancreas	3	1
Liver	1	0
Heart	1	0
Time period after transplantation at last outpatient	24 ± 8 (7-38)	$27 \pm 7 (12 - 37)$
visit (years), mean ± SD (range)		
Age at last outpatient visit (years), mean \pm SD (range)	62 ± 9 (43-79)	56 ± 10 (44-72)
Age at time of first SCC (years), mean \pm SD (range)	$52 \pm 10 (36-77)$	47 ± 8 (37–62)
No. of SCCs per patient (total), mean \pm SD (range)	8 ± 8 (1-38)	$18 \pm 11 \ (6-39)$
No. of SCCs per patient treated with curettage and	$3 \pm 3 (1-18)$	$10 \pm 7 (3-19)$
electrodesiccation, mean \pm SD (range) ^a		
Follow-up time since first SCC treated with curettage	70 ± 50 (3-186)	84 ± 39 (26-145)
and electrodesiccation (months), mean \pm SD (range)		. ,

SCC, squamous cell carcinoma. ^aThe differences between the two groups were not statistically significant with the exception of number of SCCs per patient treated with curettage and electrodesiccation (P = 0.01).

Table 2 Tumour characteristics

No. of primary SCCs (%)	No. of residual or recurrent SCCs	Time to residual or recurrent SCCs (weeks)
81 (38%)	6 (7%)	2, 6, 8, 9, 16, 40
28 (13%)	3 (11%)	5, 8, 8
20 (10%)	0	
82 (39%)	4 (5%)	3, 6, 10, 11
211 (100%)	13 (6%)	
	No. of primary SCCs (%) 81 (38%) 28 (13%) 20 (10%) 82 (39%) 211 (100%)	No. of primary No. of residual or recurrent SCCs 81 (38%) 6 (7%) 28 (13%) 3 (11%) 20 (10%) 0 82 (39%) 4 (5%) 211 (100%) 13 (6%)

SCC, squamous cell carcinoma.

Table 3 Characteristics of patients with residual or recurrent SCCs after curettage and electrodesiccation

Patient no.	Sex	Age (years)	Localization of SCC and histological type	Time to residual or recurrent SCCs (weeks)	FU after excision residual or recurrent SCC (months)
1	М	42	Finger web (I-II) right hand ^a	2	25
2	М	55	Dorsum left hand ^a	8	33
		57	Left shoulder ^a	3	8
3 ^c	М	50	Left temple ^b	5	128
4	F	45	Dorsum left hand ^a	6	85
5	F	49	Ventral part right upper leg ^a	6	26
		50	Right shoulder ^a	11	16
6	F	55	Frontal part of the scalp ^a	8	82
		55	Anterior part of the scalp ^b	8	82
7	F	39	Dorsum left hand ^a	9	53
8	F	66	Right lower leg ^a	10	81
9	F	67	Third finger right hand ^a	16	65
10	F	39	Third finger right hand ^a	40	83

SCC, squamous cell carcinoma; FU, follow-up. ^aWell-differentiated SCC; ^bpoorly differentiated SCC; ^cpatient died at age 61 years from other cause.


Fig 1. Kaplan–Meier survival analyses showing residual or recurrent SCCs on different locations of the body. The censored values are indicated.

follow-up; two developed within 40 weeks of follow-up. After this time period no additional residual or recurrent SCCs were observed (Fig. 1). No major clinical or cosmetic adverse events were registered after treatment.

Discussion

Our data show that curettage and electrodesiccation performed by a person with experience in this procedure is an effective treatment for SCCs in organ transplant recipients with multiple appropriately selected low-risk SCCs, with a low rate of residual or recurrent SCC of 6%. This rate is slightly higher than the rate of $1\cdot3-3\cdot7\%$ in earlier case series with immunocompetent patients.¹⁰⁻¹³

Although we treated a limited number of SCCs in the head and neck region without encountering significant problems, these high-risk SCCs are usually not recommended for treatment with curettage and electrodesiccation, because of the more aggressive nature of these SCCs and a higher risk of metastases. Therefore, this procedure should be discouraged for the head and neck region until the safety of curettage and electrodesiccation in these locations has been proved.

Curettage and electrodesiccation have many advantages. It should be emphasized, however, that curettage and electrodesiccation should only be performed by somebody with experience in this procedure. The cosmetic result is generally good or excellent and it is a relatively easy procedure. Clinical diagnosis, biopsy and definitive treatment can be completed in one visit. Therefore, it is possible to treat more lesions at the same time. Furthermore, no sutures have to be removed afterwards. All this makes curettage and electrodesiccation convenient for the patient who usually has more than one SCC.

SCCs on the dorsum of the hands and fingers may require reconstructive surgery or the use of a skin graft. This may necessitate a short stay in the hospital and sometimes complete anaesthesia. Most of the appropriately selected low-risk SCCs on the dorsum of the hands and fingers can be treated effectively with curettage and electrodesiccation, which is a much simpler procedure. In case of a residual or recurrent SCC reconstructive surgery is still a good option.

The main disadvantage of curettage and electrodesiccation is the lack of histopathological evaluation of the tumour margins. If a patient is examined regularly, this should not form a major problem. Other disadvantages of curettage and electrodesiccation are slow healing, the possibility of impaired wound healing, the increased risk of superficial infections and the risk of hypopigmentation and more prominent scars as a result of the procedure.

Remarkably, nearly all recurrences were observed within the first 12 weeks after treatment. This suggests that most 'recurrences' can be regarded as residual tumour. We did not observe any residual or recurrent SCCs later than 10 months after treatment, suggesting that in the case of tumour cells remaining, regrowth will occur rapidly, usually within weeks or at the most within several months after the procedure. We can conclude that the most critical period for evaluation is the first year after treatment, but patients should continue to be monitored regularly.

This is the first study that reports on the efficacy of curettage and electrodesiccation for SCCs in organ transplant recipients. A substantially long follow-up time was completed. Prospective randomized controlled trials are the best way to compare two treatment modalities, but this retrospective noncomparative study also provides valuable information. A randomized controlled trial with sufficient power to distinguish between a rate of residual or recurrent SCC of 2% and 6% after surgical excision and curettage and electrodesiccation, respectively, would require 424 SCCs in each treatment arm (power calculation with $\alpha = 0.05$ and $\beta = 0.8$), which is practically impossible to perform in one centre. In addition, the question can be asked whether a rate of residual or recurrent SCC of 2% instead of 6% would be clinically relevant, if excision offers the patient a second chance for complete cure, anyway.

In conclusion, based on the low rate of residual or recurrent SCC and the absence of significant clinical or cosmetic adverse events, we recommend curettage and electrodesiccation for organ transplant recipients who develop multiple appropriately selected low-risk SCCs, provided that the procedure is performed by an experienced person.

References

- Euvrard S, Kanitakis J, Claudy A. Skin cancers after organ transplantation. N Engl J Med 2003; 348:1681–91.
- 2 London NJ, Farmery SM, Will EJ et al. Risk of neoplasia in renal transplant patients. Lancet 1995; **346**:403-6.

- 3 Berg D, Otley CC. Skin cancer in organ transplant recipients. epidemiology, pathogenesis, and management. J Am Acad Dermatol 2002; 47:1–17.
- 4 Jemec GB, Holm EA. Nonmelanoma skin cancer in organ transplant patients. Transplantation 2003; **75**:253–7.
- 5 Anonymous. European best practice guidelines for renal transplantation. Section IV. long-term management of the transplant recipient. IV.6.2. Cancer risk after renal transplantation. Skin cancers: prevention and treatment. Nephrol Dial Transplant 2002; 17 (Suppl. 4):31–6.
- 6 Stasko T, Brown MD, Carucci JA et al. Guidelines for the management of squamous cell carcinoma in organ transplant recipients. Dermatol Surg 2004; 30:642–50.
- 7 Sheridan AT, Dawber RP. Curettage, electrosurgery and skin cancer. Australas J Dermatol 2000; **41**:19-30.
- 8 Goldman G. The current status of curettage and electrodesiccation. Dermatol Clin 2002; 20:569–78.
- 9 Motley R, Kersey P, Lawrence C. Multiprofessional guidelines for the management of the patient with primary cutaneous squamous cell carcinoma. Br J Dermatol 2002; **146**:18–25.

- 10 Rowe DE, Carroll RJ, Day CL Jr. Prognostic factors for local recurrence, metastasis, and survival rates in squamous cell carcinoma of the skin, ear, and lip. Implications for treatment modality selection. J Am Acad Dermatol 1992; 26:976–90.
- 11 Freeman R, Knox J, Heaton C. The treatment of skin cancer. A statistical study of 1,341 skin tumors comparing results obtained with irradiation, surgery, and curettage followed by electrodesiccation. Cancer 1964; 17:535–8.
- 12 Knox J, Freeman R, Duncan WC, Heaton C. Treatment of skin cancer. South Med J 1967; 60:241–6.
- 13 Tromovitch T. Skin cancer; treatment by curettage and desiccation. Calif Med 1965; 103:107–8.
- 14 Reymann F. Treatment of multiple squamous cell carcinomas of the skin in an immunosuppressed patient. Dermatologica 1981; 162:304-6.
- 15 Clayton AS, Stasko T. Treatment of nonmelanoma skin cancer in organ transplant recipients: review of responses to a survey. J Am Acad Dermatol 2003; 49:413–16.

CHAPTER 8

Summary and general discussion

Introduction

Skin carcinomas develop at a high rate in organ-transplant recipients who are kept on immune suppressive drugs to prevent graft rejection. The present study dealt with a broad range of aspects of this elevated carcinoma risk, starting from the earliest oncogenic events to the ultimate therapy. Advancements on any of these aspects may be of significant benefit to the patient and his/her physician in the management of multiple and progressive skin carcinomas.

The studies presented in Chapter 2 - 4 focused on the early pathogenesis of skin cancer in organ-transplant recipients to gain a better understanding of the underlying mechanism(s) of the increased skin cancer risk in these patients. We specifically focused on the role of p53 and beta-PV in early skin carcinogenesis. The clinical studies in Chapter 5 - 7 investigated the management of skin cancer in organ-transplant recipients. Here we summarize our most important findings and place these findings in a broader perspective.

Early oncogenic events: p53

Mutations in the p53 tumour suppressor gene, often leading to overexpression of the protein, appear to constitute an almost obligatory step in the development of squamous-cell carcinomas by chronic sun (UV) exposure. P53 patches are defined as clusters of epidermal cells that overexpress the p53 protein, and appear to be early microscopic precursor lesions of actinic keratoses and skin cancers.¹ We, therefore, investigated the early occurrence of p53 patches occurring in organ-transplant recipients (Chapter 2). Additionally, we studied a possible immune reactivity against the resulting overexpressed p53 protein in the same category of patients (Chapter 3).

P53 patches and immunosuppression

In **Chapter 2** we showed that p53 patches were more frequently present adjacent to skin cancers of renal-transplant recipients than of immunocompetent patients, indicating that p53 patches are more frequently present in immunosuppressed patients.

As the majority of the organ-transplant recipients in our study were treated with the classic immunosuppressant azathioprine, we investigated the effect of this drug on experimental p53 patch induction by chronic UV exposure in hairless mice. In earlier experiments, azathioprine had been shown to accelerate UV carcinogenesis in the hairless mouse model.² Since p53 patches precede the development of actinic keratoses and skin cancer one would expect an increased number of p53 patches in UV-irradiated mice immunosuppressed with azathioprine. In our study, however, we did not find an increase in p53 patches after daily UV irradiation in mice immunosuppressed with azathioprine compared with control mice. Hence, unlike the skin carcinomas, the number of p53 patches did not appear to be accelerated by immunosuppression of the mice. This is consistent with the results of another study in which the mutant p53 foci developed and regressed equally fast in immunocompromized RAG-1 mice and in wild-type mice.³ These findings indicate that p53 patches are not subjected to immunosurveillance and elimination.

Azathioprine inhibits DNA Repair

We were able to demonstrate the inhibiting effect of azathioprine on DNA repair of UVinduced DNA damage in primary human keratinocytes, as measured by unscheduled DNA synthesis (UDS) (Chapter 2). In humans UDS is dominated by a repair of cyclobutane pyrimidine dimers (CPD), whereas in mice the carcinogenic CPD are hardly repaired and UDS is therefore dominated by repair of less abundant DNA lesions photoproducts).^{4,5} In humans, (6-4 an azathioprine-related decreased repair of CPD is likely to explain increased amounts of p53 patches in renal-transplant recipients. In mice, on the other hand, CPD repair is already poor and consequently azathioprine will not have any apparent impact on CPDdriven mutagenesis and p53 patch formation in mice. Thus, our results indicate that carcinoma risk in renal-transplant recipients, is, at least in part, related to an increase in p53 patches, which in turn is attributable to a direct adverse effect on DNA repair from azathioprine (and similar effects from cyclosporine and tacrolimus (Prograft®)⁶⁻⁸).

In contrast to the classic immunosuppressive agent azathioprine and the calcineurin inhibitors cyclosporine and tacrolimus, the novel immunosuppressive agents, mycophenolate mofetil (MMF, Cellcept®) and sirolimus (rapamycin, Rapamune®) and everolimus appear to impair tumour growth. However, the impact of these novel immunosuppressive agents on DNA repair and early stages of UV-induced carcinogenesis has not been evaluated. On forehand, mTOR (mammalian target of rapamycin) inhibitors such as sirolimus and everolimus, act through pathways (down-stream Akt), which are not known to affect DNA repair. Additionally, there are reports that sirolimus enhances apoptosis in p53 dysfunctional cells.⁹

We would therefore expect that sirolimus carries an importantly lower risk from local adverse effects in skin cells, when compared with the classical immunosuppressive agents. (i.e. lower mutation rate, fewer p53 patches and correspondingly fewer skin carcinomas) Moreover, the mTOR inhibitors inhibit angiogenesis and tumour growth, resulting in a further anticipated beneficial effect against tumour formation.¹⁰ This is currently studied in animal experimental studies running in our institute and in several multicentre clinical studies.

Azathioprine and p53 mutations in skin carcinomas

Early studies showed the typical UVBrelated p53 mutations in a majority of squamous-cell carcinomas^{11,12}, and "founding" p53 mutations were found to be commonly present throughout the tumour mass.^{13,14} However, a recent study differed in its results and found disjunctive regions with independent p53 mutations in tumour masses of squamous-cell carcinomas, i.e. no indication of a founder p53 mutation, and a predominance of UVA-related p53 mutations in the basal layers of the tumour.¹⁵ With a lowered DNA repair, owing to either azathioprine or cyclosporine, one would suspect an enhanced induction of UVB-related p53 mutations and ultimate carcinomas dominated by such founder mutations sprouted from p53 patches. With photosensitization of the DNA to UVA radiation by azathioprine, one would anticipate mutations in relation to potential thio-guanine sites, i.e. a shift away from UVB-like mutations. We therefore performed a small pilot study on 5 squamouscell carcinomas from organ-transplant recipients. Mutation analysis was performed for exons 4b to 8 of the p53 gene from 4 different regions in each tumour mass. The results are shown in Table 1 (unpublished data). Four out of 5 squamous-cell carcinomas (80%) showed at least one p53 mutation present in at least two regions of the tumour. Two tumours showed at least one p53 mutation consistently present in all 4 regions dissected from the tumour, i.e. the apparent presence of a founder mutation.

In these 5 squamous-cell carcinomas a total of 13 p53 mutations were found of which only 2 (15%) were UVB-specific (Table 1). Hence, the p53 mutation spectrum that we found suggests a deviation from what is normally encountered in squamous-cell carcinomas from immunocompetent patients (in general 60% C \rightarrow T transitions and 10% CC \rightarrow TT tandem mutations).^{11,12} Interestingly, 4 out of 5 patients that we studied used azathioprine. These data suggest that azathioprine might cause alterations in the p53 mutation spectrum. In an earlier report. McGregor et al.¹⁶ found a total of 12 mutations in 23 squamous-cell carcinomas from renal-transplant patients. The majority of mutations were UVB-specific, although no characteristic $CC \rightarrow TT$ mutations were present. Unfortunately, the medication (azathioprine or otherwise) was not taken into account, and, as in our study, the number of mutations was probably too small. Future studies are clearly needed to infer the impact of immunosuppressive drugs, most specifically azathioprine, on the p53 mutation

Table 1. Distril	oution of P53 mut	ations in	squamous	-cell carcinomas of organ-transpl	ant recipients.		
Patientno.	medication	exon	intron	mutation	amino acid	codon	no. of sites in tumour with mutation
	P+M	4 4	4 v	gtCtggg→gtGtggg caGAggc →caATggc ttgcCctga →ttgcActga acaGtac →acaAtac	Leu→Val Glu →Met	111 68 *	ω 1 2 ω
2	P+A						
ς.	P+A	4 %	4 9	caGAggc →caATggc tgCgtgt →tgTgtgt** ttgcCctga →ttgcActga tgcAact →tgcTact	Glu →Met Arg →Cys	68 273	- 7 7 7
4	P+A	4 v	4	caGAggc →caATggc caCcccc →caTcccc** ttgcCctga →ttgcActga	Glu →Met Pro →Ser	68 151	6 4 4
S	P+A	4	4	tttCCgtc →tttATgtc ttgcCctga →ttgcActga	Phe →Tyr Arg →Cys	109 110	5 m m
Abbreviations: I *mutation at spl **C →T transiti	P=prednisone; M=1 ice site ion	mycopher	iolate mofe	til; A=azathioprine			

spectrum and founder or later stage p53 mutations in skin carcinomas.

Immune reaction to p53

Overexpression of mutant p53 may lead to immune reactions against p53, as was found in patients with colon cancers. A p53positive serology in these patients spelled a poor prognosis.¹⁷ We posed the question whether abundant overexpression of p53 in skin lesions of renal-transplant recipients would lead to p53 positive serology and whether the serology would be an early indication of malignant progression.

In **Chapter 3** we showed that p53-specific serum antibodies were not associated with a history of skin carcinoma in renal-transplant recipients and immunocompetent individuals. The increased number of p53 patches and actinic keratoses in renaltransplant recipients clearly did not induce a humoral response against p53, in contrast to colon carcinomas with obviously larger tumour masses, more necrosis and inflammation.

A recent study showed increased p53specific CD8+ T-cell responses in immunocompetent individuals with cutaneous squamous-cell carcinoma compared with controls.¹⁸ Although the cellular immunity is suppressed in organ-transplant recipients, it would be interesting to compare p53-specific T-cell responses in organ-transplant recipients with and without skin cancer.

Early oncogenic events: beta-PV and apoptosis

Besides the epidemiologic evidence for a role of beta-PV in skin carcinogenesis, recent experimental studies provided evidence of oncogenic properties of beta-PV types: E6 proteins of some beta-PV types block the pro-apoptotic protein Bak, which impairs apoptosis in response to UV exposure.¹⁹ In our study, described in **Chapter 4**, we could find no evidence in-vivo that beta-PV infection lowered the apoptotic response of UV-irradiated epidermis. However, the quantity of beta-papillomavirus DNA in our skin

samples was very low and often below detection limit of quantitative PCR. (The loads were in the range of those reported in another study.²⁰) The keratinocytes containing beta-PV DNA were sparse among an abundance of non-infected cells. Furthermore, probably not all beta-PV types carry a high carcinoma risk and exert a similar effect on UV-induced apoptosis. It has recently been reported that the E6 and E7 proteins of the beta-PV type 38 display in-vitro transforming properties²¹ by selectively activating transcription of $\Delta Np73$, an isoform of the p53-related protein p73. As a result, transcriptional functions of p53 involved in growth suppression and apoptosis are impaired.22

As an important side product of our study, we found clear evidence of a decrease in UV-induced apoptotic response with increasing age. Such a decline in clearance of damaged cells may enhance the cancer risk but also extend longevity of the tissue.²³

We chose unexposed buttock skin to ascertain how immunosuppression would affect basic levels of beta-PV infection, i.e., in absence of regular solar UV exposure. We did find beta-papillomaviruses to be present at higher densities in the organ-transplant recipients than in controls, but the difference was not statistically significant and the density in organ-transplant recipients was still extremely low. Nor did we measure any effect on beta-PV infection from a single UV exposure. Possibly, a single UV exposure is simply not sufficient, and multiple exposures might have led to a more pronounced effect. In our study we found a tendency for a higher prevalence of beta-PV in forearm skin compared to buttock skin, which is consistent with an earlier study that reported a higher prevalence of beta-PV in chronically sun-exposed lesional skin compared to nonexposed skin.²⁴ This is in line with our supposition that chronic sun-exposure may induce beta-PV replication. It has been shown that at least beta-PV types 5, 8 and 77 have UV responsive elements.^{25,26}

To ascertain the effect of beta-PV infection on UV-induced apoptosis in-vivo, studies need to focus specifically on beta-PVcarrying cells, preferably after repeated UV exposures. It is recommended to consider also the influence of the different immunosuppressive agents on UV-induced apoptosis in future studies.

Early oncogenic events: beta-PV and DNA repair

Beta-PV could also contribute to tumour development by interfering with the DNA damage repair process. It recently has been reported that thymine dimers repair after UV irradiation was impaired in cells expressing the E6 protein of beta-PV type 5.²⁷

In addition, it has been shown that the E6 protein of beta-PV types 1 and 8 binds and inhibits the human XRCC1 protein, which is required for the repair of single-strand breaks and genetic stability.²⁸

Possibly, the repair of DNA damage is impaired because of the interference of certain beta-PV types. However, the exact role of beta-PV in the etiology of skin carcinoma development is still uncertain. Additional studies are required to understand the role of beta-PV and the possible interaction with other factors in skin cancer development.

Management of skin cancer in organtransplant recipients

Two international groups of mainly dermatologists published guidelines on prevention and treatment of squamous-cell carcinomas in the transplant population.^{29,30} We contributed to improved management of skin cancer by studying both retinoids and photodynamic therapy as preventive modalities, and the use of curettage and electrodessication as a treatment of squamous-cell carcinomas in organ-transplant recipients. Tables 2 and 3 present the Leiden clinical guidelines for prevention and treatment of skin cancer in organ-transplant recipients adapted from two international guidelines.^{29,30}

Prevention

To prevent the development of skin carcinomas, cryosurgery, topical 5-fluorouracil and curettage with electrodessication are recommended to treat premalignant lesions.³⁰ Concerning the use of imiquimod, photodynamic therapy and retinoids in the prevention of skin cancer, there is no consensus in the guidelines.³⁰

In Chapter 5 we summarized the studies on the use of retinoids in the chemoprevention of skin cancer in organ-transplant recipients. We can conclude that low-dose retinoids, in particular acitretin, are effective in inhibiting the development of skin cancer. Our conclusion was supported by a recent study in which organ-transplant recipients received retinoid treatment for up to 16 years.³¹ Longterm use of systemic retinoids appears to be required. However, this may be limited by the occurrence of side effects in individual patients and a theoretical risk of graft rejection because of immunoactivation. Systemic retinoids, should therefore, in our opinion, be limited to patients with numerous hyperkeratotic skin lesions and/or at least one squamous-cell carcinoma in the medical history. The guidelines²⁹ are more strict and do only recommend systemic retinoids if multiple skin cancers have developed.

Although topical retinoids seem effective in the treatment of actinic keratoses and warts in organ-transplant recipients, the effect is clearly less than systemic retinoids. Following the guidelines, topical retinoids may be beneficial in patients with significant photodamage to normalize actinically damaged skin.²⁹

Chapter 6 shows that photodynamic therapy with aminolevulinic acid and violet light (400-450 nm) does not appear to prevent the development of new squamous-cell carcinomas in organ-transplant recipients. We observed a small, but non-significant, effect on the keratotic lesions. An explanation for this is the location of treatment in our study: an inferior response to photodynamic therapy of thick keratotic skin lesions on the hands compared to the thinner lesions on the Table 2. Guidelines for prevention of skin cancer in organ-transplant recipients (adapted from the Expert Group on Renal Transplantation²⁹ and Stasko et al³⁰).

Primary prevention	•	patient education on skin cancer risk and self examination patient education on sun protection: avoidance of sun exposure prohibition of tanning beds use of protective clothing use of an effective UVB/UVA sunscreen (SPF>15) for sunexposed skin on a daily basis
Secondary preventi- on	•	early referral to a dermatologist in case of premalignant lesions treatment of premalignant lesions by: cryotherapy curettage & electrodessication 5-fluorouracil or possibly by imiquimod, PDT*, topical retinoids* use of systemic retinoids, if well tolerated*
	•	use of systemic retinoids, if well tolerated* reduction of immunosuppression, if possible, by transplant physician

* Details on efficacy and indications are given in text

Table 3. Guidelines for treatment of skin cancer in organ-transplant recipients (adapted from the Expert Group on Renal Transplantation²⁹ and Stasko et al³⁰).

Diagnosis	Treatment of 1 st choice	Additional modalities
Basal-cell carcinoma	surgical excision curettage & electrodessication	
Squamous-cell carcinoma	surgical excision curettage & electrodessication*	resurfacing dorsum of hand
Recurrent squamous-cell	surgical excision Mobs' surgery	reduction of immunosuppression
caremonia	Wons surgery	systemic retinoids
Metastatic squamous-cell	surgical excision	radiotherapy
carcinoma	Mohs' surgery lymphadenectomy	reduction of immunosuppression systemic retinoids

* Details on efficacy and indications are given in text

face has been demonstrated in immunocompetent patients.³²

Moreover, our treatment protocol may not have been optimal, there was a 6-month period between the two treatments, which is rather long. In a recent study, photodynamic therapy with methyl aminolevulinic acid was shown to be more effective than 5fluorouracil in treating premalignant skin lesions in 8 organ-transplant recipients.³³ However, the study design is completely different from ours and therefore the results are not comparable. Although the use of red light (570-750 nm), methyl aminolevulinic acid and pretreatment of thick hyperkeratotic lesions could have resulted in a better therapeutic outcome (by an optimal penetration of light and photosensitizer) in our study, it has to be considered that photodynamic therapy might be less effective in organ-transplant recipients compared with immunocompetent patients.^{34,35} A possible explanation for this is the growing evidence that both innate and adaptive immune responses play a role in the outcome of photodynamic therapy.³⁶ In our opinion, photodynamic therapy with aminolevulinic acid and violet light is not a promising preventive therapy in organtransplant recipients. Probably repeated treatments for large areas are necessary to prevent skin cancer in these patients, which is obviously expensive and painful. Obviously, future studies on the use of photodynamic therapy in immunosuppressed patients are warranted.

Treatment

In the above mentioned guidelines, it is recommended that all skin cancers in organtransplant recipients should be completely removed with destructive or excisional modalities (Table 3).^{29,30} Non-ulcerated squamous-cell carcinomas with well-defined margins that are smaller than 2 cm and localized on low-risk locations such as trunk and extremities could form an indication for curettage and electrodessication.³⁰ However, until now there is not much evidence for its efficacy as a treatment of skin carcinomas in organ-transplant recipients. In **Chapter 7** we demonstrated that curettage and electrodessication is an effective treatment for squamous-cell carcinomas in organ-transplant recipients with multiple tumours with a low recurrence rate of 6%. The cosmetic result was good and a substantially long follow-up period was completed (50 months per tumour). Nearly all recurrences in our study were observed within the first 12 weeks after treatment. This suggests that most recurrences can be regarded as residue tumours and that in case of presence of residual tumour cells, outgrowth will occur rapidly.

A disadvantage of this treatment is the lack of histopathological evaluation of the tumour margins, and therefore patients should be monitored regularly after treatment with curettage and electrodessication. Recurrent tumours should be treated with surgical excision. Although we did not encounter significant problems in treating squamous-cell carcinomas in the head and neck region, we do not recommend curettage and electrodessication, because of a more aggressive nature of these squamous-cell carcinomas and a higher risk of metastases.³⁰ In conclusion, we can recommend curettage and electrodessication as a treatment of squamous-cell carcinomas at trunk or extremities in organ-transplant recipients.

Follow-up strategy

Recommended follow-up intervals for dermatologic examinations should be communicated to both patients and transplant physicians.³⁷ These intervals have to be determined by the individual's risk for skin cancer development.³⁰ In general, it is recommended to examine patients without skin cancer or actinic keratoses every year, patients with actinic keratoses and/or one skin cancer every 6 months and patients with multiple or high-risk skin cancers should be examined every 2 to 4 months. Patients without skin cancer and actinic keratoses can be examined annually by their transplant physician until lesions arise.^{30,37}

Conclusion and perspectives

In the first part of this thesis it was demonstrated that the number of p53 patches is increased in renal-transplant recipients compared to immunocompetent individuals. However, an increased number of p53 patches was not found in immunosuppressed mice treated with azathioprine. Unlike the skin carcinomas, the number of p53 patches did not appear to be accelerated by immunosuppression of the mice. It was also demonstrated that azathioprine inhibited the DNA repair of UV-induced DNA damage in primary human keratinocytes. Thus, the carcinoma risk in renal-transplant recipients appears to be related -at least in part- to an increase in p53 patches. And the presence of these p53 patches may be attributed to a direct adverse effect of immunosuppressive agents as azathioprine on DNA repair. Addihypothesis tionally, the that betapapillomaviruses decrease apoptosis could not be confirmed in our in-vivo study. We did find a negative correlation between age and UV-induced apoptosis.

In the second part of this thesis it was shown that systemic retinoids are effective in inhibiting skin cancer development in organtransplant recipients, in contrast to photodynamic therapy, which only had a small (nonsignificant) effect on the keratotic lesions. Furthermore, it was demonstrated that curettage and electrodessication is an effective treatment with a low recurrence rate for squamous-cell carcinomas in organtransplant recipients.

Considering the harmful effects of the classic immunosuppressives azathioprine and cyclosporine and the growing evidence of antineoplastic properties of the novel immunosuppressive agents mycophenolate mofetil and sirolimus, future studies should be aimed at the effect of these immunosuppressive agents on early UV-induced skin carcinogenesis. Besides p53 patches, these studies should focus on cyclobutane pyrimidine dimers and other photolesions (e.g. caused by phototoxicity) These photolesions are expected to be more persistent in skin of organtransplant recipients compared to immunocompetent patients.

Future studies should establish the exact role of beta-PV in the early pathogenesis of skin carcinomas in organ-transplant recipients. The impact of beta-PV infection on UVinduced apoptosis in-vivo should be studied specifically in beta-PV carrying cells. Additionally, the role and the possible relation between beta-PV and p53 in tumour development needs to be established.

These studies may finally lead to possible interventions (to prevent skin carcinomas) directed against DNA-damaged cells by administration of repair enzymes. It may also lead to the development of an immunosuppressive regimen with minimum skin cancer risk and improved therapeutic modalities for skin carcinomas in organ-transplant recipients.

References

- Rebel H, Kram N, Westerman A, Banus S, van Kranen HJ, de Gruijl FR. Relationship between UV-induced mutant p53 patches and skin tumours, analysed by mutation spectra and by induction kinetics in various DNA-repairdeficient mice. Carcinogenesis 2005;26:2123-2130.
- Kelly GE, Meikle W, Sheil AG. Effects of immunosuppressive therapy on the induction of skin tumors by ultraviolet irradiation in hairless mice. Transplantation 1987;44:429-434.
- Remenyik E, Wikonkal NM, Zhang W, Paliwal V, Brash DE. Antigen-specific immunity does not mediate acute regression of UVB-induced p53-mutant clones. Oncogene 2003;22:6369-6376.
- Ruven HJ, Berg RJ, Seelen CM, Dekkers JA, Lohman PH, Mullenders LH, van Zeeland AA. Ultraviolet-induced cyclobutane pyrimidine dimers are selectively removed from transcriptionally active genes in the epidermis of the hairless mouse. Cancer Res 1993;53:1642-1645.
- Tang JY, Hwang BJ, Ford JM, Hanawalt PC, Chu G. Xeroderma pigmentosum p48 gene enhances global genomic repair and suppresses UV-induced mutagenesis. Mol Cell 2000;5:737-744.
- Kelly GE, Meikle W, Sheil AG. Scheduled and unscheduled DNA synthesis in epidermal cells of hairless mice treated with immunosuppressive drugs and UVB-UVA irradiation. Br J Dermatol 1987;117:429-440.
- Yarosh DB, Pena AV, Nay SL, Canning MT, Brown DA. Calcineurin inhibitors decrease DNA repair and apoptosis in human keratinocytes following ultraviolet B irradiation. J Invest Dermatol 2005;125:1020-1025.
- Herman M, Weinstein T, Korzets A, Chagnac A, Ori Y, Zevin D, Malachi T, Gafter U. Effect of cyclosporin A on DNA repair and cancer incidence in kidney transplant recipients. J Lab Clin Med 2001;137:14-20.
- Huang S, Liu LN, Hosoi H, Dilling MB, Shikata T, Houghton PJ. p53/p21(CIP1) cooperate in enforcing rapamycininduced G(1) arrest and determine the cellular response to rapamycin. Cancer Res 2001;61:3373-3381.

- Koehl GE, Andrassy J, Guba M, Richter S, Kroemer A, Scherer MN, Steinbauer M, Graeb C, Schlitt HJ, Jauch KW, Geissler EK. Rapamycin protects allografts from rejection while simultaneously attacking tumors in immunosuppressed mice. Transplantation 2004;77:1319-1326.
- Brash DE, Rudolph JA, Simon JA, Lin A, McKenna GJ, Baden HP, Halperin AJ, Ponten J. A role for sunlight in skin cancer: UV-induced p53 mutations in squamous cell carcinoma. Proc Natl Acad Sci U S A 1991;88:10124-10128.
- 12. Giglia-Mari G, Sarasin A. TP53 mutations in human skin cancers. Hum Mutat 2003;21:217-228.
- Ren ZP, Hedrum A, Ponten F, Nister M, Ahmadian A, Lundeberg J, Uhlen M, Ponten J. Human epidermal cancer and accompanying precursors have identical p53 mutations different from p53 mutations in adjacent areas of clonally expanded non-neoplastic keratinocytes. Oncogene 1996;12:765-773.
- Backvall H, Asplund A, Gustafsson A, Sivertsson A, Lundeberg J, Ponten F. Genetic tumor archeology: microdissection and genetic heterogeneity in squamous and basal cell carcinoma. Mutat Res 2005;571:65-79.
- Agar NS, Halliday GM, Barnetson RS, Ananthaswamy HN, Wheeler M, Jones AM. The basal layer in human squamous tumors harbors more UVA than UVB fingerprint mutations: a role for UVA in human skin carcinogenesis. Proc Natl Acad Sci U S A 2004;101:4954-4959.
- McGregor JM, Berkhout RJ, Rozycka M, ter Schegget J, Bouwes Bavinck JN, Brooks L, Crook T. p53 mutations implicate sunlight in post-transplant skin cancer irrespective of human papillomavirus status. Oncogene 1997;15:1737-1740.
- 17. Soussi T. p53 Antibodies in the sera of patients with various types of cancer: a review. Cancer Res 2000;60:1777-1788.
- Black AP, Bailey A, Jones L, Turner RJ, Hollowood K, Ogg GS. p53-specific CD8+ T-cell responses in individuals with cutaneous squamous cell carcinoma. Br J Dermatol 2005;153:987-991.
- Jackson S, Harwood C, Thomas M, Banks L, Storey A. Role of Bak in UV-induced apoptosis in skin cancer and abrogation by HPV E6 proteins. Genes Dev 2000;14:3065-3073.
- Weissenborn SJ, Nindl I, Purdie K, Harwood C, Proby C, Breuer J, Majewski S, Pfister H, Wieland U. Human papillomavirus-DNA loads in actinic keratoses exceed those in non-melanoma skin cancers. J Invest Dermatol 2005;125:93-97.
- Caldeira S, Zehbe I, Accardi R, Malanchi I, Dong W, Giarre M, de Villiers EM, Filotico R, Boukamp P, Tommasino M. The E6 and E7 proteins of the cutaneous human papillomavirus type 38 display transforming properties. J Virol 2003;77:2195-2206.
- Accardi R, Dong W, Smet A, Cui R, Hautefeuille A, Gabet AS, Sylla BS, Gissmann L, Hainaut P, Tommasino M. Skin human papillomavirus type 38 alters p53 functions by accumulation of deltaNp73. EMBO Rep 2006;7:334-340.
- Campisi J. Cancer and ageing: rival demons? Nat Rev Cancer 2003;3:339-349.
- 24. de Jong-Tieben LM, Berkhout RJ, ter Schegget J, Vermeer BJ, de Fijter JW, Bruijn JA, Westendorp RG, Bouwes Bavinck JN. The prevalence of human papillomavirus DNA in benign keratotic skin lesions of renal transplant recipients with and without a history of skin cancer is equally high: a clinical study to assess risk factors for keratotic skin lesions and skin cancer. Transplantation 2000;69:44-49.
- Akgul B, Lemme W, Garcia-Escudero R, Storey A, Pfister HJ. UV-B irradiation stimulates the promoter activity of the high-risk, cutaneous human papillomavirus 5 and 8 in primary keratinocytes. Arch Virol 2005;150:145-151.
- Purdie KJ, Pennington J, Proby CM, Khalaf S, de Villiers EM, Leigh IM, Storey A. The promoter of a novel human papillomavirus (HPV77) associated with skin cancer displays UV responsiveness, which is mediated through a consensus p53 binding sequence. EMBO J 1999;18:5359-5369.

- Giampieri S, Storey A. Repair of UV-induced thymine dimers is compromised in cells expressing the E6 protein from human papillomaviruses types 5 and 18. Br J Cancer 2004;90:2203-2209.
- Iftner T, Elbel M, Schopp B, Hiller T, Loizou JI, Caldecott KW, Stubenrauch F. Interference of papillomavirus E6 protein with single-strand break repair by interaction with XRCC1. EMBO J 2002;21:4741-4748.
- European best practice guidelines for renal transplantation. Section IV: Long-term management of the transplant recipient. IV.6.2. Cancer risk after renal transplantation. Skin cancers: prevention and treatment. Nephrol Dial Transplant 2002;17 Suppl 4:31-36.
- Stasko T, Brown MD, Carucci JA, Euvrard S, Johnson TM, Sengelmann RD, Stockfleth E, Tope WD. Guidelines for the management of squamous cell carcinoma in organ transplant recipients. Dermatol Surg 2004;30:642-650.
- Harwood CA, Leedham-Green M, Leigh IM, Proby CM. Low-dose retinoids in the prevention of cutaneous squamous cell carcinomas in organ transplant recipients: a 16-year retrospective study. Arch Dermatol 2005;141:456-464.
- 32. Morton CA, Brown SB, Collins S, Ibbotson S, Jenkinson H, Kurwa H, Langmack K, McKenna K, Moseley H, Pearse AD, Stringer M, Taylor DK, Wong G, Rhodes LE. Guidelines for topical photodynamic therapy: report of a workshop of the British Photodermatology Group. Br J Dermatol 2002;146:552-567.
- 33. Perrett CM, McGregor JM, Warwick J, Karran P, Leigh IM, Proby CM, Harwood CA. Treatment of post-transplant premalignant skin disease: a randomized intrapatient comparative study of 5-fluorouracil cream and topical photodynamic therapy. Br J Dermatol 2007;156:320-328.
- 34. Dragieva G, Prinz BM, Hafner J, Dummer R, Burg G, Binswanger U, Kempf W. A randomized controlled clinical trial of topical photodynamic therapy with methyl aminolaevulinate in the treatment of actinic keratoses in transplant recipients. Br J Dermatol 2004;151:196-200.
- 35. Dragieva G, Hafner J, Dummer R, Schmid-Grendelmeier P, Roos M, Prinz BM, Burg G, Binswanger U, Kempf W. Topical photodynamic therapy in the treatment of actinic keratoses and Bowen's disease in transplant recipients. Transplantation 2004;77:115-121.
- Oseroff A. PDT as a cytotoxic agent and biological response modifier: Implications for cancer prevention and treatment in immunosuppressed and immunocompetent patients. J Invest Dermatol 2006;126:542-544.
- Berg D, Otley CC. Skin cancer in organ transplant recipients: Epidemiology, pathogenesis, and management. J Am Acad Dermatol 2002;47:1-17.

Nederlandse samenvatting

Inleiding

Orgaan transplantatiepatiënten hebben een verhoogd risico op niet-gepigmenteerde huidkankers, in het bijzonder het plaveiselcelcarcinoom, wat gepaard gaat met een aanzienlijke toename van morbiditeit en mortaliteit. De frequentie van huidkanker neemt toe met de tijd na transplantatie. In landen met matige zonexpositie zoals Nederland is het risico op huidkanker 40% 20 jaar na transplantatie. In zonniger oorden, zoals Australië, loopt dit risico op tot 70%. Het plaveiselcelcarcinoom komt ongeveer 65 tot 250 keer zo vaak voor bij transplantatiepatiënten als in de algemene populatie. Bovendien kennen plaveiselcelcarcinomen in deze patiëntenpopulatie een agressiever beloop dan gewoonlijk, vooral als ze in het hoofd-halsgebied zijn gelokaliseerd. Er is een verhoogd risico op lokaal recidief en metastasen. Wratachtige afwijkingen, premaligne keratosen, de ziekte van Bowen en keratoacanthomen gaan gepaard met een verhoogd risico op huidkanker na transplantatie. Andere typen huidkanker komen ook vaker dan normaal voor transplantatiepatiënten; het Kaposibii sarcoom ongeveer 84 tot 113 maal, het basaalcelcarcinoom 10 maal en het maligne melanoom 2 tot 8 maal.

Het eerste deel van dit proefschrift richt zich op vroeg oncogene gebeurtenissen in de huid welke een rol kunnen spelen bij het verhoogde risico op huidkanker bij transplantatiepatiënten. In het bijzonder werd de rol van het p53 tumorsuppressor gen en infectie met beta-papillomavirussen in de vroege carcinogenese van de huid bestudeerd. Het tweede deel van het proefschrift handelt over de preventie van huidkanker bij transplantatiepatiënten door middel van fotodynamische therapie en retinoïden en vervolgens over de behandeling van huidkanker door middel van curettage en coagulatie.

Hoofdstuk 1 geeft een overzicht van het klinische probleem van huidkanker bij orgaan transplantatiepatiënten. De belangrijkste risicofactoren voor huidkanker bij deze patiëntengroep worden besproken; ultraviolet (UV) straling, immuunsuppressieve (afweeronderdrukkende) therapie en infectie met betapapillomavirussen. Vervolgens wordt ingegaan op de preventieve en therapeutische modaliteiten welke in aanmerking komen bij deze patiëntengroep.

Pathogenese van huidkanker bij transplantatiepatiënten

Een van de belangrijkste risicofactoren voor niet-gepigmenteerde huidkanker bij transplantatiepatiënten wordt gevormd door UV straling. Dit wordt geïllustreerd door het feit dat bij transplantatiepatiënten premaligne afwijkingen, zoals actinische keratosen, en plaveiselcelcarcinomen vooral voorkomen op door de zon beschenen huid.

Mutaties in het p53 tumorsuppressor gen lijken een bijna onvermijdelijke vroege stap te zijn in de ontwikkeling van huidkanker door chronische UV blootstelling. P53 patches (P53 eilandjes), welke gedefinieerd worden als clusters van epidermale cellen die het p53 eiwit tot verhoogde expressie brengen, lijken vroege microscopische voorlopers van actinische keratosen en huidkanker te zijn. Hoofdstuk 2 beschrijft een studie waarin onderzocht werd of het aantal p53 patches in huid van transplantatiepatiënten verhoogd is in vergelijking met immuuncompetente patiënten (patiënten met een normale afweer). Wij vonden bij onderzoek van de huid rond uitgenomen carcinomen dat p53 patches inderdaad frequenter aanwezig waren in transplantatiepatiënten dan in immuuncompetente patiënten.

Een andere belangrijke factor in de ontwikkeling van huidkanker bij transplantatiepatienten is de immuunsuppressieve therapie. Bij langdurig gebruik ontstaat een situatie waarbij het (vroegtijdig) opruimen van kankercellen door het immuunsysteem is verminderd. Bovendien kunnen geneesmiddelen zoals azathioprine en ciclosporine, onafhankelijk van het immuunsuppressieve effect, een direct kankerverwekkend effect op huidcellen uitoefenen. Aangezien de meerderheid van de transplantatiepatiënten ten tijde van onze studie behandeld werd met het klassieke immuunsuppressieve middel azathioprine, werden vervolgens in **Hoofdstuk 2** twee mogelijke mechanismen bestudeerd waarmee azathioprine het aantal p53 patches zou kunnen verhogen: immuunsuppressie en een vermindering van DNA herstel.

Wij bestudeerden allereerst het effect van azathioprine op de experimentele inductie van p53 patches in de haarloze muis. Eerdere experimenten hebben laten zien dat azathioprine de huidkankervorming door UV-straling in de haarloze muis kan versnellen. In onze studie werd echter geen verhoogd aantal p53 patches gevonden in de dagelijks met UV bestraalde haarloze muis, waarbij de afweer onderdrukt werd met azathioprine, in vergelijking met controle muizen zonder azathioprine. Dit veronderstelt dat, in tegenstelling tot huidkanker, de vorming van p53 patches niet lijkt te worden versneld door immuunsuppressie in de haarloze muis. Dit is in overeenstemming met resultaten van een eerdere studie waarbij geen verschil werd gevonden in de snelheid van ontwikkeling van p53 patches tussen immuungecompromiteerde RAG-1 muizen en wild-type controle muizen. Een mogelijke verklaring voor deze bevinding zou kunnen zijn dat p53 patches niet onderhevig zijn aan herkenning of eliminatie door het immuunsysteem.

We toonden vervolgens aan dat azathioprine een remmende werking heeft op het herstel van schade geïnduceerd door UV straling in primaire humane keratinocyten, welke werd gemeten door middel van DNA herstel synthese (UDS) (Hoofdstuk 2). In de mens wordt UDS gedomineerd door herstel van cyclobutaan pyrimidine dimeren (CPD), in tegenstelling tot de muis, in welke dieren deze carcinogene fotoproducten veel slechter worden gerepareerd en voornamelijk de minder aanwezige DNA laesies, 6-4 fotoproducten, worden gerepareerd. In de mens zou een verlaagd herstel van CPD, gerelateerd aan azathioprine, het verhoogde aantal p53 patches bij transplantatiepatiënten kunnen verklaren. In de muis echter, is het herstel van de CPD al minimaal en heeft azathioprine waarschijnlijk nauwelijks effect.

Onze bevindingen doen veronderstellen dat het verhoogde huidkankerrisico bij transplantatiepatiënten tenminste ten dele toe te schrijven valt aan het verhoogde aantal p53 patches. Deze patches kunnen op hun beurt weer deels worden toegeschreven aan een direct nadelig effect op het DNA herstel door middelen als azathioprine. In tegenstelling tot de klassieke immuunsuppressieve middelen azathioprine en ciclosporine, hebben de nieuwere immuunsuppressiva mogelijk een remmende werking op tumorgroei. Het effect op vroeg oncogene veranderingen als p53 patches is echter nog niet onderzocht.

Overexpressie van mutant p53 kan een immuunreactie tegen p53 tot gevolg hebben, zoals bekend is bij patiënten met darmkanker. De aanwezigheid van antistoffen tegen p53 voorspelde een slechte prognose bij deze patiënten. Wij onderzochten of een verhoogde p53 expressie in huidlaesies ook zou kunnen resulteren in de aanwezigheid van deze antilichamen. Hoofdstuk 3 beschrijft de prevalentie van p53-specifieke antilichamen bij zowel niertransplantatiepatiënten als immuuncompetente personen met en zonder een plaveiselcelcarcinoom in de voorgeschiedenis. We vonden echter geen associatie tussen p53-specifieke antilichamen en plaveiselcelcarcinomen in niertransplantatiepatiënten en ook niet in immuuncompetente personen. Evenmin werd er een verschil gevonden in de hoeveelheid p53-specifieke antilichamen tussen de transplantatiepatiënten en immuuncompetente patiënten.

Een derde factor die mogelijk een rol speelt in de ontwikkeling van huidkanker bij transplantatiepatiënten is een infectie met betapapillomavirussen (beta-PV). Naast epidemiologische gegevens die een samenhang tonen tussen beta-PV infecties en plaveiselcelcarcinomen zijn er recentelijk ook dierexperimentele aanwijzingen voor een verband. Ook is onlangs in vitro aangetoond dat E6 eiwitten van enkele beta-PV typen het eiwit Bak kunnen remmen. Dit eiwit stimuleert apoptose, geprogrammeerde celdood. De gedachte is dat infectie met beta-PV de UVgeïnduceerde apoptose remt, waardoor de betreffende cel niet wordt opgeruimd en

waardoor genetische instabiliteit kan accumuleren in de celkernen. Hoofdstuk 4 presenteert onderzoek naar beta-PV in voorheen niet door de zon beschenen huid bij zowel transplantatiepatiënten als gezonde vrijwilligers. In het bijzonder werd het effect van beta-PV op UV-geïnduceerde apoptose onderzocht. Daarnaast werd het effect van UVB blootstelling op de aanwezigheid van beta-PV geëvalueerd. In onze studie werd echter geen effect gevonden van beta-PV op de hoeveelheid cellen die in apoptose gaan. Een waarschijnlijke verklaring hiervoor is dat de hoeveelheid virus, welke werd gevonden in onze samples, zeer laag was, meestal beneden de detectie limiet van de kwantitatieve PCR. Bovendien is het niet noodzakelijk dat elk beta-PV type een hoog-risico type is en een vergelijkbaar effect op UV-geïnduceerde apoptose uitoefent. De spaarzame hoeveelheid cellen geïnfecteerd met beta-PV werd hoogstwaarschijnlijk overschaduwd door de aanwezigheid van vele niet-geïnfecteerde cellen. Het is dan ook aan te bevelen het effect van beta-PV op apoptose op cellulair niveau te onderzoeken. In onze studie vonden we evenmin een effect van een eenmalige UV blootstelling op de aanwezigheid van beta-PV. Dit kan verklaard worden door het feit dat één UV blootstelling niet afdoende is om een eventueel effect op beta-PV replicatie te bewerkstelligen. We vonden namelijk een tendens tot een verhoogde prevalentie van beta-PV infectie in de langdurig aan zon blootgestelde huid van de onderarmen met wratachtige huidafwijkingen en actinische keratosen.

Een opmerkelijke bevinding van onze studie in zowel transplantatiepatiënten als gezonde vrijwilligers was een verlaging van UVgeïnduceerde apoptose met toenemende leeftijd.

Preventie en therapie van huidkanker bij transplantatiepatiënten

Het belangrijkste onderdeel van de preventie van huidkanker bij transplantatiepatiënten is patiënteneducatie betreffende de risico's van blootstelling aan UV-straling en zonbeschermingsadviezen. Patiënten dienen voorafgaand aan de transplantatie al geïnformeerd te worden over het verhoogde risico op huidkanker. Educatie met betrekking tot zelfonderzoek van de huid is belangrijk, opdat (pre)maligne afwijkingen vroeg worden herkend en de patiënt tijdig de arts bezoekt. Controle en behandeling van premaligne en maligne huidafwijkingen in een vroeg stadium zijn van groot belang. Transplantatiepatiënten met premaligne huidafwijkingen, zoals actinische keratosen, dienen dan ook naar een dermatoloog verwezen te worden. Voor chemopreventie van huidmaligniteiten zijn systemische retinoïden een optie. In Hoofdstuk 5 wordt een overzicht gegeven van de werkzaamheid van topicale en systemische retinoïden in de preventie van huidkanker bij transplantatiepatiënten. Hieruit kan geconcludeerd worden dat retinoïden in lage doseringen, vooral acitretine, redelijk effectief zijn in het remmen van de ontwikkeling van huidkanker. Hierbij dient opgemerkt te worden dat langdurig gebruik van systemische retinoïden nodig lijkt. Dit wordt echter vaak bemoeilijkt door de bijwerkingen. Wij zijn van mening dat systemische retinoïden dan ook voorbehouden moeten zijn aan patiënten met meerdere hyperkeratotische laesies en minimaal één plaveiselcelcarcinoom in de voorgeschiedenis.

Hoofdstuk 6 beschrijft een gerandomiseerde gecontroleerde studie met gepaarde waarnemingen bij 40 transplantatiepatiënten, bij wie het effect van fotodynamische therapie op het ontstaan van nieuwe plaveiselcelcarcinomen op de zonblootgestelde huid werd onderzocht. In deze studie werd echter geen preventief effect gevonden van fotodynamische therapie met violet licht. Er werd een klein (niet-significant) effect gevonden op het aantal hyperkeratotische laesies. Een mogelijke verklaring hiervoor is de locatie van behandeling; het is bekend dat de dikkere laesies op de armen minder goed reageren op fotodynamische therapie vergeleken met de doorgaans vlakkere laesies op de schedel. Ons behandelprotocol is mogelijk niet optimaal geweest, een tussenpoos van 6 maanden is mogelijk te lang. Bovendien zou voorbehandeling van de hyperkeratotische laesies met curettage, het gebruik van methylaminolevulinezuur en rood licht kunnen resulteren in een hogere penetratie en dientengevolge een beter effect. Meer onderzoek naar fotodynamische therapie bij immuungecompromiteerde patiënten lijkt dan ook nuttig.

Chirurgische excisie met postoperatieve snijrandbeoordeling is de eerste keus behandeling van huidkanker. Handrugtransplantaties, een procedure waarbij de gehele, actinisch zwaar beschadigde huid van de handrug wordt vervangen door weinig of niet beschadigde huid van bijvoorbeeld het bovenbeen of de nates, kunnen nuttig zijn bij patiënten bij wie veel plaveiselcelcarcinomen ontstaan in dit gebied. Deze behandeling heeft mede een preventief effect. Curettage en coagulatie vormt een behandelingsoptie bij geselecteerde tumoren op romp of extremiteiten. Echter over de werkzaamheid van deze behandeling bij transplantatiepatiënten was weinig bekend. In Hoofdstuk 7 wordt een serie plaveiselcelcarcinomen van transplantatiepatienten beschreven die behandeld werden door middel van curettage en coagulatie, met het doel om het risico op een recidief of hernieuwde uitgroei na niet volledige verwijdering na deze behandeling te bepalen en deze voor verschillende locaties te vergelijken. Uit deze studie blijkt dat curettage en coagulatie een effectieve behandeling is voor plaveiselcelcarcinomen bij transplantatiepatiënten waarbij 94% in een periode van 50 maanden was genezen. Het cosmetisch resultaat was goed en een substantiële follow-up periode werd bereikt.

Hoofdstuk 8 geeft een samenvatting van de resultaten beschreven in de voorgaande hoofdstukken en bediscussieert de bevindingen. Mogelijkheden voor toekomstig onderzoek worden ten slotte besproken, waarvan de belangrijkste is om te onderzoeken wat het effect is van de nieuwere immuunsuppressiva mycofenolaat mofetil en sirolimus op de vroege huidkanker ontwikkeling. Bovendien dienen de rol van beta-PV in de vroege pathogenese van huidkanker bij transplantatiepatiënten, en een eventuele relatie met p53, nader geëvalueerd te worden. **Curriculum vitae**

Leontien de Graaf werd op 16 juni 1976 te 's Gravenhage geboren. In 1994 behaalde zij het VWO diploma aan het Alfrink College te Zoetermeer. In hetzelfde jaar begon zij met de studie Biomedische Gezondheidswetenschappen aan de Katholieke Universiteit Nijmegen. Aan dezelfde universiteit startte zij in 1996 met de studie Geneeskunde. In deze periode verrichtte zij onderzoek naar roken en interacties met geneesmiddelen (Prof. dr. P. Smits, Afdeling Farmacologie & Toxicologie, Katholieke Universiteit Nijmegen). Voor de studie Biomedische Gezondheidswetenschappen deed zij haar afstudeeronderzoek naar het gebruik van teerpreparaten en het risico op huidkanker (dr. P. Pasker-de Jong en dr. P.G.M. van der Valk, Afdeling Klinische Epidemiologie en Biostatistiek en Afdeling Dermatologie, Academisch Ziekenhuis Nijmegen, St. Radboud). Het doctoraalexamen Biomedische Gezondheidswetenschappen, afstudeerrichting Epidemiologie, werd afgelegd op 26 april 2001. Na een keuze co-assistentschap in El Salvador behaalde zij het artsexamen op 5 oktober 2001.

In december 2001 werd zij aangesteld als AGIKO (assistent-geneeskundige in opleiding tot klinisch onderzoeker) op de Afdeling Dermatologie van het Leids Universitair Medisch Centrum en begon zij met promotieonderzoek onder begeleiding van dr. J.N. Bouwes Bavinck en dr. F.R. de Gruijl en werd de basis gelegd voor dit proefschrift. In januari 2005 startte zij in Leiden met de opleiding tot dermatoloog (opleider Prof. dr. R. Willemze).

Publicaties

Roken en interacties met geneesmiddelen. YGL de Graaf, DJTh. Wagener, P Smits. in: Tabaksgebruik; gevolgen en bestrijding, Red. K Knol, C Hilvering, DJTh. Wagener, MC Willemsen, Lemma 2005. (eerder gepubliceerd in het Geneesmiddelenbulletin; 2002; 36: 85-88).

Systemic and topical retinoids in the management of skin cancer in organ transplant recipients.

YGL de Graaf, S Euvrard, JN Bouwes Bavinck. Dermatologic Surgery 2004; 30: 656-661.

Huidkanker en andere huidaandoeningen bij patiënten na transplantatie van een solide orgaan.

YGL de Graaf, JW de Fijter, AN Posma, MCW Feltkamp, FH Claas, JN Bouwes Bavinck.

Nederlands Tijdschrijft voor Geneeskunde 2005; 149: 511-517.

P53-specific serum antibodies are not associated with a history of skin carcinoma in renal transplant recipients and immunocompetent individuals. YGL de Graaf, D Schiefer, A Redeker, BJ Vermeer, JN Bouwes Bavinck, R Willemze, FR de Gruijl, SH van der Burg. Journal of Dermatological Science 2005; 38: 228-230.

The occurrence of residual or recurrent squamous cell carcinomas in organ transplant recipients after curettage and electrodesiccation.

YGL de Graaf, VR Basdew, N van der Zwan-Kralt, R Willemze, JN Bouwes Bavinck.

British Journal of Dermatology 2006; 154: 493-497.

(tevens gepubliceerd in het Nederlands Tijdschrift voor Dermatologie en Venereologie 2006; 16: 402-406.)

Photodynamic therapy does not prevent cutaneous squamous-cell carcinoma in organtransplant recipients: results of a randomized-controlled trial. YGL de Graaf, C Kennedy, R Wolterbeek, AFS Collen, R Willemze, JN Bouwes Bavinck.

Journal of Investigative Dermatology 2006; 126: 569-574.

More epidermal p53 patches adjacent to skin carcinomas in renal-transplant recipients than in immunocompetent patients: the role of azathioprine.

YGL de Graaf, HG Rebel, A Elghalbzouri, P Cramers, RGL Nellen, R Willemze, JN Bouwes Bavinck, FR de Gruijl.

Experimental Dermatology, in press.

UV-induced apoptosis is not diminished in the presence of beta-papillomaviruses in habitually unexposed skin, but does decrease with age.

YGL de Graaf, MNC de Koning, WH Zoutman, SJ Weissenborn, H Pfister, L Struijk, MCW Feltkamp, J ter Schegget, R Willemze, JN Bouwes Bavinck, FR de Gruijl. *Submitted*.

Nawoord

Bij het tot stand komen van dit proefschrift zijn vele mensen op verschillende manieren betrokken geweest. Ik wil iedereen hartelijk danken voor alle hulp en steun de afgelopen jaren; in het bijzonder de volgende mensen:

Coby, Aat, Remco, Abdoel en Kees van het Laboratorium Dermatologie, dank voor alle hulp bij de uitvoering en begeleiding van de experimenten.

Heggert, jij was altijd beschikbaar voor advies en praktische hulp. Ik ben erg blij met onze samenwerking voor Hoofdstuk 2. Dank hiervoor!

Wim (Willie), met jou heb ik ontzettend gelachen. Bedankt voor al je hulp! Ruud, onze pogingen tot microdissectie en jouw bijzondere vorm van humor zal ik niet gauw vergeten. Ik ben blij dat we nu collega's zijn, al zit je dan helaas in Maastricht!

Linda, Mariet, Jan, Els, Maurits en Patrick van de Afdeling Medische Microbiologie. Bedankt voor de prettige samenwerking en jullie betrokkenheid bij mijn onderzoek de afgelopen jaren. Ik bewaar vele goede herinneringen aan de HPV-vergaderingen in Leiden, Venetië, Rome, Berlijn, Keulen, Heidelberg en Lyon (!). Maar vooral aan de leuke etentjes en biertjes na afloop!

Soenke Weissenborn and Herbert Pfister. Thank you for the helpful discussion and your help in performing the quantitative PCR for Chapter 4.

All other HPV friends. Thank you for the nice meetings!

Alle medewerkers van de poli en afdeling Huidziekten wil ik bedanken voor de getoonde interesse en steun. In het bijzonder wil ik onze fotograaf Paul bedanken voor alle hulp bij het verfraaien van de tekst en figuren!

Daarnaast natuurlijk alle arts-assistenten en oud-assistenten voor de gezelligheid tijdens de vele assistenten-borrels en etentjes.

De stafleden van de poli Huidziekten, hartelijk dank voor de ruimte die ik kreeg tijdens de laatste fase van het onderzoek die samenviel met de opleiding.

Nelleke van der Zwan-Kralt, dank voor je hulp, vooral het statusonderzoek voor Hoofdstuk 5. Sjoerd van der Burg, bedankt voor de goede begeleiding van Hoofdstuk 3 van dit proefschrift, dat uiteindelijk in een Japans tijdschrift terecht is gekomen!

Ron Wolterbeek, bedankt voor je hulp bij de statistische analyses voor Hoofdstuk 6.

Mijn vriendinnen, lieve Danielle, ik heb het erg getroffen met jou, en dat al sinds de basisschool. Ik vind het heel bijzonder dat ik ook dit moment met jou kan delen! Lieve Marjo, ik ben erg blij met onze vriendschap, zij het op afstand. Na de promotie hoop ik je eindelijk weer eens in Zweden op te zoeken!

Lieve Madelon en Ingrid, bedankt voor al jullie support!

De paranimfen, lieve Irene, wat een geluk dat jij vanuit Nijmegen ook in Leiden begonnen bent als AGIKO! We hebben samen vooral veel gelachen. Ik verheug me op jouw promotie! Lieve Juliette, de afgelopen jaren hebben we veel met elkaar gedeeld, zowel in als buiten het LUMC, tijdens ons 'rondje Rijnsburg'. Dank je wel hiervoor en laten we vooral blijven lopen!

Graag wil ik mijn familie bedanken, in het bijzonder mijn lieve oma's en Riet, dank voor jullie niet aflatende interesse en support.

Wilbert, Wil en Jose, dank voor jullie medeleven.

Hans, jij bent als familie. Bedankt voor al je hulp!

Lieve Rutger en Frits, ik ben erg blij met jullie. Rutger, ik zie uit naar jouw proefschrift! Frits, ga zo door dan zijn we over enkele jaren collega's!

Mijn ouders, lieve pap en mam, bedankt voor al jullie steun. Ik ben erg dankbaar dat jullie er altijd voor me zijn.

Lieve Remko, het is niet in woorden uit te drukken wat jij voor me betekent. Dank je wel voor alles.